EXHIBIT A

FANNIE MAE BYLAWS As amended through January 30, 2009

The Director of the Federal Housing Finance Agency, or FHFA, Fannie Mae's safety, soundness and mission regulator, appointed FHFA as conservator of Fannie Mae on September 6, 2008. As conservator, FHFA succeeded to all rights, titles, powers and privileges of the corporation, and of any stockholder, officer or director of the corporation with respect to the corporation and its assets, and may, by regulation or order, provide for the exercise of any function by any stockholder, director, or officer of Fannie Mae. On November 24, 2008, FHFA, as conservator, reconstituted the Fannie Mae Board of Directors (Board) and directed the functions and authorities of the Board. The Board serves on behalf of the conservator and shall exercise their authority as directed by the conservator. The Bylaws should be read in conjunction with an understanding of the Company's conservatorship status.

Article 1: General Provisions

Section 1.01. *Name.* The name of the corporation is Federal National Mortgage Association. The corporation may also do business under the name Fannie Mae.

Section 1.02. *Principal Office and Other Offices*. The principal office of the corporation shall be in the District of Columbia. Other offices of the corporation shall be in such places as may be deemed by the Board of Directors or the Chief Executive Officer to be necessary or appropriate.

Section 1.03. *Seal.* The seal of the corporation shall be of such design as shall be approved and adopted from time to time by the Board of Directors, and the seal or a facsimile thereof may be affixed by any person authorized by the Board of Directors or these Bylaws by impression, by printing, by rubber stamp, or otherwise.

Section 1.04. *Fiscal Year.* The fiscal year of the corporation shall end on the 31st day of December of each year.

Section 1.05. Corporate Governance Practices and Procedures. Pursuant to Section 1710.10(b) of the Office of Federal Housing Enterprise Oversight ("OFHEO") corporate governance regulation, 12 CFR 1710.1 et seq., to the extent not inconsistent with the Charter Act and other Federal law, rules, and regulations, the corporation has elected to follow the applicable corporate governance practices and procedures of the Delaware General Corporation Law, as the same may be amended from time to time. The inclusion of Sections 1.01, 1.02, 1.05, 2.01, 2.02, 2.03, 2.10, 3.08(b), 3.08(c), 4.01, 4.02, 4.03 and 4.18, Articles 6, 7 and 8, and any new bylaw which may be adopted from time to time and designated as a "Certificate Provision" in accordance with Section 7.01 (collectively, the "Certificate Provisions") in these Bylaws shall constitute inclusion in the corporation's "certificate of incorporation" for all purposes of the Delaware General Corporation's (collectively, the "Bylaw Provisions") shall constitute inclusion in the corporation's "bylaws" for all purposes of the Delaware General Corporation Law.

Article 2: Capital Stock

Section 2.01. Common Stock. The common stock, all of which is voting and has no par value, shall have a stated value per share as determined from time to time by the Board of Directors. Shares of the corporation may be acquired and held in the treasury of the corporation, and may be disposed of by the corporation for such consideration and for such purposes as may be determined from time to time by the Board of Directors.

Section 2.02. Preferred Stock. The corporation shall have authority to issue up to 700,000,000 shares of preferred stock having no par value. The preferred stock may be issued from time to time in one or more series upon approval by the Board of Directors, or a committee thereof appointed for such purpose, and the Board of Directors or such committee may, by resolution providing for the issuance of such preferred stock, designate with respect to such shares: (a) their voting powers; (b) their rights of redemption; (c) their right to receive dividends (which may be cumulative or noncumulative) including the dividend rate or rates, conditions to payment, and the relative preferences in relation to the dividends payable on any other class or classes or series of stock; (d) their rights upon the dissolution of, or upon any distribution of the assets of, the corporation; (e) their rights to convert into, or exchange for, shares of any other class or classes of stock of the corporation, including the price or prices or the rate of exchange; and (f) other relative, participating, optional or special rights, qualifications, limitations or restrictions. Notwithstanding Sections 4.12(a)(6) and 4.16 of these Bylaws, the Board of Directors may authorize a committee of the Board to declare dividends on preferred stock.

Section 2.03. Payment for Shares. The consideration to be received by the corporation for the issuance of common shares shall be fixed from time to time by the Board of Directors. A subscriber shall be entitled to issuance of shares upon receipt by the corporation of the consideration for which the shares are to be issued. No certificates shall be issued for any share until the share is fully paid, and, when issued, such shares shall be nonassessable.

Section 2.04. *Uncertificated Shares.* Any shares of stock of any class or series of the corporation shall be issued in uncertificated form pursuant to customary arrangements for issuing shares in such form, unless a stock certificate is requested by a stockholder.

Section 2.05. Certificates Representing Shares. Each registered holder of the capital stock of the corporation shall be entitled to a certificate or certificates signed by the Chairman of the Board of Directors or the President and by the Secretary or an Assistant Secretary of the corporation, and sealed with the seal of the corporation certifying the number of shares owned by him in the corporation. The certificates shall be in such form as the Board, from time to time, may approve. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

Section 2.06. *Transfers of Stock.* Transfers of stock shall be made upon the books of the corporation at the request of either the registered holder of the stock or the attorney, lawfully constituted in writing, of such registered holder and, in the case of a holder with

a certificate, on surrender for cancellation of the certificate for such share or, in the case of a holder with an uncertificated share, on presentment of proper evidence of succession, assignation or authority to transfer in accordance with customary procedures for transferring shares in uncertificated form.

Section 2.07. Registered *Holder*. The corporation shall be entitled to treat the registered holder of any share or shares of stock as the holder in fact thereof and accordingly shall not be bound to recognize any equitable or other claim to or interest in such share on the part of any other person whether or not it shall have express or other notice thereof, save as expressly provided by the laws of the State of Delaware insofar as they are applicable to the stock of stock corporations organized under the Delaware General Corporation Law.

Section 2.08. Loss or Destruction of Certificate of Stock. In case of loss or destruction of any certificate of stock, another may be issued in its place, pursuant to such requirements and procedures as may be established by the Secretary of the corporation with the concurrence of the General Counsel (including, without limitation, requiring provision of a surety bond).

Section 2.09. Stockholder Records.

- (a) The corporation shall keep at its principal place of business, or at the office of its transfer agent or registrar, a record of its stockholders, giving the names and addresses of all stockholders and the number of shares held by each.
- (b) The officer who has charge of the stock ledger of the corporation shall prepare and make, at least 10 days before every meeting of stockholders, a complete list of the stockholders entitled to vote at said meeting, arranged in alphabetical order, showing the address of and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, for a period of at least 10 days prior to the meeting, during ordinary business hours, at the principal place of business of the corporation or as may otherwise be permitted by the Delaware General Corporation Law. The list shall be produced and kept at the time and place of the meeting during the whole time thereof and may be inspected by any stockholder who is present.

Section 2.10. Registration of common and preferred stock. The corporation shall register its common and preferred stock with the Securities and Exchange Commission as required pursuant to Sections 12(b) or (g) of the Securities Exchange Act of 1934, as amended, and shall take appropriate steps to maintain such registration. Notwithstanding anything to the contrary contained in Section 7.02 of these Bylaws, this Section 2.10 may be altered, amended, or repealed only by the unanimous vote or consent of all the then incumbent Members of the Board then in office.

Article 3: The Stockholders

Section 3.01. Place of Meetings. Meetings of the stockholders of the corporation shall be held at such place or places, within or without the District of Columbia, as shall be determined by the Board of Directors; and the Chairman of the Board (or in his absence another person designated by the Board of Directors) shall preside at all such meetings.

Section 3.02. Annual Meeting. The annual meeting of stockholders shall be held on such date and at such time as the Board of Directors may designate.

Section 3.03. Special Meetings. Special meetings of the stockholders may be called by the Board of Directors or the Chairman of the Board, or at the request of the holders of not less than one-third of all the shares entitled to vote, to be determined as of the close of the first day of the month preceding the month in which the request is presented to the Secretary. Business transacted at all special meetings shall be confined to the subjects stated in the notice of special meeting.

Section 3.04. Notice of Meetings — Waiver and Adjourned Meetings. Written notice stating the place, date and hour of the meeting, and the purpose or purposes for which the meeting is called, shall be delivered not less than 10, nor more than 60, days before the date of the meeting, by the Secretary of the corporation, to each registered holder entitled to vote at such meeting. If mailed, such notice shall be deemed to be delivered when deposited in the United States mail addressed to the registered holder at his address as it appears on the stock transfer books of the corporation, with first class postage prepaid. Waiver by a stockholder in writing of notice of a stockholders' meeting, signed by him either before or after the time of the meeting, shall be equivalent to the giving of such notice. Attendance by a stockholder at a stockholders' meeting, whether in person or by proxy, without objection to the notice or lack thereof, shall constitute a waiver of notice of the meeting. Any meeting of stockholders may be adjourned by the chair of the meeting to reconvene at another time or place. When a meeting is adjourned to another time or place, notice need not be given of the adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken. At the adjourned meeting the corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than 30 days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting.

Section 3.05. Fixing Record Date

- (a) For the purpose of determining stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board of Directors may fix a date as the record date. Such date, in any case, shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors and shall be not more than 60 days and not less than 10 days prior to the date of such meeting. If no such record date is fixed, the close of business on the day next preceding the day on which notice is given, or, if notice is waived, the close of business on the day next preceding the date on which the meeting is held shall be the record date for the determination of stockholders entitled to notice of or to vote at a meeting of stockholders. When a determination of stockholders entitled to vote at any meeting of stockholders has been made, as provided in this section, the determination shall apply to any adjournment thereof, provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.
- (b) For the purpose of determining stockholders entitled to receive payment of any dividend, or in order to make a determination of stockholders for any other

purpose (except as provided in Section 3.05(a), the Board of Directors or a duly authorized Committee thereof may fix a date as the record date. Such date, in any case, shall not precede the date upon which the resolution fixing the record date is adopted and shall be not more than 60 days prior to the date on which the particular action is to be taken. If no such record date is fixed, the close of business on the day on which the resolution relating thereto is adopted shall be the record date for the determination of stockholders.

Section 3.06. *Quorum.* A majority of the shares entitled to vote, represented in person or by proxy, shall constitute a quorum at a meeting of stockholders. The stockholders present at a duly organized meeting may continue to do business until adjournment, notwithstanding the withdrawal of the holders of enough shares to leave less than a quorum. If a meeting cannot be organized because a quorum has not attended, either the chair of the meeting, or those stockholders present, in person or by proxy, by a majority of the votes cast by such stockholders so present, may adjourn the meeting from time to time until a quorum is present when any business may be transacted that may have been transacted at the meeting as originally called.

Section 3.07. *Proxies.* A stockholder may vote either in person or by proxy executed in writing by the stockholder or his duly authorized representative. No proxy shall be valid after 11 months from the date of its execution, unless otherwise expressly provided in the proxy.

Section 3.08. Voting

- (a) At every meeting of the stockholders, every holder of the common stock shall be entitled to one vote for each share of common stock registered in the name of such holder on the stock transfer books of the corporation at the close of the record date. A proxy purporting to be executed by a corporation shall be presumed to be valid and the burden of proving invalidity shall rest on any challenger. A proxy purporting to be executed by a partnership shall be presumed to be valid and the burden of proving invalidity shall rest on any challenger. Unless a higher percentage of affirmative votes is required by the Charter Act, these Bylaws, applicable stock exchange rules or regulations, or other applicable Federal law, rules, or regulations, the stockholders will have approved any matter if, at a meeting at which a quorum is present, the votes cast by the stockholders present, either in person or by proxy and entitled to vote thereon, in favor of such matter exceed the votes cast by such stockholders against such matter.
- (b) Except as provided in Section 308 (b) of the Charter Act, members of the Board of Directors shall be elected by a majority of the votes cast in person or by proxy at any meeting that includes the election of directors at which a quorum is present, provided that if (i) the number of nominees exceeds the number of directors to be elected or (ii) the Secretary of the Corporation received notice that a stockholder nominated a person for election to the Board of Directors in accordance with Section 4.20 of these Bylaws, and that nomination has not been withdrawn by the stockholder on or before the tenth day preceding the date the corporation first mails its meeting notice to stockholders, the directors are to be elected by a plurality of the votes cast in person or by proxy. For purposes of this Section, a majority of the votes cast means that the number of shares voted "for" a director must exceed the number of votes cast "against" that director. For purposes of this Section, if plurality voting is applicable to

the election of directors at any meeting, the director nominees who receive the highest number of votes cast "for", without regard to votes cast "against," shall be elected as directors up to the total number of directors to be elected at that meeting. Abstentions and broker non-votes will not count as a vote cast with respect to a director's election.

(c) If an incumbent director fails to receive the required vote for re-election, the Nominating and Corporate Governance Committee will review the director's previously submitted irrevocable resignation (which is contingent upon (i) his or her failure to receive the required vote and (ii) Board acceptance of such resignation), will act on an expedited basis to determine whether to accept such director's resignation, and will submit such recommendation for prompt consideration by the Board. The Board expects the director whose resignation is under consideration to abstain from participating in any decision regarding that resignation. The Nominating and Corporate Governance Committee and the Board may consider any factors they deem relevant in deciding whether to accept a director's resignation. The Board will publicly disclose (in accordance with Section 3.12 of these Bylaws) its decision regarding the tendered resignation and the rationale for the decision within 90 days after the date of certification of the election results. If such incumbent director's resignation is not accepted by the Board, such director will continue to serve until the next meeting that includes the election of directors and until his or her successor is chosen and qualified, or his or her death, resignation, or retirement or removal in accordance with applicable law or regulation, whichever event shall first occur. If a director's resignation is accepted by the Board, or if a nominee for director is not elected and the nominee is not an incumbent director, then the Board, in its sole discretion, may fill any resulting vacancy pursuant to the provisions of Section 308(b) of the Charter Act.

Section 3.09. Inspectors of Votes. The Board of Directors, in advance of any meeting of stockholders, shall appoint one or more Inspectors of Votes to act at the meeting or any adjournment thereof and make a written report thereof. One or more persons may be designated as alternates to replace any Inspector of Votes who fails to act. In case any person so appointed Inspector of Votes or alternate resigns or fails to act, the vacancy shall be filled by appointment made by the chairman of the meeting. The Inspectors of Votes shall (a) ascertain the number of shares outstanding and the voting power of each and determine all questions concerning the qualification of voters; (b) determine the shares represented at the meeting and the validity of proxies and ballots; (c) determine all questions concerning the acceptance or rejection of votes and, with respect to each vote by ballot, shall collect and count all votes and ballots; (d) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the Inspectors of Votes; and (e) report in writing to the secretary of the meeting their determination of the number of shares represented at the meeting, and their count of all votes and ballots. The Inspectors of Votes need not be stockholders of the corporation. No person who is an officer or Member of the Board of Directors of the corporation, or who is a candidate for election as a Member of the Board of Directors, shall be eligible to be an Inspector of Votes. Any report or certificate by the Inspectors of Votes shall be prima facie evidence of the facts stated and of the votes as certified by them.

Section 3.10. Stockholder Notices to the Corporation. Whenever notice is to be given to the corporation by a stockholder under any provision of law or of these Bylaws, such

notice shall be delivered to the Secretary at the principal executive offices of the corporation. If delivered by electronic mail or facsimile, the stockholder's notice shall be directed to the Secretary at the electronic mail address or facsimile number, as the case may be, specified in the corporation's most recent proxy statement.

Section 3.11. Conduct of Meetings. The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at such meeting by the person presiding over the meeting. The Board of Directors may adopt by resolution such rules or regulations for the conduct of meetings of stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board of Directors, the chair of any meeting of stockholders shall have the right and authority to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chair, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the chair of the meeting, may include, without limitation, the following: (a) the establishment of an agenda or order of business for the meeting; (b) rules and procedures for maintaining order at the meeting and the safety of those present; (c) limitations on attendance at or participation in the meeting to stockholders of record of the corporation, their duly authorized and constituted proxies, or such other persons as the chair shall permit; (d) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (e) limitations on the time allotted to questions or comments by participants. Meetings of stockholders shall not be required to be held in accordance with rules of parliamentary procedure.

Section 3.12. Notice of Business to be Brought Before an Annual Meeting. At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting. business must be (a) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors; (b) otherwise properly brought before the meeting by or at the direction of the Board of Directors; or (c) otherwise properly brought before the meeting by a stockholder. For business to be properly brought before an annual meeting by a stockholder (other than the nomination of a person for election as a director, which is governed by Section 4.20 of these Bylaws), the stockholder must have given timely notice thereof in writing to the Secretary of the corporation. To be timely, a stockholder's notice must be delivered to or mailed and received at the principal executive offices of the corporation not earlier than the close of business on the 120th day and not later than the close of business on the 60th day prior to the first anniversary of the preceding year's annual meeting; provided, however, that in the event that the date of the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of the 60th day prior to such annual meeting and the 10th day following the day on which public disclosure of the date of such meeting is first made by the corporation. In no event shall the public disclosure of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above. (For purposes of these Bylaws, public disclosure shall be deemed to include a disclosure made in a press release reported by the Dow Jones News Services, Associated Press or a comparable national news service or in a document filed by the corporation with the Securities and Exchange Commission pursuant to Section 13 of the Securities Exchange Act of 1934, as amended.) A

stockholder's notice to the Secretary shall set forth as to each matter the stockholder proposes to bring before the annual meeting (A) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting; (B) the name and address, as they appear on the corporation's books, of the stockholder proposing such business; (C) the class and number of shares of the corporation that are beneficially owned by the stockholder; and (D) any material interest of the stockholder in such business. Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at any annual meeting except in accordance with the procedures set forth in this Section 3.12. The chair of the annual meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting in accordance with the provisions of this Section 3.12, and if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

Article 4: The Board of Directors

Section 4.01. *General Policies.* General policies governing the operations of the corporation shall be determined by the Board of Directors.

Section 4.02. *Membership.* The Board of Directors shall consist of those Members appointed and elected as provided by law.

Section 4.03. Term of Members. Each Member shall hold office for the term for which he is elected or appointed and until his successor is chosen and qualified, or his death, resignation, or retirement or removal in accordance with applicable law or regulation, whichever event shall first occur.

Section 4.04. Regular Meetings. The Board of Directors shall meet in regular meetings at such times as shall be determined by the Board from time to time, except as provided in section 4.05 and except when the Chairman of the Board shall notify the Secretary of a different date prior to a scheduled regular meeting. Each regular meeting shall be held at the principal office of the corporation in the District of Columbia, unless special provision is made by the Board, in advance of any such regular meeting, to hold that meeting at another place, either within or without the District of Columbia.

Section 4.05. Annual Meeting. Immediately following the annual meeting of the stockholders, the Board of Directors shall meet each year for the purpose of considering any business that may properly be brought before the meeting, and such annual meeting of the Board shall be a regular meeting.

Section 4.06. Special Meetings. Other meetings of the Board of Directors may be held upon the call of the Chairman of the Board of Directors, or of a majority of the then incumbent Members of the Board. Each special meeting shall be held at the principal office in the District of Columbia unless the Chairman of the Board prescribes and the notice specifies another place.

Section 4.07. Notice of Meetings — Waiver. No notice of any kind to Members of the Board of Directors shall be necessary for any regular meeting that is held on a date determined by the Board, or for the annual meeting. In the case of a regular meeting on a different date, notice shall be given to each Member by the Secretary; in the case of a

special meeting, notice shall be given to each Member by the Secretary at the direction of the calling authority. Such notice shall be in writing and sent to the address on file with the Secretary of the corporation not later than during the third day immediately preceding the day for the meeting; or by word of mouth, telephone, facsimile or electronic mail, directed to the telephone number, facsimile number or electronic mail address, as the case may be, on file with the Secretary of the corporation, not later than during the second day immediately preceding the day for the meeting. The attendance of any Member at a meeting shall constitute a waiver of notice by such Member, except where such Member attends for the express purpose of protesting at the beginning of the meeting the lack of notice of the meeting. Neither the business to be transacted at, nor the purpose of, any meeting of the Board of Directors need be specified in the notice of the meeting.

Section 4.08. The Chairman of the Board of Directors. The Chairman of the Board of Directors may be chosen by the Board at any meeting of the Board from among the Members, and his tenure shall commence immediately and continue until the next succeeding annual meeting of the Board, or until his successor is chosen, whichever occurs first. The Chairman of the Board (or in his absence another person designated by the Board of Directors) shall preside at all meetings of the Board of Directors and at meetings of stockholders. In addition, the Chairman of the Board shall have such powers and perform such duties as the Board may prescribe. Except as otherwise provided by law, the Charter Act, these Bylaws, or the Board, the Chairman shall have plenary authority to perform all duties as may be assigned to him from time to time by the Board.

Section 4.08a. The Vice Chairman of the Board of Directors. The Board of Directors may from time to time elect from among the Members of the Board one or more Vice Chairmen of the Board. Any such Vice Chairman shall have such powers and shall perform such duties as the Board of Directors may prescribe or as the Chairman of the Board shall delegate to him.

Section 4.09. *Quorum.* The presence, in person or otherwise in accordance with section 4.17 hereof, of a majority of the then incumbent Members of the Board of Directors or of a Board Committee, as applicable, at the time of any meeting of the Board or such Committee, shall constitute a quorum for the transaction of business. The act of the majority of such Members present at a meeting at which a quorum is present shall be the act of the Board of Directors unless the act of a greater number is required by these Bylaws. Members may not be represented by proxy at any meeting of the Board of Directors or a Board Committee.

Section 4.10. Action Without a Meeting. Any policy or action that may be approved or taken at a meeting of the Board or of any Board Committee may be approved or taken without a meeting if all incumbent Members of the Board or the Committee, as the case may be, consent thereto in writing and the writings are filed with the minutes of the proceedings of the Board or the Committee.

Section 4.11. Facsimile Signatures. The Board of Directors, the Chairman of the Board, the Chief Executive Officer or any designee of the Chief Executive Officer may authorize the use of facsimile signatures in lieu of manual signatures.

Section 4.12. Executive Committee.

- a. The Executive Committee of the Board shall consist of at least five Members who shall be designated by the Board and serve at the pleasure of the Board. One of the members of the Executive Committee shall be the Chief Executive Officer of the corporation who may also, but is not required to, be chair of the Committee. The designation of such Committee and the delegation thereto of authority shall not alone relieve any director of any duty he owes the corporation. The Executive Committee, during the interim between Board meetings, shall have the authority of the Board, except that it shall not have the authority to take any of the following actions:
 - 1. The submission to stockholders of any action requiring stockholders' authorization.
 - 2. The filling of vacancies on the Board of Directors or on the Executive Committee.
 - 3. The fixing of compensation of the directors for serving on the Board or on the Executive Committee.
 - 4. The appointment or removal of the Chairman of the Board, Chief Executive Officer, President, any Vice Chairman, and any Executive Vice President, except that vacancies in established positions may be filled subject to ratification by the Board of Directors.
 - 5. The amendment or repeal of these Bylaws or the adoption of new bylaws.
 - 6. The declaration of dividends or the authorizing of the issuance of the corporation's stock.
 - 7. The amendment or repeal of any resolution of the Board which by its terms is not so amendable or repealable.
 - 8. The adoption of an agreement of merger or consolidation or the adoption of a certificate of ownership and merger.
 - 9. The recommendation to stockholders of the sale, lease or exchange of all or substantially all of the corporation's property and assets.
 - 10. The recommendation to stockholders of a dissolution of the corporation or a revocation of a dissolution.
- b. The Executive Committee shall meet at the call of its chairman or of a majority of its members, and a majority shall constitute a quorum. The action of the majority of the members of the Committee shall be the action of the Committee.
- c. Unless otherwise expressly provided by resolution of the Board of Directors, members of the Executive Committee shall be compensated and shall be reimbursed for travel and expenses on the same basis and at the same rate as is provided for Members of the Board of Directors for attendance at meetings of the Board.

- d. At the first regular meeting of the Board of Directors following a meeting of the Executive Committee, the Executive Committee shall present to the Board a report and such recommendations as are in its judgment necessary for the proper operation of the corporation.
- **Section 4.13**. Audit Committee. The Board of Directors shall have an Audit Committee and, as required by Section 1710.12(c)(1) of the OFHEO corporate governance regulation, as the same may be amended from time to time, the Audit Committee shall comply with the charter, independence, composition, expertise and other requirements under section 301 of the Sarbanes-Oxley Act of 2002 and under rules issued by the New York Stock Exchange, as the same may be amended from time to time.
- **Section 4.14.** Compensation Committee. The Board of Directors shall have a Compensation Committee and, as required by Section 1710.12(c)(2) of the OFHEO corporate governance regulation, as the same may be amended from time to time, the Compensation Committee shall comply with the charter, independence, composition, expertise and other requirements under the rules issued by the New York Stock Exchange, as the same may be amended from time to time. The duties of the Compensation Committee shall include overseeing the corporation's compensation policies and plans for executive officers and employees and approving the compensation of principal officers of the corporation.
- **Section 4.15.** Nominating and Corporate Governance Committee. The Board of Directors shall have a Nominating and Corporate Governance Committee, as required by Section 1710.12(c)(3) of the OFHEO corporate governance regulation, as the same may be amended from time to time. The Nominating & Corporate Governance Committee shall comply with the charter, independence, composition, expertise and other requirements under the rules issued by the New York Stock Exchange, as the same may be amended from time to time.
- **Section 4.16.** Other Committees. In addition to the Executive, Audit, Compensation, and Nominating & Corporate Governance committees, the Board of Directors may by resolution designate from among its Members such other committees as it deems appropriate, each of which, to the extent provided by resolution of the Board, may exercise all authority of the Board except those actions outside the authority of the Executive Committee. The designation of any such committee and the delegation thereto of authority shall not alone relieve any director of any duty he owes the corporation.
- **Section 4.17**. Remote Meetings. Any meeting of the Board of Directors or any meeting of a Board Committee may be held with the Members of the Board or members of such Committee participating in such meeting by telephone or by any other means of communication by which all such persons participating in the meeting are able to speak to and hear one another.
- **Section 4.18.** *Limitation on Liability.* To the fullest extent permitted by Delaware statutory and decisional law, as amended or interpreted, no director of this corporation shall be personally liable to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. This Section 4.18 does not affect the availability of equitable remedies for breach of fiduciary duties.

Section 4.19. *Eligibility to Make Nominations.* Nominations of candidates for election as directors at an annual meeting of stockholders called for election of directors may be made (i) by any stockholder entitled to vote at such meeting only in accordance with the procedures established by Section 4.20 of these Bylaws, or (ii) by the Board of Directors or by a duly authorized Committee thereof. In order to be eligible for election as a director, any director nominee must first be nominated in accordance with the provisions of these Bylaws.

Section 4.20. Procedure for Nominations by Stockholders. Any stockholder entitled to vote for the election of a director at an annual meeting may nominate one or more persons for such election only if written notice of such stockholder's intent to make such nomination is delivered to or mailed and received by the Secretary of the corporation. Such notice must be received by the Secretary not later than the following dates: with respect to an annual meeting of stockholders, not earlier than the close of business on the 120th day and not later than the close of business on the 60th day prior to the first anniversary of the preceding year's annual meeting; provided, however, that in the event that the date of the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of the 60th day prior to such annual meeting and the 10th day following the day on which public disclosure of the date of such meeting is first made by the corporation. The written notice shall set forth: (1) the name, age, business address and residence address of each nominee proposed in such notice: (2) the principal occupation or employment of each such nominee: (3) the class of securities and the number of shares of capital stock of the corporation which are beneficially owned by each such nominee; and (4) such other information concerning each such nominee as would be required, under the rules of the Securities and Exchange Commission in a proxy statement soliciting proxies for the election of such nominee as a director. Such notice shall include a signed consent of each such nominee to serve as a director of the corporation, if elected and a statement whether such nominee, if elected, intends to tender, promptly following such nominee's election or reelection, an irrevocable resignation effective upon such nominee's failure to receive the required vote for re-election at the next meeting of stockholders at which such nominee faces re-election and upon acceptance of such resignation by the board of directors. The corporation may also require any proposed nominee to furnish such other information as may be reasonably required by the corporation to determine whether such proposed nominee is eligible to serve as an independent director of the corporation or that could be material to a reasonable stockholder's understanding of independence, or lack thereof, of such nominee.

Section 4.21. Compliance with Procedures. If the chair of the stockholders' annual meeting determines that a nomination of any candidate for election as a director was not made in accordance with the applicable provisions of these Bylaws, such nomination shall be void.

Article 5: The Officers

Section 5.01. *Number.* The principal officers of the corporation shall consist of the Chief Executive Officer, a President, one or more Vice Chairmen of the Board if the Board has elected to fill such position or positions, one or more Executive Vice

Presidents and Senior Vice Presidents, a General Counsel, a Controller, a Treasurer, and a Secretary. There shall be such other officers, assistant officers, agents, and employees as may be deemed necessary. Any two or more offices may be held by the same person.

Section 5.02. General Authority and Duties. All officers, agents, and employees of the corporation shall have such authority and perform such duties in the management and conduct of the business of the corporation as may be provided for in these Bylaws, as may be established by resolution of the Board of Directors not inconsistent with these Bylaws, as generally pertain to their respective offices, and as may be delegated to them in a manner not inconsistent with these Bylaws.

Section 5.03. *Election, Tenure, and Qualifications.* The principal officers shall be selected by the Board of Directors. Each officer shall hold office until his successor is chosen and qualified, or his death, resignation, retirement, or removal from office, whichever event shall first occur. Selection or appointment without express tenure, of an officer, agent, or employee shall not of itself create contract rights.

Section 5.04. Removal. Any officer, agent, or employee may be removed by the Board of Directors. Any removal shall be in accordance with such procedures and safeguards as the corporation may establish and shall be without prejudice to the contract rights, if any, of the person so removed.

Section 5.05. *Vacancies.* Any vacancy in any office shall be filled in the manner prescribed in these Bylaws for selection or appointment to the office.

Section 5.06. Chief Executive Officer. The Chief Executive Officer shall have the general powers and duties of supervision, management and direction over the business and policies of the corporation. The Chief Executive Officer shall see that all orders and resolutions of the Board of Directors and any committee thereof are carried into effect, and shall submit reports of the current operations of the corporation to the Board of Directors at regular meetings of the Board of Directors and in annual reports to the stockholders.

Section 5.07. *The President.* The President shall have such powers and perform such duties as the Board of Directors may prescribe, or, if the President is not also the Chief Executive Officer, the Chief Executive Officer may delegate to him.

Section 5.08. The Vice Presidents. Each Vice President shall have such powers and perform such duties as the Board of Directors may prescribe or as the Chief Executive Officer may delegate to him.

Section 5.09. The Treasurer. The Treasurer shall, in general, perform all the duties ordinarily incident to the office of Treasurer and such other duties as may be assigned to him by the Board of Directors or by the Chief Executive Officer or his designee. The Treasurer shall render to the Board of Directors or the Chief Executive Officer or his designee, whenever the same shall be required, an account of all his transactions as Treasurer. The Treasurer shall, if required to do so by the Board, give the corporation a bond in such amount and with such surety or sureties as may be ordered by the Board for the faithful performance of the duties of his office and for the restoration to the corporation, in case of his death, resignation, retirement, or removal from office, of all

books, papers, vouchers, money, and other property of whatever kind in his possession or under his control belonging to the corporation. The premium for any such bond shall be paid by the corporation.

Section 5.10. The General Counsel. The General Counsel shall be the principal consulting officer of the corporation in all matters of legal significance or import; shall be responsible for and direct all counsel, attorneys, employees, and agents in the performance of all legal duties and services for and on behalf of the corporation; shall perform such other duties and have such other powers as are ordinarily incident to the office of the General Counsel; and shall perform such other duties as, from time to time, may be assigned to him by the Board of Directors or by the Chief Executive Officer.

Section 5.11. The Secretary. The Secretary shall keep or cause to be kept in books provided for the purpose the minutes of the meetings of the Board of Directors and the minutes or transcripts of the meetings of the stockholders; shall see that all notices are duly given as required by law and in accordance with the provisions of these Bylaws; shall be responsible for the custody and maintenance of all related records and the blank stock certificates of the corporation; shall be custodian of the records and of the seal of the corporation; and, in general, shall perform all the duties ordinarily incident to the office of Secretary and such other duties as may be assigned to him by the Board or by the Chief Executive Officer. The Secretary and any Assistant Secretary are expressly empowered to attest signatures of officers of the corporation and to affix the seal of the corporation to documents.

Section 5.12. The Controller. The Controller shall keep full and accurate accounts of all assets, liabilities, commitments, receipts, disbursements, and other financial transactions of the corporation; and in general, shall perform all the duties ordinarily incident to the office of Controller and such other duties as may be assigned to him by the Board of Directors or by the Chief Executive Officer or his designee.

Section 5.13. Assistant Officers.

Each assistant to an officer, including but not limited to any Assistant Vice President, any Assistant Treasurer, any Assistant General Counsel, and any Assistant Secretary, and any other such assistant to any officer, shall perform such duties as are, from time to time, delegated to him by the officer to whom he is an assistant, by the Board of Directors or by the Executive Officer or his designee. At the request of the officer to whom he is an assistant, an assistant officer may temporarily perform the duties of that officer, and when so acting shall have the powers of and be subject to the restrictions imposed upon that officer.

Section 5.14. Compensation. Subject to the approval of the Conservator, if so required, the compensation of the principal officers shall be fixed, from time to time, by the Board of Directors.

Article 6: Indemnification

Section 6.01. General Indemnification. The Board of Directors may, in such cases or categories of cases as it deems appropriate, indemnify and hold harmless, or make provision for indemnifying and holding harmless, Members of the Board of Directors, officers, employees, and agents of the corporation, and persons who formerly held such positions, and the estates of any of them against any or all claims and liabilities

(including reasonable legal fees and other expenses incurred in connection with such claims or liabilities) to which any such person shall have become subject by reason of his having held such a position or having allegedly taken or omitted to take any action in connection with such position.

Section 6.02. Indemnification of Board Members and Officers.

- a. To the fullest extent permitted by the Delaware General Corporation Law for a corporation subject to such law, as the same exists or may hereafter be amended (but, in the case of any such amendment, only to the extent that such amendment permits a Delaware corporation to provide broader indemnification rights than said law permitted such corporation to provide prior to such amendment), the corporation will indemnify and hold harmless each Member of the Board and officer of the corporation or any subsidiary against any and all claims, liabilities, and expenses (including attorneys' fees, judgments, fines, and amounts paid in settlement) actually and reasonably incurred and arising from any threatened, pending, or completed action, suit or proceeding, whether civil, criminal, administrative, or investigative, to which any such person shall have become subject by reason of having held such a position or having allegedly taken or omitted to take any action in connection with any such position. However, the foregoing shall not apply to:
 - i. any breach of such person's duty of loyalty to the corporation or its stockholders:
 - ii. any act or omission by such person not in good faith or which involves intentional misconduct or where such person had reasonable cause to believe his conduct was unlawful, or
 - iii. any transaction from which such person derived any improper personal benefit.
- b. The decision concerning whether a particular indemnitee has satisfied the foregoing shall be made by (i) the Board of Directors by a majority vote of a quorum consisting of Members who are not parties to the action, suit, or proceeding giving rise to the claim for indemnity ("Disinterested Directors"), whether or not such majority constitutes a quorum; (ii) a committee of Disinterested Directors designated by a majority vote of Disinterested Directors, whether or not such majority constitutes a quorum; (iii) if there are no Disinterested Directors, or if the Disinterested Directors so direct, by independent legal counsel in a written opinion; or (iv) a vote of the stockholders.
- c. The Board of Directors may authorize the advancement of expenses to any Member of the Board or officer, subject to a written undertaking to repay such advance if it is later determined that the indemnitee does not satisfy the standard of conduct required for indemnification. The Chairman of the Board is authorized to enter into contracts of indemnification with each Member and officer of the corporation with respect to the indemnification provided in the Bylaws and to renegotiate such contracts as necessary to reflect changing laws and business circumstances.

Article 7: Amendments

Section 7.01. Actions by the Board of Directors. The Board of Directors has the power to alter, amend, or repeal any Certificate Provision or Bylaw Provision of these Bylaws, or to adopt new bylaws, either (i) by the affirmative vote of two-thirds of the then incumbent Members of the Board of Directors, with the exception of Section 2.10, or (ii) in the manner provided in Section 4.10 of these Bylaws. Except by unanimous consent of all the then incumbent Members of the Board, no such action shall be undertaken until at least one week shall have elapsed from either (i) the introduction of the proposal at a meeting of the Board of Directors at which a quorum shall have attended, or (ii) the circulation of such proposed action to all the then incumbent Members of the Board. Any (i) new bylaw adopted by the Board of Directors and (ii) Certificate Provision, as altered or amended by the Board of Directors pursuant to this Section 7.01, shall be designated a "Certificate Provision" for all purposes under these Bylaws unless, by the affirmative vote of two-thirds of the then incumbent Members of the Board of Directors, the Board of Directors shall approve the designation of such bylaw as a "Bylaw Provision" for all purposes under these Bylaws.

Section 7.02. Actions by the Stockholders.

a. Bylaw Provisions. The stockholders have the power to alter, amend, or repeal any Bylaw Provision, or to adopt any new bylaw, the subject matter of which is the subject matter of a Bylaw Provision, by the affirmative vote of a majority of the stock issued and outstanding and entitled to vote at any regular meeting of the stockholders or at any special meeting of the stockholders if notice of such proposed action be contained in the notice of such special meeting; provided, however, that notwithstanding the foregoing, the stockholders shall not have the power to alter, amend or repeal any Bylaw Provision. or adopt any new bylaw, if (i) such Bylaw Provision, as proposed to be altered or amended, or the repeal of such Bylaw Provision, or the new bylaw proposed for adoption, is or would be inconsistent with the Charter Act or other Federal law, rules, and regulations or the safe and sound operations of the corporation, in each case as determined by the applicable regulator, (ii) the subject matter of such Bylaw Provision, as proposed to be altered or amended, or the subject matter of the new bylaw proposed for adoption is the subject matter of any Certificate Provision, or (iii) such Bylaw Provision, as proposed to be altered or amended, or the repeal of such Bylaw Provision, or the new bylaw proposed for adoption is or would be inconsistent with any Certificate Provision. Notwithstanding anything to the contrary herein, any action by the stockholders pursuant to Section 7.02 shall be null and void, without legal effect, if such action shall violate any law, rule or regulation by any government authority applicable to this corporation, including, without limitation, the Charter Act, or any rule, regulation or other requirement of any stock exchange on which the stock of this corporation is then listed. For the avoidance of doubt, any proposed action by the stockholders pursuant to this Section 7.02 will be subject to Article 8 of these Bylaws.

b. Certificate Provisions. The stockholders may not alter, amend, repeal or adopt any Certificate Provision unless such action is explicitly authorized and referred to the stockholders by the Board of Directors. No such authorization and referral shall be made by the Board of Directors unless such authorization and referral is approved pursuant to the procedures set forth in Section 7.01. For the avoidance of doubt, this Section 7.02(b)

in no way obligates the Board of Directors to seek stockholder approval for any action pursuant to Section 7.01.

Article 8: Regulatory Powers

Nothing in these Bylaws shall be deemed to affect the regulatory or conservatorship powers of the Federal Housing Finance Agency under the Federal Housing Enterprises Financial Safety and Soundness Act of 1992, Title XIII, P.L. 102-550, as amended by the Federal Housing Finance Regulatory Reform Act of 2008, P.L. 110-289.

EXHIBIT B

BYLAWS OF THE FEDERAL HOME LOAN MORTGAGE CORPORATION

As amended and restated July 13, 2015

FREDDIE MAC BYLAWS TABLE OF CONTENTS

			Page
ARTICLE 1	OFF	ICES	1
Section	on 1.1	Offices.	1
ARTICLE 2	CAP	ITAL STOCK	1
Section	on 2.1	Issuance	1
Section	on 2.2	Common Stock	1
Section 2.3		Preferred Stock	1
Section	on 2.4	Consideration	1
Section	on 2.5	Shares Owned by the Corporation	1
Section 2.6		Fractional Shares	2
Section	on 2.7	Certificates	2
Section	on 2.8	Transfer of Shares	2
Section 2.9		Lost, Destroyed and Mutilated Certificates	3
Section 2.10		Fixing a Record Date	3
Section	on 2.11	Record of Stockholders	4
Section	on 2.12	Ownership of Shares	4
Section	on 2.13	Share Options and Other Instruments	4
ARTICLE 3	MEE	TINGS OF THE STOCKHOLDERS	4
Section	on 3.1	Annual Meetings	4
Section	on 3.2	Special Meetings.	5
Section	on 3.3	Notice of Meetings	6
Section	on 3.4	Quorum	8
Section	on 3.5	Organization	8
Section	on 3.6	Conduct of Business	8
Section	on 3.7	Voting	10
Section	on 3.8	Voting Entitlement of Shares	11
Section	on 3.9	Proxies	12
Section	on 3.10	Stockholders' List	12
ARTICLE 4	ВОА	RD OF DIRECTORS	13
Section	on 4.1	General Powers	13

TABLE OF CONTENTS

(continued)

Page

			_
	Section 4.2	Number, Qualification and Term of Office	13
	Section 4.3	Nominations	13
	Section 4.4	Vacancies	15
	Section 4.5	Elections.	15
	Section 4.6	Chairman	16
	Section 4.7	Lead Director	16
	Section 4.8	Regular Meetings	16
	Section 4.9	Special Meetings	16
	Section 4.10	Quorum	17
	Section 4.11	Participation in Meetings	17
	Section 4.12	Conduct of Business	17
	Section 4.13	Reimbursement and Compensation of Directors	18
	Section 4.14	Committees of the Board of Directors	18
	Section 4.15	Resignation	19
	Section 4.16	Removal of Directors	19
	Section 4.17	Termination of Voluntary Registration of Common Stock	19
ARTIC	CLE 5 OFFIC	DERS	19
	Section 5.1	Officers of the Corporation	19
	Section 5.2	Appointment and Term	20
	Section 5.3	Removal, Resignation, Vacancy	20
	Section 5.4	Compensation	21
	Section 5.5	Duties	21
	Section 5.6	Chief Executive Officer	21
	Section 5.7	President	21
	Section 5.8	Chief Operating Officer	21
	Section 5.9	Senior Vice President – General Auditor	21
	Section 5.10	Vice Presidents	21
	Section 5.11	Corporate Secretary	22
	Section 5.12	Delegation of Authority	22

TABLE OF CONTENTS

(continued)

		· ·	age
ARTICLE 6	INSPE	ECTION OF RECORDS	22
Section (6.1	Inspection of Records by Stockholders	22
Section (6.2	Inspection of Records by Directors	
ARTICLE 7	NOTIO	DES	
Section ³	7.1	Notices	23
Section ³	7.2	Written Waivers	25
ARTICLE 8	INDE	MNIFICATION AND LIMITATION OF LIABILITY	26
Section 8	8.1	Indemnification	26
Section 8	8.2	Right of Indemnitee to Bring Suit	28
Section 8	8.3	Non-Exclusivity of Rights	
Section 8	8.4	Insurance	29
Section 8	8.5	Limitation of Liability	29
ARTICLE 9	SEAL		29
Section 9	9.1	Corporate Seal	29
ARTICLE 10	FISCA	AL YEAR	29
Section	10.1	Fiscal Year	29
ARTICLE 11	MISCI	ELLANEOUS	29
Section	11.1	Time Periods	29
Section	11.2	Severability	29
Section	11.3	Corporate Governance Practices and Procedures and Governing Law	30
Section	11.4	Certificates of Designation	30
Section	11.5	Statutory References	30
Section	11.6	Control Share Acquisitions	30
ARTICLE 12	AMEN	IDMENTS	31
Section	12.1	General	31
Section	12.2	Amendment by the Board of Directors	31
Section	12.3	Amendment by the Stockholders	31

ARTICLE 1 - OFFICES

Section 1.1 Offices. The principal office of the Corporation shall be in Fairfax County, Virginia or at any other place determined by the Board of Directors. The Corporation may have such other offices as the Board of Directors or the Chief Executive Officer shall determine appropriate.

ARTICLE 2 - CAPITAL STOCK

Section 2.1 <u>Issuance.</u> The Board of Directors shall have the power to authorize the issuance of one or more classes or series of stock of the Corporation, including, without limitation, voting common and preferred stock. All stock shall be issued on such terms and conditions as the Board of Directors shall prescribe from time to time.

Section 2.2 Common Stock. The voting common stock of the Corporation (the "Common Stock") shall consist of such number of shares as may be issued or authorized for issuance from time to time by the Board of Directors (without limitation upon the authority of the Board of Directors to authorize the issuance of additional shares from time to time). The Common Stock shall have the designation, powers, rights, privileges, qualifications, limitations, restrictions, terms and conditions set forth in the Eighth Amended and Restated Certificate of Designation, Powers, Preferences, Rights, Privileges, Qualifications, Limitations, Restrictions, Terms and Conditions of Voting Common Stock adopted on September 10, 2008, as further amended or restated from time to time (the "Common Stock Certificate of Designation"). No holder of Common Stock shall as such holder have any preemptive right to purchase or subscribe for any other shares, rights, options, or other securities of any class of the Corporation which at any time may be sold or offered for sale by the Corporation.

Section 2.3 <u>Preferred Stock.</u> The preferred stock of the Corporation shall consist of such number of shares as may be issued or authorized for issuance from time to time by the Board of Directors (without limitation upon the authority of the Board of Directors to authorize the issuance of additional shares from time to time). Each class of preferred stock shall have the designation, powers, preferences, rights, privileges, qualifications, limitations, restrictions, terms and conditions set forth in the certificate of designation approved by the Board of Directors for such class.

Section 2.4 <u>Consideration.</u> Shares of stock may be issued to the Corporation's stockholders pro rata and without consideration. Shares of stock may also be issued for consideration consisting of any tangible or intangible property or benefit to the Corporation as the Board of Directors deems appropriate. Upon the Board of Directors making a good faith determination that the consideration received for the shares to be issued is adequate, the shares issued therefor shall be fully paid and nonassessable.

Section 2.5 Shares Owned by the Corporation. Any shares of capital stock owned by the Corporation shall retain the status of issued shares, unless and until the

Corporation shall retire and cancel the same, but such shares shall not be regarded as outstanding while so owned.

Section 2.6 <u>Fractional Shares.</u> No fractional interests in shares of common stock will be created or recognized by the Corporation except as otherwise provided in the Corporation's Employee Stock Purchase Plan or any other executive compensation or employee benefit plan or any direct stock purchase plan currently in effect or hereafter adopted by the Corporation. The holder of a fractional share is entitled to exercise the rights of a shareholder, including the right to vote, to receive dividends, and to participate in the assets of the Corporation upon dissolution.

Section 2.7 Certificates.

- The Board of Directors may authorize the issuance of shares of stock of the Corporation with or without certificates. The rights and obligations of stockholders shall be identical whether or not their shares are represented by certificates. Stock certificates shall be in the form approved by the Corporate Secretary. Each stock certificate shall contain the name of the Corporation, the name of the stockholder, the number and kind of shares of stock owned by such stockholder, and reference to any other material terms of the stock represented thereby, including, without limitation, the information required to be set forth on such certificates by the Common Stock Certificate of Designation, shall be signed by the Chief Executive Officer or President and countersigned by the Corporate Secretary or an Assistant Secretary, and shall be sealed with the Corporation's seal or a facsimile of such seal. Within a reasonable time after the issuance or transfer of shares without certificates, the Corporation shall send to the registered stockholder a written statement containing the information required to be set forth on the certificates, including, without limitation, the information required to be set forth on such certificates by the Common Stock Certificate of Designation, or a statement that the Corporation will furnish such information upon request and without charge.
- (b) When any stock certificate is countersigned by a transfer agent or a registrar, other than the Corporation or its employee, any other signature on such certificate may be a facsimile. If any corporate officer who has signed any certificate ceases to be a corporate officer before such certificate is issued, whether because of death, resignation or otherwise, the certificate may nevertheless be issued and delivered by the Corporation as if such officer had not ceased to be such as of the certificate's issue date.
- **Section 2.8** <u>Transfer of Shares.</u> Except as otherwise provided in the Corporation's Employee Stock Purchase Plan or any other executive compensation or employee benefit plan or any direct stock purchase plan currently in effect or hereafter adopted by the Corporation, the Common Stock shall be transferable only in whole shares. Subject to the foregoing, the stock of the Corporation shall be transferable or

assignable only on the transfer books of the Corporation or by transfer agents designated to transfer shares of stock of the Corporation by the registered holder thereof, in person or by a duly authorized attorney, and, in the case of certificated shares, upon the surrender and cancellation of such certificates representing the shares to be transferred, properly endorsed and, if sought to be transferred by such attorney, accompanied by a written power of attorney to have the same transferred on the books of the Corporation. The Board of Directors shall have power and authority to make such other rules and regulations concerning the issuance, transfer and registration of certificates of stock as it may deem appropriate.

Section 2.9 Lost, Destroyed and Mutilated Certificates. If any holder of the shares of the Corporation in certificated form shall notify the Corporation of any loss, theft, destruction or mutilation of such certificate(s), the Chief Executive Officer may, in his or her discretion, cause one or more new certificate(s) for the same number of shares in the aggregate to be issued to such stockholder upon the surrender of the mutilated certificate, or upon delivery by the stockholder or such stockholder's legal representative of a bond, with or without surety, or such other agreement, undertaking or security as the Corporate Secretary shall determine is appropriate, to indemnify the Corporation against any claim that may be made against the Corporation on account of the alleged loss, theft or destruction of any such certificate(s) or the issuance of any new certificate(s).

Section 2.10 Fixing a Record Date. Except as otherwise provided in the Common Stock Certificate of Designation, or any other certificate of designation relating to any class of the Corporation's preferred stock, for the purpose of determining stockholders entitled to notice of, or to vote at, any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend, or in order to make a determination of stockholders for any other proper purpose, the Board of Directors may fix in advance a date as the record date for any such determination of stockholders, such date in any case to be not more than 70 days before the meeting or action requiring a determination of stockholders and, in the case of record dates for dividends on the preferred stock, subject to any additional limitations set forth in the related certificate of designation. If no record date is fixed for the determination of (i) stockholders entitled to notice of, or to vote at, a meeting of stockholders or (ii) stockholders entitled to receive payment of a dividend, the date on which notices of the meeting are first mailed or otherwise given or the date on which the resolution of the Board of Directors declaring such dividend is adopted, as the case may be, shall be the record date for such determination of stockholders. When a determination of stockholders entitled to vote at any meeting of stockholders has been made as provided in this Section, such determination shall apply to any adjournment thereof unless the Board of Directors fixes a new record date, which it shall do if the meeting is adjourned to a date more than 120 days after the date fixed for the original meeting.

- **Section 2.11** Record of Stockholders. A record shall be kept of the names of the persons and entities owning the stock represented by each share of stock of the Corporation, the number of shares represented by each certificate or information statement issued in respect of noncertificated shares and the dates of issuance thereof in a form that permits the preparation of a list of the names and addresses of all stockholders in alphabetical order showing the number of shares held by each.
- Section 2.12 Ownership of Shares. The Corporation and any agent thereof may deem and treat the holder of a share or shares of stock, as shown in the Corporation's books and records, as the absolute owner of such share or shares of stock for the purpose of receiving payment of dividends in respect of such share or shares of stock and for all other purposes whatsoever, and neither the Corporation nor any agent thereof shall be affected by any notice to the contrary. All payments made to or upon the order of any such person shall be valid and, to the extent of the sum or sums so paid, effectual to satisfy and discharge liabilities for moneys payable by the Corporation on or with respect to any such share or shares of stock.
- **Section 2.13** Share Options and Other Instruments. The Corporation may issue rights, options or warrants for the purchase of shares or other securities of the Corporation, subject to stockholder approval to the extent required by applicable laws, regulations or listing standards. The Board of Directors may authorize the issuance of rights, options or warrants and determine the terms upon which the rights, options or warrants are issued, including, without limitation, the consideration for which the shares or other securities are to be issued. The authorization for the Corporation to issue such rights, options or warrants constitutes authorization of the issuance of the shares or other securities for which the rights, options or warrants are exercisable.

ARTICLE 3 - MEETINGS OF THE STOCKHOLDERS

Section 3.1 <u>Annual Meetings.</u>

- (a) An annual meeting of the stockholders, for the election of directors and for the transaction of such other business as may properly come before the meeting, shall be held on such date, at such time and at such place as the Board of Directors shall each year fix, which date shall be within 15 months after the immediately preceding meeting of stockholders. The failure to hold an annual meeting at the time stated in, or fixed in accordance with, these Bylaws shall not affect the validity of any corporate action.
- (b) Upon notice to the Corporation, any stockholder of the Corporation entitled to participate in an annual meeting may petition the United States District Court for the district within which the Corporation's principal office is located to order an annual meeting of stockholders if an annual meeting has not been held within 15 months after the Corporation's immediately preceding annual meeting. The court may fix the time and place of the meeting, determine the shares entitled to notice of and to vote at the

meeting, prescribe the form and content of the meeting notice and enter other orders necessary to accomplish the purpose or purposes of the meeting.

Section 3.2 Special Meetings.

- (a) Special meetings of the stockholders, for any purpose or purposes prescribed in the notice of the meeting, may be called by a majority of the directors then in office or the Chairman of the Board ("Chairman"), and shall be held on such date, at such time and at such place as they or he or she shall fix.
- Special meetings of the stockholders, for any purpose or purposes prescribed in the notice of the meeting, may also be called by the Corporate Secretary upon the written request of the holders of at least a majority of all shares of voting stock entitled to vote; provided, however, a special meeting of the stockholders for the purposes of Section 4.16 of these Bylaws must be requested by the holders of at least one-third of all shares of voting stock entitled to vote. A special meeting of the stockholders called pursuant to the preceding sentence shall be referred to in these Bylaws as a "Stockholder Requested Special Meeting". A Stockholder Requested Special Meeting shall be held on such date, at such time and at such place as determined by the Corporate Secretary as soon as is reasonably practicable following the Corporation's receipt of a written request that is in compliance with this Section 3.2 (the "Delivery Date") and, unless the penultimate paragraph of this Section applies, within 75 calendar days of the Delivery Date. The record date for determining stockholders entitled to request for a Stockholder Requested Special Meeting shall be the later of (i) the date the first stockholder signs the request, or (ii) the earliest record date permitted by Section 2.10 of these Bylaws.

Upon notice to the Corporation, any stockholder of the Corporation who signed the request for a Stockholder Requested Special Meeting, may petition the United States District Court for the district within which the Corporation's principal office is located to order a special meeting of stockholders if a special meeting notice has not been held as provided herein. The court may fix the time and place of the meeting, determine the shares entitled to notice of and to vote at the meeting, prescribe the form and content of the meeting notice, and enter other orders necessary to accomplish the purpose or purposes of the meeting.

A written request of the holders of the voting stock made pursuant to this Section 3.2 must (i) be signed, dated and delivered to the Corporate Secretary at the principal executive offices of the Corporation by each stockholder making the request for a Stockholder Requested Special Meeting, (ii) identify the name and record address of each stockholder making the request for a Stockholder Requested Special Meeting, (iii) include a brief description of the business desired to be brought before such Stockholder Requested Special Meeting, the reasons for conducting such business, a statement of specific purpose(s) of the meeting and the matter proposed to be acted on

at it, which matter must be a proper subject for stockholder action, (iv) shall be accompanied by documentation to verify the class and number of shares of the Corporation that are beneficially owned by each stockholder making such request in accordance with the Securities Exchange Act of 1934, as amended (the "Exchange Act"), specifically Rule 14a-8(b)(2) of Regulation 14A (Solicitation of Proxies) (the "Proxy Rules Requirement"), (v) contain a representation that each stockholder submitting the request intends to appear in person or by proxy at the Stockholder Requested Special Meeting to transact the business specified, and (vi) contain a representation that each stockholder submitting the request intends to continue ownership of shares of the voting stock through the date of the Stockholder Requested Special Meeting. The Corporation may require any stockholder making such request to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of stockholders to make a written request pursuant to this Section 3.2. Failure of the stockholders who sign the request for a Stockholder Requested Special Meeting to comply with the representations identified in (v) and (vi) above, shall be deemed to constitute a revocation of such request. Any stockholder who submitted a request for a Stockholder Requested Special Meeting may revoke such request at any time by written revocation delivered to the Corporate Secretary at the principal executive offices of the Corporation.

The Corporate Secretary shall not be required to call a Stockholder Requested Special Meeting if (i) the Board of Directors calls an annual or special meeting of stockholders to be held not later than 75 days after the Delivery Date and the purpose(s) of such meeting includes the purpose(s) specified by the requisite number of stockholders in the special meeting request(s), or (ii) an annual or special meeting was held not more than 12 months before the Delivery Date, which included the purpose(s) specified by the requisite number of stockholders in the special meeting request(s), with such determination being made in good faith by the Board of Directors. In determining whether a request for a Stockholder Requested Special Meeting has been submitted by stockholders holding the requisite number of shares of voting stock, there may be excluded from the computation, the shares of the voting stock owned by any stockholder who has signed the request for a Stockholder Requested Special Meeting at any time during the 2 calendar years preceding the Delivery Date and has failed to comply with the representations identified in (v) and (vi) above.

(c) At a special meeting, no business shall be transacted and no action shall be taken other than as stated in the notice of the meeting; provided that nothing herein shall prohibit the Board of Directors from submitting other matters to the stockholders at a Stockholder Requested Special Meeting.

Section 3.3 Notice of Meetings.

(a) Written or printed notice of the date, time and place of all meetings of the stockholders and, in the case of a special meeting, the purpose or purposes for which

the meeting is called, shall be given no fewer than 10 nor more than 60 days before the date on which the meeting is to be held, to each holder of voting stock entitled to vote at such meeting, except as otherwise provided in these Bylaws.

When a meeting is adjourned to another date, time or place, written notice need not be given of the adjourned meeting if the date, time or place thereof are announced at the meeting at which the adjournment is taken; provided, however, that, if the date of any adjourned meeting is more than 120 days after the date for which the meeting was originally noticed, or if a new record date is fixed for the adjourned meeting, written notice of the date, time or place of the adjourned meeting shall be given in conformity herewith.

(b) A stockholder's attendance at a meeting, whether in person or by proxy, (i) waives objection to lack of notice or defective notice of the meeting, unless the stockholder at the beginning of the meeting objects to holding the meeting or transacting business at the meeting; and (ii) waives objection to consideration of a particular matter at the meeting that is not within the purpose or purposes described in the meeting notice unless the stockholder objects to considering the matter when it is presented.

Notwithstanding the foregoing, no notice of a stockholders' meeting need be given to a stockholder if (i) an annual report and proxy statements for two consecutive annual meetings of stockholders or (ii) all, and at least two, checks for payment of dividends or interest on securities during a twelve-month period have been sent by first-class United States mail, addressed to the stockholder at his or her address as it appears on the stock transfer books of the Corporation, and have been returned undeliverable. The obligation of the Corporation to give notice of stockholders' meetings to any such stockholder shall be reinstated once the Corporation has received a new address for such stockholder for entry on its stock transfer books, a reaffirmation of the address appearing therein or a written consent from the stockholder to the receipt of notices by electronic transmission, specifying the address to which such notices should be electronically transmitted.

(c) For purposes of these Bylaws, "electronic transmission" means any form or process of communication, not directly involving the physical transfer of paper or other tangible medium, that (i) is suitable for the retention, retrieval, and reproduction of information by the recipient and (ii) is retrievable in paper form by the recipient through an automated process used in conventional commercial practice; provided, however, a notice or other communication may be in the form of an electronic transmission that cannot be directly reproduced in paper form by the recipient through an automated process used in conventional commercial practice if (x) the electronic transmission is otherwise retrievable in perceivable form and (y) the sender and the recipient have consented in writing to the use of such form of electronic transmission.

Section 3.4 Quorum.

- (a) At any meeting of the stockholders, the holders of a majority of all shares of voting stock entitled to vote at the meeting, present in person or by proxy, shall constitute a quorum, except as otherwise provided in this Section. If a quorum shall fail to attend a meeting, the chairman of the meeting or the holders of a majority of all shares of voting stock entitled to vote at the meeting who are present, in person or by proxy, may adjourn the meeting to another date, time or place.
- (b) Once a share of stock is represented for any purpose at a meeting, it is deemed present for quorum purposes for the remainder of the meeting and for any adjournment of that meeting unless a new record date is or shall be set for that adjourned meeting.
- **Section 3.5** Organization. The Chairman or, in the absence of the Chairman, another director designee of the Board of Directors or, in the absence of the Chairman and such a designation, the Chief Executive Officer or in his or her absence, the President or, in the absence of both such officers, such person as may be chosen by the holders of a majority of all shares of voting stock entitled to vote who are present, in person or by proxy, shall call to order any meeting of the stockholders and shall act as chairman of the meeting. In the absence of the Corporate Secretary, the secretary of the meeting shall be such person as the chairman of the meeting appoints.

Section 3.6 Conduct of Business.

- (a) The chairman of any meeting of stockholders shall determine the order of business and shall have the authority to establish rules for the conduct of the meeting, including such regulation of the manner of voting and the conduct of discussion, restrictions on attendance at a meeting so long as stockholders or their proxies are not excluded, and adjournment of the meeting to be reconvened at a later date, as seem to him or her in order and not inconsistent with these Bylaws.
- (b) No business shall be brought before any meeting except in accordance with the procedures set forth in these Bylaws, and the rules, if any, established by the chairman of the meeting for the conduct of the meeting; provided, however, that nothing in this Section shall be deemed to preclude discussion by any stockholder of any business properly brought before such meeting.
- (c) At any meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. The chairman of any meeting shall, if the facts warrant, determine that business was not properly brought before the meeting, and, if the chairman should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be conducted. At any adjourned meeting, any business may be transacted and any action taken which might have been transacted or taken at the original meeting.

- To be properly brought before an annual meeting, business must be specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors, or otherwise brought before the meeting by or at the direction of the Board of Directors, or otherwise properly brought before the meeting by a stockholder. In addition to any other applicable requirements, including, without limitation, Rule 14a-8 under the Exchange Act, if applicable, for business other than a director nomination to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Corporate Secretary at the principal executive office of the Corporation. (The exclusive procedures for stockholders to make nominations for the election of directors at any annual meeting are set forth in Section 4.3.) The content of the stockholders' notice must comply with the requirements of this Section 3.6. The requirements of this Section 3.6 shall apply to any business a stockholder wishes to bring before an annual meeting, whether under Rule 14a-8 under the Exchange Act or otherwise. To be timely, a stockholder's notice must be received at the principal office of the Corporation no fewer than 75 days prior to such annual meeting. In the event that fewer than 90 days' notice or prior public disclosure of the date of such annual meeting is given or made to stockholders, notice by the stockholder, to be timely, must be so received not later than the close of business on the 15th day following the day on which such notice of the date of such annual meeting was mailed or such public disclosure was made, whichever first occurs. A written notice must (i) be signed, dated and delivered to the Corporate Secretary at the principal executive offices of the Corporation by the stockholder, (ii) identify the name and record address of the stockholder submitting the notice, (iii) include a brief description of the business, which must be a proper subject for stockholder action, desired to be brought before such annual meeting and the reasons for conducting such business, (iv) shall be accompanied by documentation to verify the class and number of shares of the Corporation that are beneficially owned by the stockholder submitting such notice in accordance with the Exchange Act, specifically the Proxy Rules Requirement, (v) contain a representation that the stockholder submitting the notice intends to appear in person or by proxy at the annual meeting to transact the business specified, (vi) contain a representation that the stockholder submitting the notice intends to continue ownership of shares of the voting stock through the date of the annual meeting, and (vii) set forth any material interest of the stockholder in the matter identified in (iii) above. The Corporation may require any stockholder submitting such notice to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of stockholders to give a written notice pursuant to this Section 3.6.
- (e) At a special meeting, no business shall be transacted and no action shall be taken other than as stated in the notice of the meeting; provided that nothing herein shall prohibit the Board of Directors from submitting other matters to the stockholders at a Stockholder Requested Special Meeting.

(f) To the extent authorized by the Board of Directors with respect to any meeting, stockholders may participate in such a meeting by use of any means of communication by which all stockholders participating may simultaneously hear each other during the meeting including, without limitation, by use of internet accessible electronic meeting facilities. A stockholder participating in a stockholders' meeting by such means is deemed to be present in person at the meeting. Authorization of such electronic participation in such a meeting shall not eliminate the requirement that the meeting take place at a physical location at which stockholders may attend the meeting in person.

Section 3.7 Voting.

- (a) Except as otherwise provided in Section 8.1(c) of these Bylaws, each stockholder shall have one vote for every share of stock entitled to vote which is registered in his or her name on the record date for the meeting.
- (b) Subject to the determination of the Board of Directors of the Corporation to authorize action by written consent under Section 3(b) of the Common Stock Certificate of Designation, votes shall be cast in person or by proxy at annual or special meetings of the holders of the shares of voting stock entitled to vote. All voting, including the election of directors, may be by a voice vote; provided, however, that upon demand by a holder of voting stock entitled to vote or his or her proxy or in the discretion of the chairman of the meeting, a vote shall be taken by ballots, each of which shall state the name of the stockholder or proxy voting and such other information as may be required under the rules of conduct established for the meeting.
- (c) The Corporation shall appoint one or more inspectors to oversee, determine and certify attendance and results of any voting at the meeting and make a written report of the inspector's determinations. Each inspector, before entering upon the discharge of his duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspectors may appoint or retain other persons or entities to assist the inspectors in the performance of their duties.
- (d) All elections for a seat on the Board of Directors shall be determined as provided by Section 4.5 of these Bylaws, and on all other matters, action shall be approved if the votes cast favoring the action exceed the votes cast opposing the action, unless a greater vote is prescribed by the Board of Directors or other person proposing the matter or otherwise required by these Bylaws.
- (e) An abstention or an election by a stockholder not to vote on an action because of failure to receive voting instructions from the beneficial owner of the shares shall not be considered a vote cast.
 - (f) Stockholders shall not vote their shares cumulatively.

Section 3.8 Voting Entitlement of Shares.

- (a) Shares standing in the name of another corporation may be voted by such officer, agent or proxy as the bylaws of such corporation may prescribe, or, in the absence of such provision, as the board of directors of such corporation may determine.
- (b) Shares standing in the name of a partnership may be voted by any partner. Shares standing in the name of a limited liability company may be voted as the articles of organization or an operating agreement may prescribe, or in the absence of any such provision as the managers, or if there are no managers, the members of the limited liability company may determine.
- (c) Shares held by two or more persons as joint tenants or tenants in common or tenants by the entirety may be voted by any of such persons. If more than one of such tenants vote such shares, the vote shall be divided among them in proportion to the number of such tenants voting.
- (d) Shares held by an administrator, executor, guardian, committee or curator representing the holder of voting stock may be voted by such person without a transfer of such shares into such person's name. Shares standing in the name of a trustee may be voted by the trustee, but no trustee shall be entitled to vote shares held by him or her without a transfer of such shares into the trustee's name.
- (e) Shares standing in the name of a receiver or a trustee in proceedings under the applicable bankruptcy laws may be voted by such person. Shares held by or under the control of a receiver or a trustee in proceedings under the applicable bankruptcy laws may be voted by such person without the transfer thereof into his or her name if authority to do so is contained in an order of the court by which such person was appointed.
- (f) Nothing herein contained shall prevent trustees or other fiduciaries holding shares registered in the name of a nominee from causing such shares to be voted by such nominee as the trustee or other fiduciary may direct. Such nominee may vote shares as directed by a trustee or other fiduciary without the necessity of transferring the shares to the name of the trustee or other fiduciary.
- (g) When shares are held by more than one fiduciary, the shares shall be voted as determined by a majority of such fiduciaries, except that: (i) if they are equally divided as to a vote, the vote of shares shall be divided equally and (ii) if only one of such fiduciaries is present in person or by proxy at a meeting, such fiduciary shall be entitled to vote all the shares.
- (h) A holder of voting stock whose shares are pledged shall be entitled to vote such shares until the shares have been transferred into the name of the pledgee, and thereafter the pledgee shall be entitled to vote the shares so transferred.

Section 3.9 Proxies.

- (a) At any meeting of the stockholders, every holder of voting stock entitled to vote may vote in person or by proxy. A stockholder or the stockholder's agent or attorney-in-fact may appoint a proxy to vote or otherwise act for the stockholder by signing an appointment form or by an electronic transmission. An electronic transmission shall contain or be accompanied by information from which one can determine that the stockholder, the stockholder's agent or the stockholder's attorney-infact authorized the transmission. Any copy, facsimile telecommunication or other reliable reproduction of the writing or transmission created pursuant to this Subsection may be substituted or used in lieu of the original writing or transmission for any and all purposes for which the original writing or transmission could be used, provided that such copy, facsimile telecommunication or other reproduction shall be a complete reproduction of the entire original writing or transmission. No proxy shall be voted after eleven months from its date, unless a longer period is expressly provided in the appointment form or electronic transmission.
- (b) An appointment of a proxy is revocable unless the appointment form or electronic transmission states that it is irrevocable and the appointment is coupled with an interest. An appointment made irrevocable in accordance with the immediately preceding sentence is revoked when the interest with which it is coupled is extinguished.
- (c) The death or incapacity of the stockholder appointing a proxy does not affect the right of the Corporation to accept the proxy's authority unless notice of the death or incapacity is received by the Corporate Secretary or other officer or agent authorized to tabulate votes before the proxy exercises his or her authority under the appointment.
- (d) Subject to Section 3.8 of these Bylaws and to any express limitation on the proxy's authority stated in the appointment form or electronic transmission, the Corporation is entitled to accept the proxy's vote or other action as that of the stockholder making the appointment.
- (e) Any fiduciary who is entitled to vote any shares may vote such shares by proxy.

Section 3.10 Stockholders' List.

(a) A complete list of stockholders entitled to vote at any meeting of stockholders, showing the address of each such stockholder and the number of shares registered in his or her name, shall be open to the examination of any such stockholder, for any purpose germane to the meeting, during ordinary business hours for a period of at least 10 days prior to the meeting, either at the principal office of the Corporation or at the office of its transfer agent.

- (b) The stockholders' list shall also be kept at the place of the meeting during the whole time thereof and shall be open to the examination of any such stockholder who is present. This list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each of them.
- (c) If the requirements of this Section have not been substantially complied with, the meeting shall, on the demand of any stockholder in person or by proxy, be adjourned until such requirements are complied with. Refusal or failure to prepare or make available the stockholders' list shall not affect the validity of action taken at the meeting prior to the making of any such demand, but any action taken by the stockholders after the making of any such demand shall be invalid and of no effect.

ARTICLE 4 - BOARD OF DIRECTORS

Section 4.1 <u>General Powers.</u> Subject to the limitations of law and regulation, the Board of Directors shall determine the general policies that govern the operations of the Corporation, and the Corporation shall be under the direction of the Board of Directors.

Section 4.2 Number, Qualification and Term of Office.

- (a) The Board of Directors of the Corporation shall consist of 13 persons, or such other number as the Director of the Federal Housing Finance Agency determines appropriate who shall be elected annually by the stockholders. The Board of Directors shall at all times have as members at least one person from the homebuilding industry, at least one person from the mortgage lending industry, at least one person from the real estate industry, and at least one person from an organization that has represented consumer or community interests for not less than two years or one person who has demonstrated a career commitment to the provision of housing for low-income households.
- (b) Each member of the Board of Directors shall be elected for a term ending on the date of the next annual meeting of the stockholders.
- (c) The Board of Directors shall establish standards and qualifications relating to independence from management and may establish other qualifications for service on the Board of Directors, including limitations on length of service and age, as required under, or consistent with, applicable laws, regulations and stock exchange listing standards.

Section 4.3 Nominations.

(a) Only persons who are nominated in accordance with the following procedures shall be eligible for election as directors at any annual meeting of stockholders. Nominations of persons for election to the Board of Directors of the

Corporation at any annual meeting of stockholders may be made only by (i) the Board of Directors, or (ii) by any stockholder of the Corporation entitled to vote for the election of directors at the meeting who complies with the notice procedures set forth in this Section; this clause (ii) shall be the exclusive means for a stockholder to make nominations for election of directors at any annual meeting of stockholders. (The exclusive procedures for stockholders to bring business other than nominations before a stockholders' meeting are set forth in Section 3.6.) Such nominations made by a stockholder entitled to vote in the election of directors shall be made pursuant to timely notice in writing to the Corporate Secretary at the principal executive office of the Corporation. To be timely, a stockholder's notice shall be received no fewer than 75 days prior to the meeting. In the event that fewer than 90 days' notice or prior public disclosure of the date of the meeting is given or made to stockholders, notice by the stockholder, to be timely, must be so received not later than the close of business on the 15th day following the day on which such notice of the date of the meeting was mailed or such public disclosure was made, whichever first occurs. Such stockholder's notice to the Corporate Secretary shall set forth (a) as to each person who the stockholder proposes to nominate for election or reelection as a director, (i) the name, age, business address and residential address of the person, (ii) the principal occupation or employment of the person, (iii) a description of all arrangements or understandings between the stockholder and each nominee and any other person(s) (naming such person(s)) pursuant to which arrangements or understandings the nominations(s) are to be made by the stockholder and (iv) such other information regarding each nominee proposed by such stockholder as is required to be disclosed in solicitations for proxies for election of directors pursuant to Section 14(a) of the Exchange Act; and (b) as to the stockholder submitting the notice, the notice must (i) be signed and dated by the stockholder, (ii) identify the name and record address of the stockholder submitting the notice, (iii) be accompanied by documentation to verify the class and number of shares of the Corporation that are beneficially owned by the stockholder submitting such notice in accordance with the Exchange Act, specifically the Proxy Rules Requirement, (iv) contain a representation that the stockholder submitting the notice intends to appear in person or by proxy at the annual meeting to make the nomination(s) the stockholder has proposed, and (v) contain a representation that the stockholder submitting the notice intends to continue ownership of shares of the voting stock through the date of the annual meeting. The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as director of the Corporation and such nominee's independence. No person shall be eligible for election as a director of the Corporation unless nominated in accordance with the procedures set forth in this Section.

(b) The chairman of the meeting shall, if the facts warrant, determine that a nomination was not made in accordance with the foregoing procedures, and, if the chairman should so determine, he or she shall so declare to the meeting and the defective nomination shall be disregarded.

Section 4.4 <u>Vacancies.</u> Any seat on the Board of Directors that becomes vacant after the annual election of the directors may be filled only by the Board of Directors, but only for the unexpired portion of the term. If the directors remaining in office constitute fewer than six directors, the remaining directors may fill such vacant seats on the Board of Directors as may exist by the affirmative vote of a majority of such remaining directors.

Section 4.5 Elections.

- (a) Each director nominated for a seat on the Board of Directors of the Corporation shall be elected only if he or she receives a majority of the votes cast with respect to his or her election at the annual meeting of stockholders, provided that if it is determined that the number of persons properly nominated to serve as directors of the Corporation exceeds the number of directors to be elected (a "contested election"), the directors shall be elected by a plurality of the votes of the shares represented at the meeting and entitled to vote on the election of directors. A "majority of the votes cast" means that the number of votes cast "for" a director must exceed the number of votes cast "against" that director.
- Following any uncontested election, any incumbent director who was a nominee for a seat and who did not receive a majority of the votes cast by the stockholders shall promptly tender to the committee responsible for nominating and governance matters his or her offer of resignation for consideration by the Board of Directors. Within 60 days following certification of the election results, the committee of the Board of Directors responsible for nominating and governance matters shall recommend to the Board of Directors the action to be taken with respect to such offer of resignation. Within 90 days following certification of the election results, the Board of Directors shall act on the offered resignation. In determining whether or not to accept the offered resignation, the Board of Directors shall consider any recommendation of the committee responsible for nominating and governance matters, the factors considered by that committee and any additional information and factors that the Board of Directors believes to be relevant. No director who submits his or her resignation pursuant to this Section 4.5 shall participate in the deliberations or decisions of the committee responsible for nominating and governance matters or the Board of Directors regarding such director's resignation.
- (c) If the submitted resignation is not accepted by the Board of Directors, the director shall continue to serve until the next annual meeting and until his or her successor is duly elected, or his or her death, resignation, retirement or removal in accordance with these Bylaws, applicable law or regulation, whichever event shall first occur. If a director's resignation is accepted by the Board of Directors, or if a nominee for director who is not an incumbent director is not elected by the majority of the votes cast by the stockholders, then the Board of Directors, in its sole discretion, may fill any resulting vacancy in accordance with Section 4.4 of these Bylaws.

- **Section 4.6** <u>Chairman.</u> Each year at the first meeting of the Board of Directors following the annual stockholders' meeting, the Board of Directors shall elect from among its members a person to serve as Chairman of the Board. The Chairman shall be a member of the Board of Directors and shall preside at all meetings of the Board of Directors at which he or she is present.
- Section 4.7 <u>Lead Director.</u> If the Chairman is not independent from management under standards established by the Board of Directors pursuant to Section 4.2 of these Bylaws, those directors who are not employed by the Corporation shall elect from among themselves a person who is independent from management to serve as Lead Director. That election, if necessary, shall take place each year at the first meeting of the Board of Directors following the annual stockholders' meeting and at any other time that the Board of Directors lacks a Lead Director who is independent from management. If the Chairman is independent from management, the Chairman shall serve as Lead Director unless the directors who are not employed by the Corporation elect from among themselves another director who is independent from management to serve as Lead Director. The Lead Director shall preside at all meetings of the directors who are not employees of the Corporation and of the directors who are independent from management at which he or she is present and shall perform such other duties as may be assigned by those directors.
- **Section 4.8** Regular Meetings. A regular meeting of the Board of Directors shall be held as soon as practicable after adjournment of the annual meeting of stockholders at such place as the Board of Directors may designate by resolution and without other notice than such resolution. The Board of Directors may provide, by resolution, for the date, time and place of additional regular meetings without other notice than such resolution.

Section 4.9 Special Meetings.

- (a) Special meetings of the Board of Directors may be called by a majority of the directors then in office or by the Chairman and shall be held on such date, at such time and at such place as they or he or she shall fix. Notice of the date, time and place of each such special meeting shall be given in accordance with Section 7.1 of these Bylaws to each director by (i) written notice given by mail, private courier or in person not less than 48 hours before the meeting, (ii) oral notice given in person or by telephone not less than 24 hours before the meeting or (iii) electronic transmission sent not less than 24 hours before the meeting. Unless otherwise indicated in the notice thereof, any and all business may be transacted at a special meeting.
- (b) A director's attendance at or participation in a meeting waives any required notice to him or her of the meeting unless the director at the beginning of the meeting or promptly upon arrival objects to holding the meeting or transacting business at the meeting and does not thereafter vote for or assent to action taken at the meeting.

Section 4.10 Quorum. At any meeting of the Board of Directors, a quorum shall consist of the greater of (i) one-third of the fixed number of directors or the prescribed number within a variable range of directors, as applicable, or (ii) a majority of the directors then in office. If a quorum shall fail to attend any meeting, a majority of those present may adjourn the meeting to another date, time or place, without further notice or waiver thereof. Members may not be represented by proxy at any meeting of the Board of Directors.

Section 4.11 <u>Participation in Meetings.</u> Members of the Board of Directors, or of any committee thereof, may participate in a meeting of the Board of Directors or of such committee by any means of communication by which all members participating in the meeting can simultaneously hear each other during the meeting and such participation shall constitute presence in person at such meeting.

Section 4.12 Conduct of Business.

- (a) At each meeting of the Board of Directors, the Chairman shall preside or, in the absence of the Chairman, a director selected by the Board of Directors. The Corporate Secretary or an Assistant Secretary designated by the Corporate Secretary shall act as secretary for the meeting, unless the Chairman or director presiding at the meeting appoints another individual present at the meeting to act as secretary for the meeting.
- (b) At any meeting of the Board of Directors, business shall be transacted in such order and manner as the Board of Directors may from time to time determine, and all matters shall be determined by the vote of a majority of the directors present at a meeting at which a quorum is present, except as otherwise provided in these Bylaws or in the Virginia Stock Corporation Act. A director shall not vote by proxy. A director present at a meeting is presumed to assent to actions unless the director objects at the beginning of the meeting, or promptly upon his or her arrival, to holding the meeting or transacting specified business at the meeting or the director votes against or abstains from the action and such objection, dissent or abstention is entered into the minutes of the meeting.
- (c) Action may be taken by the Board of Directors without a meeting if each director signs a consent describing the action to be taken and delivers it to the Corporation. A director's consent may be withdrawn by a revocation signed by the director and delivered to the Corporation prior to delivery to the Corporation of unrevoked written consents signed by all the directors. Such written consent and the signing thereof may be accomplished by one or more electronic transmissions.
- (d) Action taken under the preceding paragraph shall be effective when the last director signs the consent unless the consent specifies a different effective date, in which event such action shall be effective as of the date specified therein, provided that the consent contains the date of execution of each director.

(e) A consent signed under this Section shall have the effect of action taken at a meeting of the Board of Directors and may be described as such in any document.

Section 4.13 Reimbursement and Compensation of Directors. Pursuant to resolution of the Board of Directors, directors, as such, may receive fixed fees and other compensation for their services as directors, including, without limitation, their services as members of committees of the Board of Directors and may receive reimbursement of expenses incurred in respect of rendering such services; except that any member of the Board of Directors who is a full-time officer or employee of the federal government or full-time officer or employee of the Corporation shall not receive compensation for services as a member of the Board of Directors or as a member of any committee of the Board of Directors.

Section 4.14 Committees of the Board of Directors.

- The Board of Directors may, from time to time, designate committees of the Board of Directors (and subcommittees of those committees), with such delegable powers and duties as it thereby confers, to serve at the pleasure of the Board of Directors and shall, for those committees and any others provided for herein, elect at least two directors to serve as the members. Any committee or subcommittee so designated may exercise the power and authority of the Board of Directors to the extent any resolution of the Board of Directors shall so provide, except that a committee or subcommittee may not: (i) approve or recommend to stockholders action that these Bylaws require be approved by stockholders; (ii) fill vacancies on the Board of Directors or on any of its committees; (iii) adopt, amend or repeal these Bylaws; (iv) approve a plan of merger not requiring stockholder approval; (v) authorize or approve a distribution, except according to a general formula or method prescribed by the Board of Directors; or (vi) authorize or approve the issuance or sale or contract for sale of shares, or determine the designation and rights, preferences and limitations of a class or series of shares, except that the Board of Directors may authorize a committee or the Chairman to do so subject to such limits, if any, as may be prescribed by the Board of Directors.
- (b) Unless otherwise provided by Board of Directors or committee or subcommittee resolution, the provisions of these Bylaws on date, time and place of, and notice required for, meetings of the Board of Directors shall govern committees of the Board of Directors (and their subcommittees). A majority of directors then appointed as members of a committee or a subcommittee shall constitute a quorum and all matters shall be determined by a majority vote of the members present at a meeting at which a quorum is present. Action may be taken by a committee (or a subcommittee) without a meeting if each member signs a consent describing the action to be taken and delivers it to the Corporation. A member's consent may be withdrawn by a revocation signed by the member and delivered to the Corporation prior to delivery to the Corporation of unrevoked written consents signed by all the members. Such written consent and the

signing thereof may be accomplished by one or more electronic transmissions. All minutes of committee or subcommittee meetings and unanimous consents of action taken by a committee or a subcommittee without a meeting shall also be submitted to the Board of Directors.

- (c) The Board of Directors shall designate committees responsible for overseeing the Corporation's financial statements and relationship with its independent auditor, executive compensation matters and governance and nominating matters. The membership of each of those committees shall consist solely of directors who are independent from management under the standards established pursuant to Section 4.2 of these Bylaws and shall comply with applicable laws, regulations and listing standards.
- **Section 4.15** Resignation. Any director may resign at any time by delivering a written resignation to the Board of Directors, the Chairman or the Corporate Secretary. A resignation shall be effective upon delivery unless the notice specifies a later effective date. If a resignation is made effective at a later date, the Board of Directors may fill the pending vacancy before the effective date if the successor does not take office until the effective date.
- **Section 4.16** Removal of Directors. At a Stockholder Requested Special Meeting called expressly for that purpose, any director may be removed, but only with cause, by a vote of the holders of a majority of the voting stock then entitled to vote at an election of directors.
- Section 4.17 <u>Termination of Voluntary Registration of Common Stock.</u>
 The Corporation shall take no action to terminate the registration of the Corporation's common stock under Section 12(g) of the Exchange Act, unless such action has been approved by unanimous action of all members of the Board of Directors then in office.

ARTICLE 5 – OFFICERS

Section 5.1 Officers of the Corporation. There shall be a Chief Executive Officer of the Corporation and a Senior Vice President – General Auditor. Other officers of the Corporation may include a President, a Chief Operating Officer, a Chief Compliance Officer, a Chief Enterprise Risk Officer, one or more Vice Presidents (any one or more of whom may be designated Executive Vice President or Senior Vice President and may be given other descriptive titles), a Corporate Secretary and all other officers or assistant officers deemed necessary and desirable for the conduct of the Corporation's business. Any of the above offices may be held by the same person, except that the office of the Corporate Secretary may not be held by the same person that holds the office of Chief Executive Officer, President, Chief Operating Officer or Senior Vice President – General Auditor.

Section 5.2 Appointment and Term.

- (a) The Board of Directors shall elect the Chief Executive Officer.
- (b) The Audit Committee of the Board of Directors shall elect the Senior Vice President General Auditor.
- (c) Except as otherwise determined by the Board of Directors or provided herein, the Chief Executive Officer shall appoint all additional officers at the Executive Vice President and Senior Vice President level or above.
- (d) Except as otherwise determined by the Board of Directors or provided herein, the Chief Executive Officer or his or her designee(s) (each individually an "Appointing Officer" and collectively the "Appointing Officers") shall appoint all officers at the Vice President level and below, other than those identified in Sections 5.2 (a), (b) and (c).
- (e) Except as provided herein, the appointment of a President and/or Chief Operating Officer, if one or both of such positions are to be filled, shall be subject to prior review by the Board of Directors.
- (f) Any appointment by an Appointing Officer under this section is subject to the legal, regulatory or supervisory limitations, requirements and approvals that apply to appointments by the Board of Directors. Each officer elected by the Board of Directors or appointed by an Appointing Officer shall hold office until his or her successor is elected or appointed and qualified or until his or her death, resignation or removal as provided in this Article 5. Election or appointment of an officer shall not, in and of itself, create any contract rights in the officer against the Corporation.

Section 5.3 Removal, Resignation, Vacancy.

- (a) Any officer may be removed, with or without cause, by a vote of the Board of Directors. The Senior Vice President General Auditor may be removed, with or without cause, by a vote of the Audit Committee. Except as otherwise determined by these Bylaws or the Board of Directors, an Appointing Officer may remove, with or without cause, any officer he or she may appoint.
- (b) Any officer may resign at any time by delivering a notice of resignation to the Corporation. A resignation shall be effective upon delivery unless the notice specifies a later effective time. If a resignation is made effective at a later time, the Board of Directors or the appropriate Appointing Officer may fill the pending vacancy before the effective time if the successor does not take office until the effective time. A vacancy in any office shall be filled in the manner prescribed in these Bylaws for election or appointment to such office.

- **Section 5.4** <u>Compensation.</u> The compensation of all officers of the Corporation shall be fixed by or under the authority of the Board of Directors. No officer shall be prevented from receiving such compensation by reason of the fact that such officer is also a director of the Corporation.
- **Section 5.5** <u>Duties.</u> The officers of the Corporation shall have such powers and duties as are provided for in these Bylaws as well as such other authority as provided by the Board of Directors or, in the case of the officers other than the Chief Executive Officer, by the appropriate Appointing Officer.
- Section 5.6 <u>Chief Executive Officer.</u> The Chief Executive Officer of the Corporation shall be primarily responsible for the implementation of the policies, orders and resolutions of the Board of Directors. Subject to the direction of the Board of Directors, he or she shall have general charge of and responsibility for supervision of the business and affairs of the Corporation. The Chief Executive Officer may sign and execute in the name of the Corporation all certificates, contracts and instruments. The Chief Executive Officer may vote stock in other corporations, in person or by proxy, and shall perform such other duties of management as may be commonly incident to the office of chief executive or as may be prescribed by resolution or as otherwise may be assigned to the Chief Executive Officer by the Board of Directors.
- **Section 5.7** President. The President shall perform such duties as from time to time may be assigned by the Board of Directors or the Chief Executive Officer.
- **Section 5.8** <u>Chief Operating Officer.</u> The Chief Operating Officer shall perform such duties as from time to time may be assigned by the Board of Directors or the Chief Executive Officer.
- Section 5.9 <u>Senior Vice President General Auditor.</u> The Senior Vice President General Auditor shall report to, and be hired, supervised and terminated, if deemed appropriate, by the Audit Committee of the Board of Directors. The Senior Vice President General Auditor shall be responsible for examining and evaluating the adequacy and effectiveness of the Corporation's system of internal controls. The Senior Vice President General Auditor shall perform such other duties as from time to time may be assigned by the Audit Committee of the Board of Directors.
- **Section 5.10** <u>Vice Presidents.</u> The Corporation shall have one or more Vice Presidents, which may include Executive Vice Presidents or Senior Vice Presidents, elected or appointed as herein provided. Each such Vice President shall have such duties as from time to time may be assigned to him or her by the Board of Directors, the appropriate Appointing Officer or the supervising officer to whom a Vice President reports directly or that are commonly incident to such Vice President's office.

Section 5.11 Corporate Secretary. The Corporate Secretary shall keep the minutes of the meetings of the stockholders and of the Board of Directors and of committees of the Board of Directors and their subcommittees in books provided for that purpose; shall see that all notices of such meetings are duly given in accordance with the provisions of these Bylaws; may sign certificates of stock of the Corporation with the Chief Executive Officer; shall be custodian of the corporate seal; shall see that the corporate seal is affixed to all documents as appropriate; shall certify all documents pertaining to actions of the stockholders and the Board of Directors and any of its committees (and their subcommittees) and all other corporate documents and, in general, shall perform all duties and have all powers as may be commonly incident to the office of a secretary of a corporation, and such other duties as from time to time may be assigned to the Corporate Secretary by the Board of Directors or the Chief Executive Officer. The Corporate Secretary may appoint such Assistant Secretaries as he or she deems appropriate. The duties of the Corporate Secretary may be performed by one or more Assistant Secretaries.

Section 5.12 <u>Delegation of Authority.</u> Subject to the control of the Board of Directors, the functions delegated to the holder of a particular office pursuant to these Bylaws (the "Officer") shall be performed by such holder, or under his or her direction, by such individuals as may from time to time be delegated authority to perform such functions by the Officer. A person to whom a function is delegated by the Board of Directors may further delegate that function to another person under his or her direction to the extent that such person is permitted to do so by the original delegation to him or her by the Board of Directors.

ARTICLE 6 - INSPECTION OF RECORDS

Section 6.1 <u>Inspection of Records by Stockholders.</u>

- (a) The Corporation's Bylaws and all amendments thereto, all Board of Directors resolutions creating one or more classes or series of shares and minutes of all stockholders' meetings for the then most recent three years, all written communications to stockholders generally within the past three years (including all financial statements furnished for the past three years) and the names and business addresses of its current directors and officers shall be open to inspection at the Corporation's principal office during its regular business hours upon written request therefor, received by the Corporation at least five business days prior to the date such inspection is requested, from any person who is a stockholder.
- (b) Excerpts from the minutes of any meeting of the Board of Directors, records of any action of a committee of the Board of Directors (or a subcommittee of a committee) while acting in place of the Board of Directors on behalf of the Corporation, minutes of any meeting of the stockholders, and records of action taken by the Board of Directors without a meeting, to the extent not subject to inspection under Subsection

- (a), accounting records of the Corporation and the record of stockholders shall be open to inspection at a reasonable location specified by the Corporation during its regular business hours upon written request therefor, received by the Corporation at least five business days prior to the date such inspection is requested, provided that the request is made by a stockholder who has been a stockholder of record for at least six months immediately preceding such request or is the holder of record of at least five percent of all of the Corporation's outstanding shares, and provided further that (i) such request is made in good faith and for a proper purpose, (ii) such request describes with reasonable particularity the purpose of such request and the records to be inspected, and (iii) the records requested are directly connected with his or her purpose. The stockholder's written request shall be accompanied by (i) documentation to verify the class and number of shares of the Corporation that are beneficially owned by the stockholder in accordance with the Proxy Rules Requirements and (ii) proof of the stockholder's ownership when making a request under this Subsection (b).
- (c) Any inspection made pursuant to this Section may be made in person or by an agent or attorney and shall include the right to make copies, including copies through an electronic transmission if available and so requested by the stockholder. A request for any such inspection shall be served upon the Chief Executive Officer or the Corporate Secretary. This right of inspection is in addition to the stockholders' right to inspect the stockholders' list as provided in Section 3.10 of these Bylaws.
- (d) For purposes of this Section, stockholder includes a beneficial owner whose shares are held in a voting trust or by a nominee on the stockholder's behalf.

Section 6.2 <u>Inspection of Records by Directors.</u> A director of the Corporation is entitled to inspect and copy the books, records and documents of the Corporation at any reasonable time to the extent reasonably related to the performance of the director's duties as a director, including duties as a member of a committee or a subcommittee, but not for any other purpose or in any manner that would violate any duty to the Corporation.

ARTICLE 7 - NOTICES

Section 7.1 Notices.

(a) Except as otherwise specifically provided in these Bylaws, all notices required to be given to any stockholder, director, officer, employee or agent shall be in writing, in the English language (unless otherwise agreed by the sender and the recipient) and may in every instance be effectively given by hand delivery; by mail or commercial courier; or by electronic transmission. Where these Bylaws expressly permit oral notice, such notice may be communicated in person, by telephone, voice mail or by other electronic means. If these forms of personal notice are impracticable, notice may be communicated by a newspaper of general circulation in the area where the notice is intended to be given or by radio, television or other form of public

communication in the area where notice is intended to be given. In addition to the manner in which notices may be given under these Bylaws, notices may also be given as set forth in the Common Stock Certificate of Designation.

- effectively to stockholders or directors, any notice to stockholders or directors given by the Corporation under these Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder or the director to whom notice is given, provided that it shall not be necessary to obtain consent to send notices of meetings of directors to directors by electronic transmission except to the extent required under Section 3.3(c)(y) of these Bylaws. In the case of stockholders' meeting notices given to employee stockholders where such employees have regular access to electronic mail delivery in the course of their employment, consent to receipt of such notices by electronic transmission to the employee's employment related e-mail address shall be implied unless and until the employee specifies a different address to which notices should be electronically transmitted or requests delivery of notice in print.
- (c) Any consent to receive notices by electronic transmission shall be revocable by the stockholder or director by written notice to the Corporation. Any such consent shall be deemed revoked if (i) the Corporation is unable to deliver by electronic transmission two consecutive notices given by the Corporation in accordance with such consent and (ii) such inability becomes known to the Corporate Secretary or an Assistant Secretary of the Corporation or to the transfer agent or other person responsible for the giving of notice; provided however, that inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.
- (d) Any such notice shall be addressed to such stockholder, director, officer, employee or agent at his or her last known address as the same appears on the books of the Corporation, or in the case of a notice given by electronic transmission, shall be sent to the address at which the stockholder or director has consented to receive notice or, in the case of notices of directors' meetings delivered by electronic transmission, sent to an information processing system that the director has designated or uses for the purpose of receiving electronic transmissions.
- (e) Notice to a stockholder, if given by mail or physical delivery, is effective (i) upon deposit in the United States mail or (ii) when the physical delivery is actually received or left at the stockholder's address of record. Notice to persons other than stockholders, if given by physical delivery or mail, is effective at the earliest of the following (i) when it is actually received; (ii) five days after its deposit in the United States mail; (iii) on the date shown on the return receipt if sent by registered or certified mail, return receipt requested, and the receipt is signed by the addressee; (iv) in the case of notice to a director, when it is left at the director's residence or usual place of business; or (v) in the case of notice to the Corporation, when it is left at the

Corporation's principal place of business or at the Corporation's registered office when left with the Corporation's registered agent.

- (f) Oral notice is effective when communicated.
- (g) Notice given by electronic transmission to a stockholder, director or other person, who has consented to receive notices by such means, or whose consent is not required under Subsection (b) of this Section, is effective when received. An electronic transmission is received when (i) it enters an information processing system that the recipient has designated or uses for the purpose of receiving electronic transmissions or information of the type sent, and from which the recipient is able to retrieve the electronic transmission and (ii) it is in a form capable of being processed by that system. An electronic transmission is received even if no individual is aware of its receipt.
- (h) The Corporation shall be deemed to have delivered notice of an annual or special meeting of stockholders who share a common address as shown on the Corporation's current record of stockholders if (i) the Corporation delivers one meeting notice to the common address; (ii) the Corporation addresses the meeting notice (or a proxy statement, annual report, or notice of Internet availability of proxy materials containing such meeting notice)(collectively, the "Meeting Notice") to those stockholders sharing a common address either as a group (for example, "Jane Doe and Household" or "The Smith Family"), to each of them individually (for example, "Jane Doe, John Doe and Richard Doe") or to the stockholders in a form to which each of those stockholders has consented in writing; (iii) each of those stockholders consents, including any implied consent pursuant to Virginia Stock Corporation Act § 13.1-610.1(B), in accordance with procedures required by Rule 14a-3(e) of Regulation 14A (Solicitation of Proxies) under the Exchange Act of 1934, to the delivery of a single Meeting Notice to the stockholders' common address; (iv) the Corporation delivers a separate proxy card for each stockholder at the common address; and (v) the Corporation includes in the Meeting Notice an undertaking to deliver promptly, upon written or oral request, a separate copy of the Meeting Notice to a stockholder at a common address to which a single copy of the Meeting Notice was delivered. If a stockholder, orally or in writing revokes a consent to delivery of one Meeting Notice to a common address, the Corporation shall begin providing individual notices to the revoking stockholder no later than 30 days after the Corporation receives revocation of the stockholder's consent.
- **Section 7.2** <u>Written Waivers.</u> A written waiver of any notice, signed by a stockholder, director, officer, employee or agent, whether before or after the time of the event for which notice is to be given, shall be deemed equivalent to the notice required to be given to such stockholder, director, officer, employee or agent. Neither the business nor the purpose of any meeting need be specified in such a waiver.

ARTICLE 8 - INDEMNIFICATION AND LIMITATION OF LIABILITY

Section 8.1 Indemnification.

- Subject to the conditions set forth in Subsection (b) of this Section, each person who was or is made a party or is threatened to be made a party to any action, suit or proceeding, whether civil, criminal, administrative, arbitrative or investigative and whether formal or informal, including a derivative action or action brought by the Corporation (hereinafter, a "proceeding"), by reason of the fact that he or she is or was a director, officer or employee of the Corporation or is or was serving at the request of the Corporation as a director, officer, manager, partner, trustee, fiduciary, employee or agent of another corporation, limited liability company, partnership, joint venture, trust or other entity, including service with respect to an employee benefit plan (hereinafter, an "indemnitee"), shall be indemnified and held harmless by the Corporation against all liability (including the obligation to pay a judgment, settlement, penalty or fine, including any excise tax assessed with respect to an employee benefit plan) and expense (including attorneys' fees) reasonably incurred or suffered by such indemnitee in connection therewith, except such liabilities and expenses as are incurred because of the indemnitee's willful misconduct or knowing violation of the criminal law; provided, however, that the Corporation may not indemnify an indemnitee in connection with any proceeding charging improper personal benefit to the indemnitee, whether or not involving action in his or her official capacity, to the extent the indemnitee was adjudged liable on the basis that personal benefit was improperly received by the indemnitee. Such indemnification shall continue as to an indemnitee who has ceased to be a director, officer or employee and shall inure to the benefit of the indemnitee's heirs, executors and administrators.
- (b) Indemnification shall be made by the Corporation only as authorized in the specific case after a determination has been made as provided in Subsection (c) of this Section that the indemnitee met the relevant standard of conduct set forth in Subsection (a) of this Section. The termination of a proceeding by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, is not, of itself, determinative that the indemnitee did not meet the standard of conduct set forth in Subsection (a) of this Section.
- (c) The determination of whether the indemnitee met the standard of conduct set forth in Subsection (a) of this Section shall be made: (i) by the Board of Directors by a majority vote of a quorum consisting of directors not at the time parties to the proceeding; (ii) by a majority vote of a committee duly designated by the Board of Directors (in which designation directors who are parties may participate), consisting solely of two or more directors not at the time parties to the proceeding; (iii) by special legal counsel (1) selected by the Board of Directors or its committee in a manner prescribed in Subsection (c)(i) or (c)(ii) hereof, or (2) if a quorum of the Board of Directors cannot be obtained under Subsection (c)(i) hereof and a committee cannot be

designated under Subsection (c)(ii) hereof, selected by a majority vote of the full Board of Directors (in which selection directors who are parties may participate); or (iv) by the stockholders, provided, however, that shares owned by or voted under the control of directors who are at the time parties to the proceeding may not be voted on the determination.

- (d) Authorization of indemnification and evaluation as to reasonableness of expenses shall be made in the same manner as the determination that indemnification is permissible, as provided in Subsection (c) of this Section, provided however, that, if the determination is made by special legal counsel, authorization of indemnification and evaluation as to the reasonableness of expenses shall be made by those entitled under Subsection (c)(iii) of this Section to select such special legal counsel.
- (e) Notwithstanding any other provision of this Section, the Corporation shall indemnify a director or indemnitee who entirely prevails, on the merits or otherwise, in the defense of any proceeding to which the indemnitee was a party because he or she is or was director or officer or employee of the Corporation or was serving at the request of the Corporation as a director, officer, manager, partner, trustee, fiduciary, employee or agent of another corporation, limited liability company, partnership, joint venture, trust or other entity, including service with respect to an employee benefit plan, against reasonable expenses (including attorneys' fees) incurred by the indemnitee in connection with the proceeding.
- (f) Except as provided in Section 8.2 of these Bylaws with respect to proceedings to enforce rights to indemnification, the Corporation shall indemnify an indemnitee in connection with a proceeding (or part thereof) initiated by such indemnitee only if such proceeding (or part thereof) was authorized by the Board of Directors.
- (g) The rights to indemnification and liability limitation conferred in this Article 8 shall be deemed a contract right between an individual indemnitee and the Corporation, and any subsequent repeal or modification of these Bylaws shall not diminish the indemnitee's rights under this Article 8 with respect to any act or omission occurring before such amendment.
- (h) The indemnitee shall have the right to be paid by the Corporation the expenses reasonably incurred or suffered in defending any proceeding in advance of its final disposition (hereinafter, an "advancement of expenses"); provided, however, that an advancement of expenses shall be made (i) only upon delivery to the Corporation of a written statement by the indemnitee of the indemnitee's good faith belief that he or she has met the standard of conduct set forth in Subsection (a) of this Section, and (ii) only if the indemnitee furnishes to the Corporation a written undertaking, executed by or on behalf of such indemnitee, to repay any funds advanced if the indemnitee is not entitled to mandatory indemnification under Subsection (e) of this Section and it is

ultimately determined that such indemnitee did not meet the standard of conduct set forth in Subsection (a) of this Section. The undertaking required by provision (h)(ii) of this Subsection shall be an unlimited general obligation of the indemnitee but need not be secured and shall be accepted without reference to the financial ability of the indemnitee to make repayment.

- (i) The Corporation may, by action of its Board of Directors, provide indemnification to agents of the Corporation with the same scope and effect as the indemnification of indemnitees as provided in this Article 8.
- (j) The Chief Executive Officer is authorized to enter into contracts of indemnification with each indemnitee of the Corporation with respect to the indemnification provided in this Article 8 and renegotiate such contracts as necessary to reflect changing laws and business circumstances.

Section 8.2 Right of Indemnitee to Bring Suit. If a claim under Section 8.1 of these Bylaws is not paid in full by the Corporation within 90 days after a written claim has been received by the Corporation, except in the case of a claim for an advancement of expenses, in which case the applicable period shall be 20 days, the indemnitee may at any time thereafter apply to either the United States District Court for the district within which the Corporation's principal office is located or to the court where the proceeding is pending, if any, for an order directing the Corporation to make an advancement of expenses or to provide indemnification. The court shall order the Corporation to make an advancement of expenses or to provide indemnification, as the case may be, if it determines that the indemnitee is entitled under these Bylaws to such an advancement of expenses or indemnification, and in such event shall order the Corporation to pay the indemnitee's reasonable expenses (including attorneys' fees) to obtain the order. Neither the failure of the Corporation (including its Board of Directors, committee, special legal counsel or its stockholders) to have made a determination, as provided in Subsection (c) of Section 8.1 of these Bylaws, prior to the commencement of such action permitted by this Section, that the indemnitee is entitled to receive an advancement of expenses or indemnification, nor the determination by the Corporation (including its Board of Directors, committee, special legal counsel or its stockholders) that the indemnitee is not entitled to an advancement of expenses or indemnification, shall create a presumption to that effect or otherwise itself be a defense to that indemnitee's application for an advancement of expenses or indemnification.

Section 8.3 Non-Exclusivity of Rights. The rights to indemnification and to the advancement of expenses conferred in these Bylaws shall not be exclusive of any other right which any person may have or hereafter acquire under any statute (including the Corporation's enabling legislation), or any agreement, vote of stockholders or disinterested directors or otherwise.

Section 8.4 <u>Insurance.</u> The Corporation may purchase and maintain insurance, at its expense, on behalf of itself and also on behalf of any individual who is or was a director, officer, employee or agent of the Corporation or who, while a director, officer, employee or agent of the Corporation, is or was serving at the request of the Corporation as a director, officer, manager, partner, trustee, employee, or agent of another corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other entity against any expense, liability or loss, asserted against or incurred or suffered by him or her in that capacity or arising from his or her status as a director, officer, manager, employee or agent, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under this Article 8.

Section 8.5 <u>Limitation of Liability.</u> No monetary damages or monetary liability of any kind may be assessed against an officer or director in any proceeding brought by or in the right of the Corporation or brought by or on behalf of the stockholders of the Corporation; provided, however, that this elimination of liability shall not be applicable if the officer or director engaged in willful misconduct, a transaction from which the director or officer derived an improper personal benefit, or a knowing violation of the criminal law or of any federal or state securities law, including, without limitation, any claim of unlawful insider trading or the manipulation of the market for any security.

ARTICLE 9 - SEAL

Section 9.1 Corporate Seal. The Board of Directors may adopt a suitable seal, containing the name of the Corporation, which seal shall be in the charge of the Corporate Secretary.

ARTICLE 10 - FISCAL YEAR

Section 10.1 Fiscal Year. The fiscal year of the Corporation shall be the calendar year.

ARTICLE 11 - MISCELLANEOUS

Section 11.1 <u>Time Periods.</u> In applying any provision of these Bylaws which requires that an act be done or not be done a specified number of days prior to an event or that an act be done during a period of a specified number of days prior to an event, calendar days shall be used in making such computations, the day of the doing of the act shall be excluded and the day of the event shall be included.

Section 11.2 <u>Severability.</u> If any provision or provisions of these Bylaws shall be held invalid or unenforceable for any reason whatsoever, the validity and enforceability of the remaining provisions shall not in any way be affected or impaired thereby and, to the fullest extent possible, the remaining provisions shall be construed

so as to effectuate the intent manifested by the invalidated or unenforceable provision(s).

Section 11.3 <u>Corporate Governance Practices and Procedures and Governing Law.</u>

- (a) The corporate governance practices and procedures of the Corporation shall comply with the Corporation's enabling legislation and other Federal law, rules, and regulations, and shall be consistent with the safe and sound operation of the Corporation. To the extent not inconsistent with the foregoing, the Corporation shall follow the corporate governance practices and procedures of the law of the Commonwealth of Virginia, including without limitation the Virginia Stock Corporation Act as the same may be amended from time to time. Subject to all of the foregoing, these Bylaws and any rights and obligations created by these Bylaws shall be construed in accordance with, and governed by, the laws of the United States, using the law of the Commonwealth of Virginia as the federal rule of decision in all instances.
- (b) Section 1.1, Section 2.1, Section 2.2, Section 2.3, Section 2.4, Section 2.5, Section 2.6, Section 2.7, Section 3.2, Section 3.7, Section 4.1, Section 4.2, Section 4.5, Section 4.17, Article 8, Section 11.3, Section 11.4, Section 11.6, Article 12 of these Bylaws, and any new bylaw which may be adopted from time to time and designated as a "Level 1 Provision" in accordance with Article 12 of these Bylaws shall collectively be referred to herein as "Level 1 Provisions." Level 1 Provisions shall be deemed to constitute provisions of the Corporation's "articles of incorporation" for all purposes of the Virginia Stock Corporation Act. Any bylaw that is not a Level 1 Provision and any new bylaw that may be adopted from time to time and is not designated as a "Level 1 Provision" by the Board of Directors shall collectively be referred to herein as "Level 2 Provisions."
- **Section 11.4** Certificates of Designation. The provisions of these Bylaws shall supplement the terms of the Common Stock Certificate of Designation and any certificate of designation approved by the Board of Directors with respect to any class of the Corporation's preferred stock. In the event of any conflict between the terms of any such certificate of designation and these Bylaws, the terms of the certificate of designation shall govern.
- **Section 11.5** <u>Statutory References.</u> Each reference in these Bylaws to a particular statute or regulation, or a provision thereof, is a reference to such provision as amended or re-enacted or as modified by other statutory provisions from time to time and includes subsequent legislation and regulations made under the relevant statute.
- **Section 11.6** Control Share Acquisitions. Article 14.1 of the Virginia Stock Corporation Act shall not apply to acquisitions of shares of the Corporation. This section has been included in these Bylaws for the avoidance of doubt as to whether the election of Virginia law in Section 11.3 and other sections of these Bylaws for purposes of the

Corporation's corporate governance practices and procedures was intended to include Article 14.1 of the Virginia Stock Corporation Act.

ARTICLE 12 - AMENDMENTS

Section 12.1 General. Subject to the provisions of this Article 12, Level 1 Provisions or Level 2 Provisions may be amended, adopted, rescinded or repealed by the Board of Directors or the stockholders at any meeting, provided that in the case of such an action by the stockholders, notice of the proposed change must be given in the notice of the meeting.

Section 12.2 Amendment by the Board of Directors. Any new bylaw adopted by the Board of Directors and any Level 1 Provision, may be amended only by the Board of Directors pursuant to Section 12.1 of these Bylaws. Upon adopting or amending such bylaw, as the case may be, the Board of Directors shall designate such bylaw as a "Level 1 Provision" for all purposes under these Bylaws. If the Board of Directors does not designate a bylaw as a Level 1 Provision or if the Board of Directors is otherwise silent on the designation, the bylaw shall be deemed to be a Level 2 Provision.

Section 12.3 Amendment by the Stockholders. Notwithstanding any other provisions of these Bylaws, or any provisions of law, which might otherwise permit a lesser vote or no vote, but in addition to any affirmative vote of the stockholders required by law, the Common Stock Certificate of Designation, or these Bylaws, the affirmative vote of the holders of at least a majority of all shares of voting stock then outstanding and entitled to vote shall be required to amend, adopt, rescind or repeal any Level 2 Provision of these Bylaws (the "Proposed Level 2 Provision"); provided, however, that, as determined by the Board of Directors, (i) the Proposed Level 2 Provision shall comply with the Corporation's enabling legislation and other Federal law, rules, regulations, regulatory guidance and other issuances, (ii) the Proposed Level 2 Provision shall be consistent with the safe and sound operation of the Corporation, (iii) the subject matter of the Proposed Level 2 Provision does not or would not involve the subject matter of any Level 1 Provision, the Common Stock Certificate of Designation or any other certificate of designation of the Corporation, and (iv) the Proposed Level 2 Provision does not or would not be inconsistent with any Level 1 Provision, the Common Stock Certificate of Designation or any other certificate of designation of the Corporation. The stockholders may not amend, adopt, rescind or repeal any Level 1 Provision unless such action is explicitly authorized and referred to the stockholders by the Board of Directors (for the avoidance of doubt, this Section 12.3 in no way obligates the Board of Directors to seek stockholder approval for any action pursuant to Section 12.2 of these Bylaws) in which case such amendment, adoption, rescission, or repeal shall be by the affirmative vote of the holders of at least a majority of all shares of the voting stock then outstanding and entitled to vote.

EXHIBIT C

Exhibit 99.2

FEDERAL HOUSING FINANCE AGENCY STATEMENT





Contact: Corinne Russell (202) 414-6921 Stefanie Mullin (202) 414-6376

For Immediate Release September 7, 2008

STATEMENT OF FHFA DIRECTOR JAMES B. LOCKHART

Good Morning

Fannie Mae and Freddie Mac share the critical mission of providing stability and liquidity to the housing market. Between them, the Enterprises have \$5.4 trillion of guaranteed mortgage-backed securities (MBS) and debt outstanding, which is equal to the publicly held debt of the United States. Their market share of all new mortgages reached over 80 percent earlier this year, but it is now falling. During the turmoil last year, they played a very important role in providing liquidity to the conforming mortgage market. That has required a very careful and delicate balance of mission and safety and soundness. A key component of this balance has been their ability to raise and maintain capital. Given recent market conditions, the

balance has been lost. Unfortunately, as house prices, earnings and capital have continued to deteriorate, their ability to fulfill their mission has deteriorated. In particular, the capacity of their capital to absorb further losses while supporting new business activity is in doubt.

Today's action addresses safety and soundness concerns. FHFA's rating system is called GSE Enterprise Risk or G-Seer. It stands for Governance, Solvency, Earnings and Enterprise Risk which includes credit, market and operational risk. There are pervasive weaknesses across the board, which have been getting worse in this market.

Over the last three years OFHEO, and now FHFA, have worked hard to encourage the Enterprises to rectify their accounting, systems, controls and risk management issues. They have made good progress in many areas, but market conditions have overwhelmed that progress.

The result has been that they have been unable to provide needed stability to the market. They also find themselves unable to meet their affordable housing mission. Rather than letting these conditions fester and worsen and put our markets in jeopardy, FHFA, after painstaking review, has decided to take action now.

Key events over the past six months have demonstrated the increasing challenge faced by the companies in striving to balance mission and safety and soundness, and the ultimate disruption of that balance that led to today's announcements. In the first few months of this year, the secondary market showed significant deterioration, with buyers demanding much higher prices for mortgage backed securities.

In February, in recognition of the remediation progress in financial reporting, we removed the portfolio caps on each company, but they did not have the capital to use that flexibility.

In March, we announced with the Enterprises an initiative to increase mortgage market liquidity and market confidence. We reduced the OFHEO-directed capital requirements in return for their commitments to raise significant capital and to maintain overall capital levels well in excess of requirements.

In April, we released our Annual Report to Congress, identifying each company as a significant supervisory concern and noting, in particular, the deteriorating mortgage credit environment and the risks it posed to the companies.

3

In May OFHEO lifted its 2006 Consent Order with Fannie Mae after the company completed the terms of that order. Subsequently, Fannie Mae successfully raised \$7.4 billion of new capital, but Freddie Mac never completed the capital raise promised in March.

Since then credit conditions in the mortgage market continued to deteriorate, with home prices continuing to decline and mortgage delinquency rates reaching alarming levels. FHFA intensified its reviews of each company's capital planning and capital position, their earnings forecasts and the effect of falling house prices and increasing delinquencies on the credit quality of their mortgage book.

In getting to today, the supervision team has spent countless hours reviewing with each company various forecasts, stress tests, and projections, and has evaluated the performance of their internal models in these analyses. We have had many meetings with each company's management teams, and have had frank exchanges regarding loss projections, asset valuations, and capital adequacy. More recently, we have gone the extra step of inviting the Federal Reserve and the OCC to have some of their senior mortgage credit experts join our team in these assessments.

The conclusions we reach today, while our own, have had the added benefit of their insight and perspective.

After this exhaustive review, I have determined that the companies cannot continue to operate safely and soundly and fulfill their critical public mission, without significant action to address our concerns, which are:

- the safety and soundness issues I mentioned, including current capitalization;
- current market conditions;
- the financial performance and condition of each company;
- the inability of the companies to fund themselves according to normal practices and prices; and
- · the critical importance each company has in supporting the residential mortgage market in this country,

Therefore, in order to restore the balance between safety and soundness and mission, FHFA has placed Fannie Mae and Freddie Mac into conservatorship. That is a statutory process designed to stabilize a troubled institution with the

objective of returning the entities to normal business operations. FHFA will act as the conservator to operate the Enterprises until they are stabilized.

The Boards of both companies consented yesterday to the conservatorship. I appreciate the cooperation we have received from the boards and the management of both Enterprises. These individuals did not create the inherent conflict and flawed business model embedded in the Enterprises' structure.

The goal of these actions is to help restore confidence in Fannie Mae and Freddie Mac, enhance their capacity to fulfill their mission, and mitigate the systemic risk that has contributed directly to the instability in the current market. The lack of confidence has resulted in continuing spread widening of their MBS, which means that virtually none of the large drop in interest rates over the past year has been passed on to the mortgage markets. On top of that, Freddie Mac and Fannie Mae, in order to try to build capital, have continued to raise prices and tighten credit standards.

FHFA has not undertaken this action lightly. We have consulted with the Chairman of the Board of Governors of the Federal Reserve System, Ben Bernanke, who was appointed a consultant to FHFA under the new legislation. We

6

have also consulted with the Secretary of the Treasury, not only as an FHFA Oversight Board member, but also in his duties under the law to provide financing to the GSEs. They both concurred with me that conservatorship needed to be undertaken now.

There are several key components of this conservatorship:

First, Monday morning the businesses will open as normal, only with stronger backing for the holders of MBS, senior debt and subordinated debt.

Second, the Enterprises will be allowed to grow their guarantee MBS books without limits and continue to purchase replacement securities for their portfolios, about \$20 billion per month without capital constraints.

Third, as the conservator, FHFA will assume the power of the Board and management.

Fourth, the present CEOs will be leaving, but we have asked them to stay on to help with the transition.

Fifth, I am announcing today I have selected Herb Allison to be the new CEO of Fannie Mae and David Moffett the CEO of Freddie Mac. Herb has been the Vice Chairman of Merrill Lynch and for the last eight years chairman of TIAA-CREF. David was the Vice Chairman and CFO of US Bancorp. I appreciate the willingness of these two men to take on these tough jobs during these challenging times. Their compensation will be significantly lower than the outgoing CEOs. They will be joined by equally strong non-executive chairmen.

Sixth, at this time any other management action will be very limited. In fact, the new CEOs have agreed with me that it is very important to work with the current management teams and employees to encourage them to stay and to continue to make important improvements to the Enterprises.

Seventh, in order to conserve over \$2 billion in capital every year, the common stock and preferred stock dividends will be eliminated, but the common and all preferred stocks will continue to remain outstanding. Subordinated debt interest and principal payments will continue to be made.

Eighth, all political activities — including all lobbying — will be halted immediately. We will review the charitable activities.

Lastly and very importantly, there will be the financing and investing relationship with the U.S. Treasury, which Secretary Paulson will be discussing. We believe that these facilities will provide the critically needed support to Freddie Mac and Fannie Mae and importantly the liquidity of the mortgage market.

One of the three facilities he will be mentioning is a secured liquidity facility which will be not only for Fannie Mae and Freddie Mac, but also for the 12 Federal Home Loan Banks that FHFA also regulates. The Federal Home Loan Banks have performed remarkably well over the last year as they have a different business model than Fannie Mae and Freddie Mac and a different capital structure that grows as their lending activity grows. They are joint and severally liable for the Bank System's debt obligations and all but one of the 12 are profitable. Therefore, it is very unlikely that they will use the facility.

During the conservatorship period, FHFA will continue to work expeditiously on the many regulations needed to implement the new law. Some of the key regulations will be minimum capital standards, prudential safety and soundness standards and portfolio limits. It is critical to complete these regulations so that any new investor will understand the investment proposition.

This decision was a tough one for the FHFA team as they have worked so hard to help the Enterprises remain strong suppliers of support to the secondary mortgage markets. Unfortunately, the antiquated capital requirements and the turmoil in housing markets over-whelmed all the good and hard work put in by the FHFA teams and the Enterprises' managers and employees. Conservatorship will give the Enterprises the time to restore the balances between safety and soundness and provide affordable housing and stability and liquidity to the mortgage markets. I want to thank the FHFA employees for their work during this intense regulatory process. They represent the best in public service. I would also like to thank the employees of Fannie Mae and Freddie Mac for all their hard work. Working together we can finish the job of restoring confidence in the Enterprises and with the new legislation build a stronger and safer future for the mortgage markets, homeowners and renters in America.

Thank you and I will now turn it back to Secretary Paulson.

(link to) QUESTIONS AND ANSWERS ON CONSERVATORSHIP

EXHIBIT D

FEDERAL HOUSING FINANCE AGENCY





FACT SHEET

Contact: Corinne Russell (202) 414-6921

Stefanie Mullin (202) 414-6376

****EMBARGOED UNTIL 11AM****

QUESTIONS AND ANSWERS ON CONSERVATORSHIP

- Q: What is a conservatorship?
- A: A conservatorship is the legal process in which a person or entity is appointed to establish control and oversight of a Company to put it in a sound and solvent condition. In a conservatorship, the powers of the Company's directors, officers, and shareholders are transferred to the designated Conservator.
- Q: What is a Conservator?
- A: A Conservator is the person or entity appointed to oversee the affairs of a Company for the purpose of bringing the Company back to financial health.
 - In this instance, the Federal Housing Finance Agency ("FHFA") has been appointed by its Director to be the Conservator of the Company in accordance with the Federal Housing Finance Regulatory Reform Act of 2008 (Public Law 110-289) and the Federal Housing Enterprises Financial Safety and Soundness Act of 1992 (12 U.S.C. 4501, et seq., as amended) to keep the Company in a safe and solvent financial condition.
- Q: How is a Conservator appointed?
- A: By statute, the FHFA is appointed Conservator by its Director after the Director determines, in his discretion, that the Company is in need of reorganization or rehabilitation of its affairs.

- Q: What are the goals of this conservatorship?
- A: The purpose of appointing the Conservator is to preserve and conserve the Company's assets and property and to put the Company in a sound and solvent condition. The goals of the conservatorship are to help restore confidence in the Company, enhance its capacity to fulfill its mission, and mitigate the systemic risk that has contributed directly to the instability in the current market.

There is no reason for concern regarding the ongoing operations of the Company. The Company's operation will not be impaired and business will continue without interruption.

- Q: When will the conservatorship period end?
- A: Upon the Director's determination that the Conservator's plan to restore the Company to a safe and solvent condition has been completed successfully, the Director will issue an order terminating the conservatorship. At present, there is no exact time frame that can be given as to when this conservatorship may end.
- Q: What are the powers of the Conservator?
- A: The FHFA, as Conservator, may take all actions necessary and appropriate to (1) put the Company in a sound and solvent condition and (2) carry on the Company's business and preserve and conserve the assets and property of the Company.
- Q: What happens upon appointment of a Conservator?
- A: Once an "Order Appointing a Conservator" is signed by the Director of FHFA, the Conservator immediately succeeds to the (1) rights, titles, powers, and privileges of the Company, and any stockholder, officer, or director of such the Company with respect to the Company and its assets, and (2) title to all books, records and assets of the Company held by any other custodian or third-party. The Conservator is then charged with the duty to operate the Company.
- Q: What does the Conservator do during a conservatorship?
- A: The Conservator controls and directs the operations of the Company. The Conservator may (1) take over the assets of and operate the Company with all the powers of the shareholders, the directors, and the officers of the Company and conduct all business of the Company; (2) collect all obligations and money due to the Company; (3) perform all functions of the Company which are consistent with the Conservator's appointment; (4) preserve and conserve the assets and property of the Company; and (5) contract for assistance in fulfilling any function, activity, action or duty of the Conservator.
- Q: How will the Company run during the conservatorship?

- A: The Company will continue to run as usual during the conservatorship. The Conservator will delegate authorities to the Company's management to move forward with the business operations. The Conservator encourages all Company employees to continue to perform their job functions without interruption.
- Q: Will the Company continue to pays its obligations during the conservatorship?
- A: Yes, the Company's obligations will be paid in the normal course of business during the Conservatorship. The Treasury Department, through a secured lending credit facility and a Senior Preferred Stock Purchase Agreement, has significantly enhanced the ability of the Company to meet its obligations. The Conservator does not anticipate that there will be any disruption in the Company's pattern of payments or ongoing business operations.
- Q: What happens to the Company's stock during the conservatorship?
- A: During the conservatorship, the Company's stock will continue to trade. However, by statute, the powers of the stockholders are suspended until the conservatorship is terminated. Stockholders will continue to retain all rights in the stock's financial worth; as such worth is determined by the market.
- Q: Is the Company able to buy and sell investments and complete financial transactions during the conservatorship?
- A: Yes, the Company's operations continue subject to the oversight of the Conservator.
- Q: What happens if the Company is liquidated?
- A: Under a conservatorship, the Company is not liquidated.
- Q: Can the Conservator determine to liquidate the Company?
- A: The Conservator cannot make a determination to liquidate the Company, although, short of that, the Conservator has the authority to run the company in whatever way will best achieve the Conservator's goals (discussed above). However, assuming a statutory ground exists and the Director of FHFA determines that the financial condition of the company requires it, the Director does have the discretion to place any regulated entity, including the Company, into receivership. Receivership is a statutory process for the liquidation of a regulated entity. There are no plans to liquidate the Company.
- Q: Can the Company be dissolved?
- A: Although the company can be liquidated as explained above, by statute the charter of the Company must be transferred to a new entity and can only be dissolved by an Act of Congress.

EXHIBIT E

AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT

AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT (this "Agreement") dated as of September 26, 2008, between the UNITED STATES DEPARTMENT OF THE TREASURY ("Purchaser") and FEDERAL NATIONAL MORTGAGE ASSOCIATION ("Seller"), acting through the Federal Housing Finance Agency (the "Agency") as its duly appointed conservator (the Agency in such capacity, "Conservator"). Reference is made to Article 1 below for the meaning of capitalized terms used herein without definition.

Background

- A. The Agency has been duly appointed as Conservator for Seller pursuant to Section 1367(a) of the Federal Housing Enterprises Financial Safety and Soundness Act of 1992 (as amended, the "FHE Act"). Conservator has determined that entry into this Agreement is (i) necessary to put Seller in a sound and solvent condition; (ii) appropriate to carry on the business of Seller and preserve and conserve the assets and property of Seller; and (iii) otherwise consistent with its powers, authorities and responsibilities.
- B. Purchaser is authorized to purchase obligations and other securities issued by Seller pursuant to Section 304(g) of the Federal National Mortgage Association Charter Act, as amended (the "Charter Act"). The Secretary of the Treasury has determined, after taking into consideration the matters set forth in Section 304(g)(1)(C) of the Charter Act, that the purchases contemplated herein are necessary to (i) provide stability to the financial markets; (ii) prevent disruptions in the availability of mortgage finance; and (iii) protect the taxpayer.
- C. Purchaser and Seller executed and delivered the Senior Preferred Stock Purchase Agreement dated as of September 7, 2008 (the "Original Agreement"), and the parties thereto desire to amend and restate the Original Agreement in its entirety as set forth herein.

THEREFORE, the parties hereto agree as follows:

Terms and Conditions

1. **DEFINITIONS**

As used in this Agreement, the following terms shall have the meanings set forth below:

"Affiliate" means, when used with respect to a specified Person (i) any direct or indirect holder or group (as defined in Sections 13(d) and 14(d) of the Exchange Act) of holders of 10.0% or more of any class of capital stock of such Person and (ii) any current or former director or officer of such Person, or any other current or former employee of such Person that currently exercises or formerly exercised a material degree of Control over such Person, including without limitation each current or former Named Executive Officer of such Person.

- "Available Amount" means, as of any date of determination, the lesser of (a) the Deficiency Amount as of such date and (b) the Maximum Amount as of such date.
- "Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under United States federal law and the law of the State of New York.
- "Capital Lease Obligations" of any Person shall mean the obligations of such Person to pay rent or other amounts under any lease of (or other similar arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP and, for purposes hereof, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.
- "Control" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities, by contract or otherwise.
- "Deficiency Amount" means, as of any date of determination, the amount, if any, by which (a) the total liabilities of Seller exceed (b) the total assets of Seller (such assets excluding the Commitment and any unfunded amounts thereof), in each case as reflected on the balance sheet of Seller as of the applicable date set forth in this Agreement, prepared in accordance with GAAP; provided, however, that:
 - (i) for the avoidance of doubt, in measuring the Deficiency Amount liabilities shall exclude any obligation in respect of any capital stock of Seller, including the Senior Preferred Stock contemplated herein;
 - (ii) in the event that Seller becomes subject to receivership or other liquidation process or proceeding, "Deficiency Amount" shall mean, as of any date of determination, the amount, if any, by which (a) the total allowed claims against the receivership or other applicable estate (excluding any liabilities of or transferred to any LLRE (as defined in Section 5.4(a)) created by a receiver) exceed (b) the total assets of such receivership or other estate (excluding the Commitment, any unfunded amounts thereof and any assets of or transferred to any LLRE, but including the value of the receiver's interest in any LLRE);
 - (iii) to the extent Conservator or a receiver of Seller, or any statute, rule, regulation or court of competent jurisdiction, specifies or determines that a liability of Seller (including without limitation a claim against Seller arising from rescission of a purchase or sale of a security issued by Seller (or guaranteed by Seller or with respect to which Seller is otherwise liable) or for damages arising from the purchase, sale or retention of such a security) shall be subordinated (other than pursuant to a contract providing for such subordination) to all other liabilities of Seller or shall be treated on par with any class of equity of Seller, then such liability shall be excluded in the calculation of Deficiency Amount; and

- (iv) the Deficiency Amount may be increased above the otherwise applicable amount by the mutual written agreement of Purchaser and Seller, each acting in its sole discretion.
- "Designated Representative" means Conservator or (a) if Conservator has been superseded by a receiver pursuant to Section 1367(a) of the FHE Act, such receiver, or (b) if Seller is not in conservatorship or receivership pursuant to Section 1367(a) of the FHE Act, Seller's chief financial officer.
- "Director" shall mean the Director of the Agency.
- "Effective Date" means the date on which this Agreement shall have been executed and delivered by both of the parties hereto.
- "Equity Interests" of any Person shall mean any and all shares, interests, rights to purchase or otherwise acquire, warrants, options, participations or other equivalents of or interests in (however designated) equity, ownership or profits of such Person, including any preferred stock, any limited or general partnership interest and any limited liability company membership interest, and any securities or other rights or interests convertible into or exchangeable for any of the foregoing.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the SEC promulgated thereunder.
- "GAAP" means generally accepted accounting principles in effect in the United States as set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board from time to time.
- "Indebtedness" of any Person means, for purposes of Section 5.5 only, without duplication, (a) all obligations of such Person for money borrowed by such Person, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (d) all obligations of such Person issued or assumed as the deferred purchase price of property or services, other than trade accounts payable, (e) all Capital Lease Obligations of such Person, (f) obligations, whether contingent or liquidated, in respect of letters of credit (including standby and commercial), bankers' acceptances and similar instruments and (g) any obligation of such Person, contingent or otherwise, guaranteeing or having the economic effect of guaranteeing any Indebtedness of the types set forth in clauses (a) through (f) payable by another Person other than Mortgage Guarantee Obligations.
- "Liquidation End Date" means the date of completion of the liquidation of Seller's assets.
- "Maximum Amount" means, as of any date of determination, \$100,000,000,000 (one hundred billion dollars), less the aggregate amount of funding under the Commitment prior to such date.

- "Mortgage Assets" of any Person means assets of such Person consisting of mortgages, mortgage loans, mortgage-related securities, participation certificates, mortgage-backed commercial paper, obligations of real estate mortgage investment conduits and similar assets, in each case to the extent such assets would appear on the balance sheet of such Person in accordance with GAAP as in effect as of the date hereof (and, for the avoidance of doubt, without giving effect to any change that may be made hereafter in respect of Statement of Financial Accounting Standards No. 140 or any similar accounting standard).
- "Mortgage Guarantee Obligations" means guarantees, standby commitments, credit enhancements and other similar obligations of Seller, in each case in respect of Mortgage Assets.
- "Named Executive Officer" has the meaning given to such term in Item 402(a)(3) of Regulation S-K under the Exchange Act, as in effect on the date hereof.
- "Person" shall mean any individual, corporation, limited liability company, partnership, joint venture, association, joint-stock company, trust, estate, unincorporated organization or government or any agency or political subdivision thereof, or any other entity whatsoever.
- "SEC" means the Securities and Exchange Commission.
- "Senior Preferred Stock" means the Variable Liquidation Preference Senior Preferred Stock of Seller, substantially in the form of Exhibit A hereto.
- "Warrant" means a warrant for the purchase of common stock of Seller representing 79.9% of the common stock of Seller on a fully-diluted basis, substantially in the form of Exhibit B hereto.

2. COMMITMENT

- 2.1. Commitment. Purchaser hereby commits to provide to Seller, on the terms and conditions set forth herein, immediately available funds in an amount up to but not in excess of the Available Amount, as determined from time to time (the "Commitment"); provided, that in no event shall the aggregate amount funded under the Commitment exceed \$100,000,000,000 (one hundred billion dollars). The liquidation preference of the Senior Preferred Stock shall increase in connection with draws on the Commitment, as set forth in Section 3.3 below.
- 2.2. Quarterly Draws on Commitment. Within fifteen (15) Business Days following the determination of the Deficiency Amount, if any, as of the end of each fiscal quarter of Seller which ends on or before the Liquidation End Date, the Designated Representative may, on behalf of Seller, request that Purchaser provide immediately available funds to Seller in an amount up to but not in excess of the Available Amount as of the end of such quarter. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains a certification of the Designated Representative that the requested amount does not exceed the Available Amount as of the end of the applicable quarter. Purchaser shall provide such funds within sixty (60) days of its receipt of such request or, following any determination by the Director that the Director will be mandated by law to appoint a receiver for Seller if such funds are not received sooner, such shorter period as may be necessary

to avoid such mandatory appointment of a receiver if reasonably practicable taking into consideration Purchaser's access to funds.

- 2.3. Accelerated Draws on Commitment. Immediately following any determination by the Director that the Director will be mandated by law to appoint a receiver for Seller prior to the Liquidation End Date unless Seller's capital is increased by an amount (the "Special Amount") up to but not in excess of the then current Available Amount (computed based on a balance sheet of Seller prepared in accordance with GAAP that differs from the most recent balance sheet of Seller delivered in accordance with Section 5.9(a) or (b)) on a date that is prior to the date that funds will be available to Seller pursuant to Section 2.2, Conservator may, on behalf of Seller, request that Purchaser provide to Seller the Special Amount in immediately available funds. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains certifications of Conservator that (i) the requested amount does not exceed the Available Amount (including computations in reasonable detail and satisfactory to Purchaser of the then existing Deficiency Amount) and (ii) the requested amount is required to avoid the imminent mandatory appointment of a receiver for Seller. Purchaser shall provide such funds within thirty (30) days of its receipt of such request or, if reasonably practicable taking into consideration Purchaser's access to funds, any shorter period as may be necessary to avoid mandatory appointment of a receiver.
- 2.4. Final Draw on Commitment. Within fifteen (15) Business Days following the determination of the Deficiency Amount, if any, as of the Liquidation End Date (computed based on a balance sheet of Seller as of the Liquidation End Date prepared in accordance with GAAP), the Designated Representative may, on behalf of Seller, request that Purchaser provide immediately available funds to Seller in an amount up to but not in excess of the Available Amount as of the Liquidation End Date. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains a certification of the Designated Representative that the requested amount does not exceed the Available Amount (including computations in reasonable detail and satisfactory to Purchaser of the Deficiency Amount as of the Liquidation End Date). Purchaser shall provide such funds within sixty (60) days of its receipt of such request.
- 2.5. Termination of Purchaser's Obligations. Subject to earlier termination pursuant to Section 6.7, all of Purchaser's obligations under and in respect of the Commitment shall terminate upon the earliest of: (a) if the Liquidation End Date shall have occurred, (i) the payment in full of Purchaser's obligations with respect to any valid request for funds pursuant to Section 2.4 or (ii) if there is no Deficiency Amount on the Liquidation End Date or if no such request pursuant to Section 2.4 has been made, the close of business on the 15th Business Day following the determination of the Deficiency Amount, if any, as of the Liquidation End Date; (b) the payment in full of, defeasance of or other reasonable provision for all liabilities of Seller, whether or not contingent, including payment of any amounts that may become payable on, or expiry of or other provision for, all Mortgage Guarantee Obligations and provision for unmatured debts; and (c) the funding by Purchaser under the Commitment of an aggregate of \$100,000,000,000 (one hundred billion dollars). For the avoidance of doubt, the Commitment shall *not* be terminable by Purchaser solely by reason of (i) the conservatorship, receivership or other insolvency proceeding of Seller or (ii) the Seller's financial condition or any adverse change in Seller's financial condition.

3. PURCHASE OF SENIOR PREFERRED STOCK AND WARRANT; FEES

- 3.1. *Initial Commitment Fee.* In consideration of the Commitment, and for no additional consideration, on the Effective Date (or as soon thereafter as is practicable) Seller shall sell and issue to Purchaser, and Purchaser shall purchase from Seller, (a) one million (1,000,000) shares of Senior Preferred Stock, with an initial liquidation preference equal to \$1,000 per share (\$1,000,000,000 (one billion dollars) liquidation preference in the aggregate), and (b) the Warrant.
- 3.2. *Periodic Commitment Fee*. (a) Commencing March 31, 2010, Seller shall pay to Purchaser quarterly, on the last day of March, June, September and December of each calendar year (each a "<u>Periodic Fee Date</u>"), a periodic commitment fee (the "<u>Periodic Commitment Fee</u>"). The Periodic Commitment Fee shall accrue from January 1, 2010.
- (b) The Periodic Commitment Fee is intended to fully compensate Purchaser for the support provided by the ongoing Commitment following December 31, 2009. The amount of the Periodic Commitment Fee shall be set not later than December 31, 2009 with respect to the ensuing five-year period, shall be reset every five years thereafter and shall be determined with reference to the market value of the Commitment as then in effect. The amount of the Periodic Commitment Fee shall be mutually agreed by Purchaser and Seller, subject to their reasonable discretion and in consultation with the Chairman of the Federal Reserve; provided, that Purchaser may waive the Periodic Commitment Fee for up to one year at a time, in its sole discretion, based on adverse conditions in the United States mortgage market.
- (c) At the election of Seller, the Periodic Commitment Fee may be paid in cash or by adding the amount thereof ratably to the liquidation preference of each outstanding share of Senior Preferred Stock so that the aggregate liquidation preference of all such outstanding shares of Senior Preferred Stock is increased by an amount equal to the Periodic Commitment Fee. Seller shall deliver notice of such election not later than three (3) Business Days prior to each Periodic Fee Date. If the Periodic Commitment Fee is not paid in cash by 12:00 pm (New York time) on the applicable Periodic Fee Date (irrespective of Seller's election pursuant to this subsection), Seller shall be deemed to have elected to pay the Periodic Commitment Fee by adding the amount thereof to the liquidation preference of the Senior Preferred Stock, and the aggregate liquidation preference of the outstanding shares of Senior Preferred Stock shall thereupon be automatically increased, in the manner contemplated by the first sentence of this section, by an aggregate amount equal to the Periodic Commitment Fee then due.
- 3.3. Increases of Senior Preferred Stock Liquidation Preference as a Result of Funding under the Commitment. The aggregate liquidation preference of the outstanding shares of Senior Preferred Stock shall be automatically increased by an amount equal to the amount of each draw on the Commitment pursuant to Article 2 that is funded by Purchaser to Seller, such increase to occur simultaneously with such funding and ratably with respect to each share of Senior Preferred Stock.
- 3.4. *Notation of Increase in Liquidation Preference*. Seller shall duly mark its records to reflect each increase in the liquidation preference of the Senior Preferred Stock contemplated

herein (but, for the avoidance of doubt, such increase shall be effective regardless of whether Seller has properly marked its records).

4. REPRESENTATIONS

Seller represents and warrants as of the Effective Date, and shall be deemed to have represented and warranted as of the date of each request for and funding of an advance under the Commitment pursuant to Article 2, as follows:

- 4.1. *Organization and Good Standing*. Seller is a corporation, chartered by the Congress of the United States, duly organized, validly existing and in good standing under the laws of the United States and has all corporate power and authority to carry on its business as now conducted and as proposed to be conducted.
- 4.2. *Organizational Documents*. Seller has made available to Purchaser a complete and correct copy of its charter and bylaws, each as amended to date (the "<u>Organizational Documents</u>"). The Organizational Documents are in full force and effect. Seller is not in violation of any provision of its Organizational Documents.
- 4.3. Authorization and Enforceability. All corporate or other action on the part of Seller or Conservator necessary for the authorization, execution, delivery and performance of this Agreement by Seller and for the authorization, issuance and delivery of the Senior Preferred Stock and the Warrant being purchased under this Agreement, has been taken. This Agreement has been duly and validly executed and delivered by Seller and (assuming due authorization, execution and delivery by the Purchaser) shall constitute the valid and legally binding obligation of Seller, enforceable against Seller in accordance with its terms, except to the extent the enforceability thereof may be limited by bankruptcy laws, insolvency laws, reorganization laws, moratorium laws or other laws of general applicability affecting creditors' rights generally or by general equitable principles (regardless of whether enforcement is sought in a proceeding in equity or at law). The Agency is acting as conservator for Seller under Section 1367 of the FHE Act. The Board of Directors of Seller, by valid action at a duly called meeting of the Board of Directors on September 6, 2008, consented to the appointment of the Agency as conservator for purposes of Section 1367(a)(3)(I) of the FHE Act, and the Director of the Agency has appointed the Agency as Conservator for Seller pursuant to Section 1367(a)(1) of the FHE Act, and each such action has not been rescinded, revoked or modified in any respect.
- 4.4. *Valid Issuance*. When issued in accordance with the terms of this Agreement, the Senior Preferred Stock and the Warrant will be duly authorized, validly issued, fully paid and non-assessable, free and clear of all liens and preemptive rights. The shares of common stock to which the holder of the Warrant is entitled have been duly and validly reserved for issuance. When issued and delivered in accordance with the terms of this Agreement and the Warrant, such shares will be duly authorized, validly issued, fully paid and nonassessable, free and clear of all liens and preemptive rights.

4.5. Non-Contravention.

- (a) The execution, delivery or performance by Seller of this Agreement and the consummation by Seller of the transactions contemplated hereby do not and will not (i) conflict with or violate any provision of the Organizational Documents of Seller; (ii) conflict with or violate any law, decree or regulation applicable to Seller or by which any property or asset of Seller is bound or affected, or (iii) result in any breach of, or constitute a default (with or without notice or lapse of time, or both) under, or give to others any right of termination, amendment, acceleration or cancellation of, or result in the creation of a lien upon any of the properties or assets of Seller, pursuant to any note, bond, mortgage, indenture or credit agreement, or any other contract, agreement, lease, license, permit, franchise or other instrument or obligation to which Seller is a party or by which Seller is bound or affected, other than, in the case of clause (iii), any such breach, default, termination, amendment, acceleration, cancellation or lien that would not have and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the business, property, operations or condition of the Seller, the authority of the Conservator or the validity or enforceability of this Agreement (a "Material Adverse Effect").
- (b) The execution and delivery of this Agreement by Seller does not, and the consummation by Seller of the transactions contemplated by this Agreement will not, require any consent, approval, authorization, waiver or permit of, or filing with or notification to, any governmental authority or any other person, except for such as have already been obtained.

5. COVENANTS

From the Effective Date until such time as the Senior Preferred Stock shall have been repaid or redeemed in full in accordance with its terms:

- 5.1. Restricted Payments. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, declare or pay any dividend (preferred or otherwise) or make any other distribution (by reduction of capital or otherwise), whether in cash, property, securities or a combination thereof, with respect to any of Seller's Equity Interests (other than with respect to the Senior Preferred Stock or the Warrant) or directly or indirectly redeem, purchase, retire or otherwise acquire for value any of Seller's Equity Interests (other than the Senior Preferred Stock or the Warrant), or set aside any amount for any such purpose.
- 5.2. Issuance of Capital Stock. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, sell or issue Equity Interests of Seller or any of its subsidiaries of any kind or nature, in any amount, other than the sale and issuance of the Senior Preferred Stock and Warrant on the Effective Date and the common stock subject to the Warrant upon exercise thereof, and other than as required by (and pursuant to) the terms of any binding agreement as in effect on the date hereof.
- 5.3. *Conservatorship*. Seller shall not (and Conservator, by its signature below, agrees that it shall not), without the prior written consent of Purchaser, terminate, seek termination of or permit to be terminated the conservatorship of Seller pursuant to Section 1367 of the FHE Act, other

than in connection with a receivership pursuant to Section 1367 of the FHE Act.

- 5.4. *Transfer of Assets*. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, sell, transfer, lease or otherwise dispose of (in one transaction or a series of related transactions) all or any portion of its assets (including Equity Interests in other persons, including subsidiaries), whether now owned or hereafter acquired (any such sale, transfer, lease or disposition, a "<u>Disposition</u>"), other than Dispositions for fair market value:
- (a) to a limited life regulated entity (" \underline{LLRE} ") pursuant to Section 1367(i) of the FHE Act;
- (b) of assets and properties in the ordinary course of business, consistent with past practice;
- (c) in connection with a liquidation of Seller by a receiver appointed pursuant to Section 1367(a) of the FHE Act;
 - (d) of cash or cash equivalents for cash or cash equivalents; or
 - (e) to the extent necessary to comply with the covenant set forth in Section 5.7 below.
- 5.5. Indebtedness. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, incur, assume or otherwise become liable for (a) any Indebtedness if, after giving effect to the incurrence thereof, the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis would exceed 110.0% of the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis as of June 30, 2008 or (b) any Indebtedness if such Indebtedness is subordinated by its terms to any other Indebtedness of Seller or the applicable subsidiary. For purposes of this covenant the acquisition of a subsidiary with Indebtedness will be deemed to be the incurrence of such Indebtedness at the time of such acquisition.
- 5.6. Fundamental Changes. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, (i) merge into or consolidate or amalgamate with any other Person, or permit any other Person to merge into or consolidate or amalgamate with it, (ii) effect a reorganization or recapitalization involving the common stock of Seller, a reclassification of the common stock of Seller or similar corporate transaction or event or (iii) purchase, lease or otherwise acquire (in one transaction or a series of transactions) all or substantially all of the assets of any other Person or any division, unit or business of any Person.
- 5.7. *Mortgage Assets*. Seller shall not own, as of any applicable date, Mortgage Assets in excess of (i) on December 31, 2009, \$850 billion, or (ii) on December 31 of each year thereafter, 90.0% of the aggregate amount of Mortgage Assets of Seller as of December 31 of the immediately preceding calendar year; <u>provided</u>, that in no event shall Seller be required under this Section 5.7 to own less than \$250 billion in Mortgage Assets.

5.8. Transactions with Affiliates. Seller shall not, and shall not permit any of its subsidiaries to, without the prior written consent of Purchaser, engage in any transaction of any kind or nature with an Affiliate of Seller unless such transaction is (i) pursuant to this Agreement, the Senior Preferred Stock or the Warrant, (ii) upon terms no less favorable to Seller than would be obtained in a comparable arm's-length transaction with a Person that is not an Affiliate of Seller or (iii) a transaction undertaken in the ordinary course or pursuant to a contractual obligation or customary employment arrangement in existence as of the date hereof.

5.9. Reporting. Seller shall provide to Purchaser:

- (a) not later than the time period specified in the SEC's rules and regulations with respect to issuers as to which Section 13 and 15(d) of the Exchange Act apply, annual reports on Form 10-K (or any successor or comparable form) containing the information required to be contained therein (or required in such successor or comparable form);
- (b) not later than the time period specified in the SEC's rules and regulations with respect to issuers as to which Section 13 and 15(d) of the Exchange Act apply, reports on Form 10-Q (or any successor or comparable form) containing the information required to be contained therein (or required in such successor or comparable form);
- (c) promptly from time to time after the occurrence of an event required to be therein reported (and in any event within the time period specified in the SEC's rules and regulations), such other reports on Form 8-K (or any successor or comparable form);
- (d) concurrently with any delivery of financial statements under paragraphs (a) or (b) above, a certificate of the Designated Representative, (i) certifying that Seller is (and since the last such certificate has at all times been) in compliance with each of the covenants contained herein and that no representation made by Seller herein or in any document delivered pursuant hereto or in connection herewith was false or misleading in any material respect when made, or, if the foregoing is not true, specifying the nature and extent of the breach of covenant and/or representation and any corrective action taken or proposed to be taken with respect thereto, and (ii) setting forth computations in reasonable detail and satisfactory to the Purchaser of the Deficiency Amount, if any;
- (e) promptly, from time to time, such other information regarding the operations, business affairs, plans, projections and financial condition of Seller, or compliance with the terms of this Agreement, as Purchaser may reasonably request; and
 - (f) as promptly as reasonably practicable, written notice of the following:
 - (i) the occurrence of the Liquidation End Date;
 - (ii) the filing or commencement of, or any written threat or notice of intention of any Person to file or commence, any action, suit or proceeding, whether at law or in equity or by or before any governmental authority or in arbitration, against Conservator, Seller or any other Person which, if adversely determined, would reasonably be expected to have a Material Adverse Effect;

- (iii) any other development that is not a matter of general public knowledge and that has had, or would reasonably be expected to have, a Material Adverse Effect.
- 5.10. Executive Compensation. Seller shall not, without the consent of the Director, in consultation with the Secretary of the Treasury, enter into any new compensation arrangements with, or increase amounts or benefits payable under existing compensation arrangements of, any Named Executive Officer of Seller.

6. MISCELLANEOUS

- 6.1. No Third-Party Beneficiaries. Until the termination of the Commitment, at any time during the existence and continuance of a payment default with respect to debt securities issued by Seller and/or a default by Seller with respect to any Mortgage Guarantee Obligations, any holder of such defaulted debt securities or beneficiary of such Mortgage Guarantee Obligations (collectively, the "Holders") may (a) deliver notice to the Seller and the Designated Representative requesting exercise of all rights available to them under this Agreement to draw on the Commitment up to the lesser of the amount necessary to cure the outstanding payment defaults and the Available Amount as of the last day of the immediately preceding fiscal quarter (the "Demand Amount"), (b) if Seller and the Designated Representative fail to act as requested within thirty (30) days of such notice, seek judicial relief for failure of the Seller to draw on the Commitment, and (c) if Purchaser shall fail to perform its obligations in respect of any draw on the Commitment, and Seller and/or the Designated Representative shall not be diligently pursuing remedies in respect of such failure, file a claim in the United States Court of Federal Claims for relief requiring Purchaser to pay Seller the Demand Amount in the form of liquidated damages. Any payment of liquidated damages to Seller under the previous sentence shall be treated for all purposes, including the provisions of the Senior Preferred Stock and Section 3.3 of this Agreement, as a draw and funding of the Commitment pursuant to Article 2. The Holders shall have no other rights under or in respect of this Agreement, and the Commitment shall not otherwise be enforceable by any creditor of Seller or by any other Person other than the parties hereto, and no such creditor or other Person is intended to be, or shall be, a third party beneficiary of any provision of this Agreement.
- 6.2. Non-Transferable; Successors. The Commitment is solely for the benefit of Seller and shall not inure to the benefit of any other Person (other than the Holders to the extent set forth in Section 6.1), including any entity to which the charter of Seller may be transferred, to any LLRE or to any other successor to the assets, liabilities or operations of Seller. The Commitment may not be assigned or otherwise transferred, in whole or in part, to any Person (including, for the avoidance of doubt, any LLRE to which a receiver has assigned all or a portion of Seller's assets) without the prior written consent of Purchaser (which may be withheld in its sole discretion). In no event shall any successor to Seller (including such an LLRE) be entitled to the benefit of the Commitment without the prior written consent of Purchaser. Seller and Conservator, for themselves and on behalf of their permitted successors, covenant and agree not to transfer or purport to transfer the Commitment in contravention of the terms hereof, and any such attempted transfer shall be null and void *ab initio*. It is the expectation of the parties that, in the event Seller were placed into receivership and an LLRE formed to purchase certain of its assets and assume certain of its liabilities, the Commitment would remain with Seller for the benefit of the holders of the

debt of Seller not assumed by the LLRE.

- 6.3. Amendments; Waivers. This Agreement may be waived or amended solely by a writing executed by both of the parties hereto, and, with respect to amendments to or waivers of the provisions of Sections 5.3, 6.2 and 6.11, the Conservator; provided, however, that no such waiver or amendment shall decrease the aggregate Commitment or add conditions to funding the amounts required to be funded by Purchaser under the Commitment if such waiver or amendment would, in the reasonable opinion of Seller, adversely affect in any material respect the holders of debt securities of Seller and/or the beneficiaries of Mortgage Guarantee Obligations, in each case in their capacities as such, after taking into account any alternative arrangements that may be implemented concurrently with such waiver or amendment. In no event shall any rights granted hereunder prevent the parties hereto from waiving or amending in any manner whatsoever the covenants of Seller hereunder.
- 6.4. Governing Law; Jurisdiction; Venue. This Agreement and the Warrant shall be governed by, and construed in accordance with, the federal law of the United States of America if and to the extent such federal law is applicable, and otherwise in accordance with the laws of the State of New York. The Senior Preferred Stock shall be governed as set forth in the terms thereof. Except as provided in section 6.1 and as otherwise required by law, the United States District Court for the District of Columbia shall have exclusive jurisdiction over all civil actions arising out of this Agreement, the Commitment, the Senior Preferred Stock and the Warrant, and venue for any such civil action shall lie exclusively in the United States District Court for the District of Columbia.
- 6.5. *Notices*. Any notices delivered pursuant to or in connection with this Agreement shall be delivered to the applicable parties at the addresses set forth below:

If to Seller:

Federal National Mortgage Association c/o Federal Housing Finance Authority 1700 G Street, NW 4th Floor Washington, DC 20552 Attention: General Counsel

If to Purchaser:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington DC 20220 Attention: Under Secretary for Domestic Finance with a copy to:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington DC 20220 Attention: General Counsel

If to Conservator:

Federal Housing Finance Authority 1700 G Street, NW 4th Floor Washington, DC 20552 Attention: General Counsel

All notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail. All notices hereunder shall be effective upon receipt.

- 6.6. *Disclaimer of Guarantee*. This Agreement and the Commitment are not intended to and shall not be deemed to constitute a guarantee by Purchaser or any other agency or instrumentality of the United States of the payment or performance of any debt security or any other obligation, indebtedness or liability of Seller of any kind or character whatsoever.
- 6.7. Effect of Order; Injunction; Decree. If any order, injunction or decree is issued by any court of competent jurisdiction that vacates, modifies, amends, conditions, enjoins, stays or otherwise affects the appointment of Conservator as conservator of Seller or otherwise curtails Conservator's powers as such conservator (except in each case any order converting the conservatorship to a receivership under Section 1367(a) of the FHE Act), Purchaser may by written notice to Conservator and Seller declare this Agreement null and void, whereupon all transfers hereunder (including the issuance of the Senior Preferred Stock and the Warrant and any funding of the Commitment) shall be rescinded and unwound and all obligations of the parties (other than to effectuate such rescission and unwind) shall immediately and automatically terminate.
- 6.8. Business Day. To the extent that any deadline or date of performance of any right or obligation set forth herein shall fall on a day other than a Business Day, then such deadline or date of performance shall automatically be extended to the next succeeding Business Day.
- 6.9. *Entire Agreement*. This Agreement, together with the Senior Preferred Stock and Warrant, contains the entire agreement between the parties hereto with respect to the transactions contemplated hereby and supersedes and cancels all prior agreements, including, but not limited to, all proposals, term sheets, statements, letters of intent or representations, written or oral, with respect thereto.
- 6.10. *Remedies*. In the event of a breach by Seller of any covenant or representation of Seller set forth herein, Purchaser shall be entitled to specific performance (in the case of a breach of

covenant), damages and such other remedies as may be available at law or in equity; <u>provided</u>, that Purchaser shall not have the right to terminate the Commitment solely as a result of any such breach, and compliance with the covenants and the accuracy of the representations set forth in this Agreement shall not be conditions to funding the Commitment.

- 6.11. *Tax Reporting*. Neither Seller nor Conservator shall take, or shall permit any of their respective successors or assigns to take, a position for any tax, accounting or other purpose that is inconsistent with Internal Revenue Service Notice 2008-76 (or the regulations to be issued pursuant to such Notice) regarding the application of Section 382 of the Internal Revenue Code of 1986, as amended, a copy of which Notice has been provided to Seller in connection with the execution of this Agreement.
- 6.12. Non-Severability. Each of the provisions of this Agreement is integrated with and integral to the whole and shall not be severable from the remainder of the Agreement. In the event that any provision of this Agreement, the Senior Preferred Stock or the Warrant is determined to be illegal or unenforceable, then Purchaser may, in its sole discretion, by written notice to Conservator and Seller, declare this Agreement null and void, whereupon all transfers hereunder (including the issuance of the Senior Preferred Stock and the Warrant and any funding of the Commitment) shall be rescinded and unwound and all obligations of the parties (other than to effectuate such rescission and unwind) shall immediately and automatically terminate.

[Signature Page Follows]

EXHIBIT F

AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT

AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT (this "Agreement") dated as of September 26, 2008, between the UNITED STATES DEPARTMENT OF THE TREASURY ("Purchaser") and FEDERAL HOME LOAN MORTGAGE CORPORATION ("Seller"), acting through the Federal Housing Finance Agency (the "Agency") as its duly appointed conservator (the Agency in such capacity, "Conservator"). Reference is made to Article 1 below for the meaning of capitalized terms used herein without definition.

Background

- A. The Agency has been duly appointed as Conservator for Seller pursuant to Section 1367(a) of the Federal Housing Enterprises Financial Safety and Soundness Act of 1992 (as amended, the "FHE Act"). Conservator has determined that entry into this Agreement is (i) necessary to put Seller in a sound and solvent condition; (ii) appropriate to carry on the business of Seller and preserve and conserve the assets and property of Seller; and (iii) otherwise consistent with its powers, authorities and responsibilities.
- B. Purchaser is authorized to purchase obligations and other securities issued by Seller pursuant to Section 306(l) of the Federal Home Loan Mortgage Corporation Act, as amended (the "Charter Act"). The Secretary of the Treasury has determined, after taking into consideration the matters set forth in Section 306(l)(1)(C) of the Charter Act, that the purchases contemplated herein are necessary to (i) provide stability to the financial markets; (ii) prevent disruptions in the availability of mortgage finance; and (iii) protect the taxpayer.
- C. Purchaser and Seller executed and delivered the Senior Preferred Stock Purchase Agreement dated as of September 7, 2008 (the "Original Agreement"), and the parties thereto desire to amend and restate the Original Agreement in its entirety as set forth herein.

THEREFORE, the parties hereto agree as follows:

Terms and Conditions

1. **DEFINITIONS**

As used in this Agreement, the following terms shall have the meanings set forth below:

"Affiliate" means, when used with respect to a specified Person (i) any direct or indirect holder or group (as defined in Sections 13(d) and 14(d) of the Exchange Act) of holders of 10.0% or more of any class of capital stock of such Person and (ii) any current or former director or officer of such Person, or any other current or former employee of such Person that currently exercises or formerly exercised a material degree of Control over such Person, including without limitation each current or former Named Executive Officer of such Person.

- "Available Amount" means, as of any date of determination, the lesser of (a) the Deficiency Amount as of such date and (b) the Maximum Amount as of such date.
- "Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under United States federal law and the law of the State of New York.
- "Capital Lease Obligations" of any Person shall mean the obligations of such Person to pay rent or other amounts under any lease of (or other similar arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP and, for purposes hereof, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.
- "Control" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities, by contract or otherwise.
- "Deficiency Amount" means, as of any date of determination, the amount, if any, by which (a) the total liabilities of Seller exceed (b) the total assets of Seller (such assets excluding the Commitment and any unfunded amounts thereof), in each case as reflected on the balance sheet of Seller as of the applicable date set forth in this Agreement, prepared in accordance with GAAP; provided, however, that:
 - (i) for the avoidance of doubt, in measuring the Deficiency Amount liabilities shall exclude any obligation in respect of any capital stock of Seller, including the Senior Preferred Stock contemplated herein;
 - (ii) in the event that Seller becomes subject to receivership or other liquidation process or proceeding, "Deficiency Amount" shall mean, as of any date of determination, the amount, if any, by which (a) the total allowed claims against the receivership or other applicable estate (excluding any liabilities of or transferred to any LLRE (as defined in Section 5.4(a)) created by a receiver) exceed (b) the total assets of such receivership or other estate (excluding the Commitment, any unfunded amounts thereof and any assets of or transferred to any LLRE, but including the value of the receiver's interest in any LLRE);
 - (iii) to the extent Conservator or a receiver of Seller, or any statute, rule, regulation or court of competent jurisdiction, specifies or determines that a liability of Seller (including without limitation a claim against Seller arising from rescission of a purchase or sale of a security issued by Seller (or guaranteed by Seller or with respect to which Seller is otherwise liable) or for damages arising from the purchase, sale or retention of such a security) shall be subordinated (other than pursuant to a contract providing for such subordination) to all other liabilities of Seller or shall be treated on par with any class of equity of Seller, then such liability shall be excluded in the calculation of Deficiency Amount; and

- (iv) the Deficiency Amount may be increased above the otherwise applicable amount by the mutual written agreement of Purchaser and Seller, each acting in its sole discretion.
- "Designated Representative" means Conservator or (a) if Conservator has been superseded by a receiver pursuant to Section 1367(a) of the FHE Act, such receiver, or (b) if Seller is not in conservatorship or receivership pursuant to Section 1367(a) of the FHE Act, Seller's chief financial officer.
- "Director" shall mean the Director of the Agency.
- "Effective Date" means the date on which this Agreement shall have been executed and delivered by both of the parties hereto.
- "Equity Interests" of any Person shall mean any and all shares, interests, rights to purchase or otherwise acquire, warrants, options, participations or other equivalents of or interests in (however designated) equity, ownership or profits of such Person, including any preferred stock, any limited or general partnership interest and any limited liability company membership interest, and any securities or other rights or interests convertible into or exchangeable for any of the foregoing.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the SEC promulgated thereunder.
- "GAAP" means generally accepted accounting principles in effect in the United States as set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board from time to time.
- "Indebtedness" of any Person means, for purposes of Section 5.5 only, without duplication, (a) all obligations of such Person for money borrowed by such Person, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (d) all obligations of such Person issued or assumed as the deferred purchase price of property or services, other than trade accounts payable, (e) all Capital Lease Obligations of such Person, (f) obligations, whether contingent or liquidated, in respect of letters of credit (including standby and commercial), bankers' acceptances and similar instruments and (g) any obligation of such Person, contingent or otherwise, guaranteeing or having the economic effect of guaranteeing any Indebtedness of the types set forth in clauses (a) through (f) payable by another Person other than Mortgage Guarantee Obligations.
- "Liquidation End Date" means the date of completion of the liquidation of Seller's assets.
- "Maximum Amount" means, as of any date of determination, \$100,000,000,000 (one hundred billion dollars), less the aggregate amount of funding under the Commitment prior to such date.

- "Mortgage Assets" of any Person means assets of such Person consisting of mortgages, mortgage loans, mortgage-related securities, participation certificates, mortgage-backed commercial paper, obligations of real estate mortgage investment conduits and similar assets, in each case to the extent such assets would appear on the balance sheet of such Person in accordance with GAAP as in effect as of the date hereof (and, for the avoidance of doubt, without giving effect to any change that may be made hereafter in respect of Statement of Financial Accounting Standards No. 140 or any similar accounting standard).
- "Mortgage Guarantee Obligations" means guarantees, standby commitments, credit enhancements and other similar obligations of Seller, in each case in respect of Mortgage Assets.
- "Named Executive Officer" has the meaning given to such term in Item 402(a)(3) of Regulation S-K under the Exchange Act, as in effect on the date hereof.
- "Person" shall mean any individual, corporation, limited liability company, partnership, joint venture, association, joint-stock company, trust, estate, unincorporated organization or government or any agency or political subdivision thereof, or any other entity whatsoever.
- "SEC" means the Securities and Exchange Commission.
- "Senior Preferred Stock" means the Variable Liquidation Preference Senior Preferred Stock of Seller, substantially in the form of Exhibit A hereto.
- "Warrant" means a warrant for the purchase of common stock of Seller representing 79.9% of the common stock of Seller on a fully-diluted basis, substantially in the form of Exhibit B hereto.

2. COMMITMENT

- 2.1. Commitment. Purchaser hereby commits to provide to Seller, on the terms and conditions set forth herein, immediately available funds in an amount up to but not in excess of the Available Amount, as determined from time to time (the "Commitment"); provided, that in no event shall the aggregate amount funded under the Commitment exceed \$100,000,000,000 (one hundred billion dollars). The liquidation preference of the Senior Preferred Stock shall increase in connection with draws on the Commitment, as set forth in Section 3.3 below.
- 2.2. Quarterly Draws on Commitment. Within fifteen (15) Business Days following the determination of the Deficiency Amount, if any, as of the end of each fiscal quarter of Seller which ends on or before the Liquidation End Date, the Designated Representative may, on behalf of Seller, request that Purchaser provide immediately available funds to Seller in an amount up to but not in excess of the Available Amount as of the end of such quarter. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains a certification of the Designated Representative that the requested amount does not exceed the Available Amount as of the end of the applicable quarter. Purchaser shall provide such funds within sixty (60) days of its receipt of such request or, following any determination by the Director that the Director will be mandated by law to appoint a receiver for Seller if such funds are not received sooner, such shorter period as may be necessary

to avoid such mandatory appointment of a receiver if reasonably practicable taking into consideration Purchaser's access to funds.

- 2.3. Accelerated Draws on Commitment. Immediately following any determination by the Director that the Director will be mandated by law to appoint a receiver for Seller prior to the Liquidation End Date unless Seller's capital is increased by an amount (the "Special Amount") up to but not in excess of the then current Available Amount (computed based on a balance sheet of Seller prepared in accordance with GAAP that differs from the most recent balance sheet of Seller delivered in accordance with Section 5.9(a) or (b)) on a date that is prior to the date that funds will be available to Seller pursuant to Section 2.2, Conservator may, on behalf of Seller, request that Purchaser provide to Seller the Special Amount in immediately available funds. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains certifications of Conservator that (i) the requested amount does not exceed the Available Amount (including computations in reasonable detail and satisfactory to Purchaser of the then existing Deficiency Amount) and (ii) the requested amount is required to avoid the imminent mandatory appointment of a receiver for Seller. Purchaser shall provide such funds within thirty (30) days of its receipt of such request or, if reasonably practicable taking into consideration Purchaser's access to funds, any shorter period as may be necessary to avoid mandatory appointment of a receiver.
- 2.4. Final Draw on Commitment. Within fifteen (15) Business Days following the determination of the Deficiency Amount, if any, as of the Liquidation End Date (computed based on a balance sheet of Seller as of the Liquidation End Date prepared in accordance with GAAP), the Designated Representative may, on behalf of Seller, request that Purchaser provide immediately available funds to Seller in an amount up to but not in excess of the Available Amount as of the Liquidation End Date. Any such request shall be valid only if it is in writing, is timely made, specifies the account of Seller to which such funds are to be transferred, and contains a certification of the Designated Representative that the requested amount does not exceed the Available Amount (including computations in reasonable detail and satisfactory to Purchaser of the Deficiency Amount as of the Liquidation End Date). Purchaser shall provide such funds within sixty (60) days of its receipt of such request.
- 2.5. Termination of Purchaser's Obligations. Subject to earlier termination pursuant to Section 6.7, all of Purchaser's obligations under and in respect of the Commitment shall terminate upon the earliest of: (a) if the Liquidation End Date shall have occurred, (i) the payment in full of Purchaser's obligations with respect to any valid request for funds pursuant to Section 2.4 or (ii) if there is no Deficiency Amount on the Liquidation End Date or if no such request pursuant to Section 2.4 has been made, the close of business on the 15th Business Day following the determination of the Deficiency Amount, if any, as of the Liquidation End Date; (b) the payment in full of, defeasance of or other reasonable provision for all liabilities of Seller, whether or not contingent, including payment of any amounts that may become payable on, or expiry of or other provision for, all Mortgage Guarantee Obligations and provision for unmatured debts; and (c) the funding by Purchaser under the Commitment of an aggregate of \$100,000,000,000,000 (one hundred billion dollars). For the avoidance of doubt, the Commitment shall not be terminable by Purchaser solely by reason of (i) the conservatorship, receivership or other insolvency proceeding of Seller or (ii) the Seller's financial condition or any adverse change in Seller's financial condition.

3. PURCHASE OF SENIOR PREFERRED STOCK AND WARRANT; FEES

- 3.1. *Initial Commitment Fee.* In consideration of the Commitment, and for no additional consideration, on the Effective Date (or as soon thereafter as is practicable) Seller shall sell and issue to Purchaser, and Purchaser shall purchase from Seller, (a) one million (1,000,000) shares of Senior Preferred Stock, with an initial liquidation preference equal to \$1,000 per share (\$1,000,000,000 (one billion dollars) liquidation preference in the aggregate), and (b) the Warrant.
- 3.2. *Periodic Commitment Fee*. (a) Commencing March 31, 2010, Seller shall pay to Purchaser quarterly, on the last day of March, June, September and December of each calendar year (each a "<u>Periodic Fee Date</u>"), a periodic commitment fee (the "<u>Periodic Commitment Fee</u>"). The Periodic Commitment Fee shall accrue from January 1, 2010.
- (b) The Periodic Commitment Fee is intended to fully compensate Purchaser for the support provided by the ongoing Commitment following December 31, 2009. The amount of the Periodic Commitment Fee shall be set not later than December 31, 2009 with respect to the ensuing five-year period, shall be reset every five years thereafter and shall be determined with reference to the market value of the Commitment as then in effect. The amount of the Periodic Commitment Fee shall be mutually agreed by Purchaser and Seller, subject to their reasonable discretion and in consultation with the Chairman of the Federal Reserve; provided, that Purchaser may waive the Periodic Commitment Fee for up to one year at a time, in its sole discretion, based on adverse conditions in the United States mortgage market.
- (c) At the election of Seller, the Periodic Commitment Fee may be paid in cash or by adding the amount thereof ratably to the liquidation preference of each outstanding share of Senior Preferred Stock so that the aggregate liquidation preference of all such outstanding shares of Senior Preferred Stock is increased by an amount equal to the Periodic Commitment Fee. Seller shall deliver notice of such election not later than three (3) Business Days prior to each Periodic Fee Date. If the Periodic Commitment Fee is not paid in cash by 12:00 pm (New York time) on the applicable Periodic Fee Date (irrespective of Seller's election pursuant to this subsection), Seller shall be deemed to have elected to pay the Periodic Commitment Fee by adding the amount thereof to the liquidation preference of the Senior Preferred Stock, and the aggregate liquidation preference of the outstanding shares of Senior Preferred Stock shall thereupon be automatically increased, in the manner contemplated by the first sentence of this section, by an aggregate amount equal to the Periodic Commitment Fee then due.
- 3.3. Increases of Senior Preferred Stock Liquidation Preference as a Result of Funding under the Commitment. The aggregate liquidation preference of the outstanding shares of Senior Preferred Stock shall be automatically increased by an amount equal to the amount of each draw on the Commitment pursuant to Article 2 that is funded by Purchaser to Seller, such increase to occur simultaneously with such funding and ratably with respect to each share of Senior Preferred Stock.
- 3.4. *Notation of Increase in Liquidation Preference*. Seller shall duly mark its records to reflect each increase in the liquidation preference of the Senior Preferred Stock contemplated

herein (but, for the avoidance of doubt, such increase shall be effective regardless of whether Seller has properly marked its records).

4. REPRESENTATIONS

Seller represents and warrants as of the Effective Date, and shall be deemed to have represented and warranted as of the date of each request for and funding of an advance under the Commitment pursuant to Article 2, as follows:

- 4.1. *Organization and Good Standing*. Seller is a corporation, chartered by the Congress of the United States, duly organized, validly existing and in good standing under the laws of the United States and has all corporate power and authority to carry on its business as now conducted and as proposed to be conducted.
- 4.2. *Organizational Documents*. Seller has made available to Purchaser a complete and correct copy of its charter and bylaws, each as amended to date (the "<u>Organizational Documents</u>"). The Organizational Documents are in full force and effect. Seller is not in violation of any provision of its Organizational Documents.
- 4.3. Authorization and Enforceability. All corporate or other action on the part of Seller or Conservator necessary for the authorization, execution, delivery and performance of this Agreement by Seller and for the authorization, issuance and delivery of the Senior Preferred Stock and the Warrant being purchased under this Agreement, has been taken. This Agreement has been duly and validly executed and delivered by Seller and (assuming due authorization, execution and delivery by the Purchaser) shall constitute the valid and legally binding obligation of Seller, enforceable against Seller in accordance with its terms, except to the extent the enforceability thereof may be limited by bankruptcy laws, insolvency laws, reorganization laws, moratorium laws or other laws of general applicability affecting creditors' rights generally or by general equitable principles (regardless of whether enforcement is sought in a proceeding in equity or at law). The Agency is acting as conservator for Seller under Section 1367 of the FHE Act. The Board of Directors of Seller, by valid action at a duly called meeting of the Board of Directors on September 6, 2008, consented to the appointment of the Agency as conservator for purposes of Section 1367(a)(3)(I) of the FHE Act, and the Director of the Agency has appointed the Agency as Conservator for Seller pursuant to Section 1367(a)(1) of the FHE Act, and each such action has not been rescinded, revoked or modified in any respect.
- 4.4. *Valid Issuance*. When issued in accordance with the terms of this Agreement, the Senior Preferred Stock and the Warrant will be duly authorized, validly issued, fully paid and non-assessable, free and clear of all liens and preemptive rights. The shares of common stock to which the holder of the Warrant is entitled have been duly and validly reserved for issuance. When issued and delivered in accordance with the terms of this Agreement and the Warrant, such shares will be duly authorized, validly issued, fully paid and nonassessable, free and clear of all liens and preemptive rights.

4.5. Non-Contravention.

- (a) The execution, delivery or performance by Seller of this Agreement and the consummation by Seller of the transactions contemplated hereby do not and will not (i) conflict with or violate any provision of the Organizational Documents of Seller; (ii) conflict with or violate any law, decree or regulation applicable to Seller or by which any property or asset of Seller is bound or affected, or (iii) result in any breach of, or constitute a default (with or without notice or lapse of time, or both) under, or give to others any right of termination, amendment, acceleration or cancellation of, or result in the creation of a lien upon any of the properties or assets of Seller, pursuant to any note, bond, mortgage, indenture or credit agreement, or any other contract, agreement, lease, license, permit, franchise or other instrument or obligation to which Seller is a party or by which Seller is bound or affected, other than, in the case of clause (iii), any such breach, default, termination, amendment, acceleration, cancellation or lien that would not have and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the business, property, operations or condition of the Seller, the authority of the Conservator or the validity or enforceability of this Agreement (a "Material Adverse Effect").
- (b) The execution and delivery of this Agreement by Seller does not, and the consummation by Seller of the transactions contemplated by this Agreement will not, require any consent, approval, authorization, waiver or permit of, or filing with or notification to, any governmental authority or any other person, except for such as have already been obtained.

5. COVENANTS

From the Effective Date until such time as the Senior Preferred Stock shall have been repaid or redeemed in full in accordance with its terms:

- 5.1. Restricted Payments. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, declare or pay any dividend (preferred or otherwise) or make any other distribution (by reduction of capital or otherwise), whether in cash, property, securities or a combination thereof, with respect to any of Seller's Equity Interests (other than with respect to the Senior Preferred Stock or the Warrant) or directly or indirectly redeem, purchase, retire or otherwise acquire for value any of Seller's Equity Interests (other than the Senior Preferred Stock or the Warrant), or set aside any amount for any such purpose.
- 5.2. Issuance of Capital Stock. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, sell or issue Equity Interests of Seller or any of its subsidiaries of any kind or nature, in any amount, other than the sale and issuance of the Senior Preferred Stock and Warrant on the Effective Date and the common stock subject to the Warrant upon exercise thereof, and other than as required by (and pursuant to) the terms of any binding agreement as in effect on the date hereof.
- 5.3. *Conservatorship*. Seller shall not (and Conservator, by its signature below, agrees that it shall not), without the prior written consent of Purchaser, terminate, seek termination of or permit to be terminated the conservatorship of Seller pursuant to Section 1367 of the FHE Act, other

than in connection with a receivership pursuant to Section 1367 of the FHE Act.

- 5.4. *Transfer of Assets*. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, sell, transfer, lease or otherwise dispose of (in one transaction or a series of related transactions) all or any portion of its assets (including Equity Interests in other persons, including subsidiaries), whether now owned or hereafter acquired (any such sale, transfer, lease or disposition, a "<u>Disposition</u>"), other than Dispositions for fair market value:
- (a) to a limited life regulated entity ("<u>LLRE</u>") pursuant to Section 1367(i) of the FHE Act;
- (b) of assets and properties in the ordinary course of business, consistent with past practice;
- (c) in connection with a liquidation of Seller by a receiver appointed pursuant to Section 1367(a) of the FHE Act;
 - (d) of cash or cash equivalents for cash or cash equivalents; or
 - (e) to the extent necessary to comply with the covenant set forth in Section 5.7 below.
- 5.5. *Indebtedness*. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, incur, assume or otherwise become liable for (a) any Indebtedness if, after giving effect to the incurrence thereof, the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis would exceed 110.0% of the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis as of June 30, 2008 or (b) any Indebtedness if such Indebtedness is subordinated by its terms to any other Indebtedness of Seller or the applicable subsidiary. For purposes of this covenant the acquisition of a subsidiary with Indebtedness will be deemed to be the incurrence of such Indebtedness at the time of such acquisition.
- 5.6. Fundamental Changes. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, (i) merge into or consolidate or amalgamate with any other Person, or permit any other Person to merge into or consolidate or amalgamate with it, (ii) effect a reorganization or recapitalization involving the common stock of Seller, a reclassification of the common stock of Seller or similar corporate transaction or event or (iii) purchase, lease or otherwise acquire (in one transaction or a series of transactions) all or substantially all of the assets of any other Person or any division, unit or business of any Person.
- 5.7. *Mortgage Assets*. Seller shall not own, as of any applicable date, Mortgage Assets in excess of (i) on December 31, 2009, \$850 billion, or (ii) on December 31 of each year thereafter, 90.0% of the aggregate amount of Mortgage Assets of Seller as of December 31 of the immediately preceding calendar year; <u>provided</u>, that in no event shall Seller be required under this Section 5.7 to own less than \$250 billion in Mortgage Assets.

5.8. *Transactions with Affiliates*. Seller shall not, and shall not permit any of its subsidiaries to, without the prior written consent of Purchaser, engage in any transaction of any kind or nature with an Affiliate of Seller unless such transaction is (i) pursuant to this Agreement, the Senior Preferred Stock or the Warrant, (ii) upon terms no less favorable to Seller than would be obtained in a comparable arm's-length transaction with a Person that is not an Affiliate of Seller or (iii) a transaction undertaken in the ordinary course or pursuant to a contractual obligation or customary employment arrangement in existence as of the date hereof.

5.9. *Reporting*. Seller shall provide to Purchaser:

- (a) not later than the time period specified in the SEC's rules and regulations with respect to issuers as to which Section 13 and 15(d) of the Exchange Act apply, annual reports on Form 10-K (or any successor or comparable form) containing the information required to be contained therein (or required in such successor or comparable form);
- (b) not later than the time period specified in the SEC's rules and regulations with respect to issuers as to which Section 13 and 15(d) of the Exchange Act apply, reports on Form 10-Q (or any successor or comparable form) containing the information required to be contained therein (or required in such successor or comparable form);
- (c) promptly from time to time after the occurrence of an event required to be therein reported (and in any event within the time period specified in the SEC's rules and regulations), such other reports on Form 8-K (or any successor or comparable form);
- (d) concurrently with any delivery of financial statements under paragraphs (a) or (b) above, a certificate of the Designated Representative, (i) certifying that Seller is (and since the last such certificate has at all times been) in compliance with each of the covenants contained herein and that no representation made by Seller herein or in any document delivered pursuant hereto or in connection herewith was false or misleading in any material respect when made, or, if the foregoing is not true, specifying the nature and extent of the breach of covenant and/or representation and any corrective action taken or proposed to be taken with respect thereto, and (ii) setting forth computations in reasonable detail and satisfactory to the Purchaser of the Deficiency Amount, if any;
- (e) promptly, from time to time, such other information regarding the operations, business affairs, plans, projections and financial condition of Seller, or compliance with the terms of this Agreement, as Purchaser may reasonably request; and
 - (f) as promptly as reasonably practicable, written notice of the following:
 - (i) the occurrence of the Liquidation End Date;
 - (ii) the filing or commencement of, or any written threat or notice of intention of any Person to file or commence, any action, suit or proceeding, whether at law or in equity or by or before any governmental authority or in arbitration, against Conservator, Seller or any other Person which, if adversely determined, would reasonably be expected to have a Material Adverse Effect;

- (iii) any other development that is not a matter of general public knowledge and that has had, or would reasonably be expected to have, a Material Adverse Effect.
- 5.10. Executive Compensation. Seller shall not, without the consent of the Director, in consultation with the Secretary of the Treasury, enter into any new compensation arrangements with, or increase amounts or benefits payable under existing compensation arrangements of, any Named Executive Officer of Seller

6. MISCELLANEOUS

- 6.1. No Third-Party Beneficiaries. Until the termination of the Commitment, at any time during the existence and continuance of a payment default with respect to debt securities issued by Seller and/or a default by Seller with respect to any Mortgage Guarantee Obligations, any holder of such defaulted debt securities or beneficiary of such Mortgage Guarantee Obligations (collectively, the "Holders") may (a) deliver notice to the Seller and the Designated Representative requesting exercise of all rights available to them under this Agreement to draw on the Commitment up to the lesser of the amount necessary to cure the outstanding payment defaults and the Available Amount as of the last day of the immediately preceding fiscal quarter (the "Demand Amount"), (b) if Seller and the Designated Representative fail to act as requested within thirty (30) days of such notice, seek judicial relief for failure of the Seller to draw on the Commitment, and (c) if Purchaser shall fail to perform its obligations in respect of any draw on the Commitment, and Seller and/or the Designated Representative shall not be diligently pursuing remedies in respect of such failure, file a claim in the United States Court of Federal Claims for relief requiring Purchaser to pay Seller the Demand Amount in the form of liquidated damages. Any payment of liquidated damages to Seller under the previous sentence shall be treated for all purposes, including the provisions of the Senior Preferred Stock and Section 3.3 of this Agreement, as a draw and funding of the Commitment pursuant to Article 2. The Holders shall have no other rights under or in respect of this Agreement, and the Commitment shall not otherwise be enforceable by any creditor of Seller or by any other Person other than the parties hereto, and no such creditor or other Person is intended to be, or shall be, a third party beneficiary of any provision of this Agreement.
- 6.2. Non-Transferable; Successors. The Commitment is solely for the benefit of Seller and shall not inure to the benefit of any other Person (other than the Holders to the extent set forth in Section 6.1), including any entity to which the charter of Seller may be transferred, to any LLRE or to any other successor to the assets, liabilities or operations of Seller. The Commitment may not be assigned or otherwise transferred, in whole or in part, to any Person (including, for the avoidance of doubt, any LLRE to which a receiver has assigned all or a portion of Seller's assets) without the prior written consent of Purchaser (which may be withheld in its sole discretion). In no event shall any successor to Seller (including such an LLRE) be entitled to the benefit of the Commitment without the prior written consent of Purchaser. Seller and Conservator, for themselves and on behalf of their permitted successors, covenant and agree not to transfer or purport to transfer the Commitment in contravention of the terms hereof, and any such attempted transfer shall be null and void *ab initio*. It is the expectation of the parties that, in the event Seller were placed into receivership and an LLRE formed to purchase certain of its assets and assume certain of its liabilities, the Commitment would remain with Seller for the benefit of the holders of the

debt of Seller not assumed by the LLRE.

- 6.3. Amendments; Waivers. This Agreement may be waived or amended solely by a writing executed by both of the parties hereto, and, with respect to amendments to or waivers of the provisions of Sections 5.3, 6.2 and 6.11, the Conservator; provided, however, that no such waiver or amendment shall decrease the aggregate Commitment or add conditions to funding the amounts required to be funded by Purchaser under the Commitment if such waiver or amendment would, in the reasonable opinion of Seller, adversely affect in any material respect the holders of debt securities of Seller and/or the beneficiaries of Mortgage Guarantee Obligations, in each case in their capacities as such, after taking into account any alternative arrangements that may be implemented concurrently with such waiver or amendment. In no event shall any rights granted hereunder prevent the parties hereto from waiving or amending in any manner whatsoever the covenants of Seller hereunder.
- 6.4. Governing Law; Jurisdiction; Venue. This Agreement and the Warrant shall be governed by, and construed in accordance with, the federal law of the United States of America if and to the extent such federal law is applicable, and otherwise in accordance with the laws of the State of New York. The Senior Preferred Stock shall be governed as set forth in the terms thereof. Except as provided in section 6.1 and as otherwise required by law, the United States District Court for the District of Columbia shall have exclusive jurisdiction over all civil actions arising out of this Agreement, the Commitment, the Senior Preferred Stock and the Warrant, and venue for any such civil action shall lie exclusively in the United States District Court for the District of Columbia.
- 6.5. *Notices*. Any notices delivered pursuant to or in connection with this Agreement shall be delivered to the applicable parties at the addresses set forth below:

If to Seller:

Federal Home Loan Mortgage Corporation c/o Federal Housing Finance Authority 1700 G Street, NW 4th Floor Washington, DC 20552 Attention: General Counsel

If to Purchaser:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington DC 20220

Attention: Under Secretary for Domestic Finance

with a copy to:

United States Department of the Treasury 1500 Pennsylvania Avenue, NW Washington DC 20220 Attention: General Counsel

If to Conservator:

Federal Housing Finance Authority 1700 G Street, NW 4th Floor Washington, DC 20552 Attention: General Counsel

All notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail. All notices hereunder shall be effective upon receipt.

- 6.6. Disclaimer of Guarantee. This Agreement and the Commitment are not intended to and shall not be deemed to constitute a guarantee by Purchaser or any other agency or instrumentality of the United States of the payment or performance of any debt security or any other obligation, indebtedness or liability of Seller of any kind or character whatsoever.
- 6.7. Effect of Order; Injunction; Decree. If any order, injunction or decree is issued by any court of competent jurisdiction that vacates, modifies, amends, conditions, enjoins, stays or otherwise affects the appointment of Conservator as conservator of Seller or otherwise curtails Conservator's powers as such conservator (except in each case any order converting the conservatorship to a receivership under Section 1367(a) of the FHE Act), Purchaser may by written notice to Conservator and Seller declare this Agreement null and void, whereupon all transfers hereunder (including the issuance of the Senior Preferred Stock and the Warrant and any funding of the Commitment) shall be rescinded and unwound and all obligations of the parties (other than to effectuate such rescission and unwind) shall immediately and automatically terminate.
- 6.8. Business Day. To the extent that any deadline or date of performance of any right or obligation set forth herein shall fall on a day other than a Business Day, then such deadline or date of performance shall automatically be extended to the next succeeding Business Day.
- 6.9. *Entire Agreement*. This Agreement, together with the Senior Preferred Stock and Warrant, contains the entire agreement between the parties hereto with respect to the transactions contemplated hereby and supersedes and cancels all prior agreements, including, but not limited to, all proposals, term sheets, statements, letters of intent or representations, written or oral, with respect thereto.
- 6.10. *Remedies*. In the event of a breach by Seller of any covenant or representation of Seller set forth herein, Purchaser shall be entitled to specific performance (in the case of a breach of

covenant), damages and such other remedies as may be available at law or in equity; <u>provided</u>, that Purchaser shall not have the right to terminate the Commitment solely as a result of any such breach, and compliance with the covenants and the accuracy of the representations set forth in this Agreement shall not be conditions to funding the Commitment.

- 6.11. *Tax Reporting*. Neither Seller nor Conservator shall take, or shall permit any of their respective successors or assigns to take, a position for any tax, accounting or other purpose that is inconsistent with Internal Revenue Service Notice 2008-76 (or the regulations to be issued pursuant to such Notice) regarding the application of Section 382 of the Internal Revenue Code of 1986, as amended, a copy of which Notice has been provided to Seller in connection with the execution of this Agreement.
- 6.12. Non-Severability. Each of the provisions of this Agreement is integrated with and integral to the whole and shall not be severable from the remainder of the Agreement. In the event that any provision of this Agreement, the Senior Preferred Stock or the Warrant is determined to be illegal or unenforceable, then Purchaser may, in its sole discretion, by written notice to Conservator and Seller, declare this Agreement null and void, whereupon all transfers hereunder (including the issuance of the Senior Preferred Stock and the Warrant and any funding of the Commitment) shall be rescinded and unwound and all obligations of the parties (other than to effectuate such rescission and unwind) shall immediately and automatically terminate.

[Signature Page Follows]

EXHIBIT G

AMENDMENT TO AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT

AMENDMENT dated as of May 6, 2009, to the AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT dated as of September 26, 2008, between the UNITED STATES DEPARTMENT OF THE TREASURY ("Purchaser"), and FEDERAL NATIONAL MORTGAGE ASSOCIATION ("Seller"), acting through the Federal Housing Finance Agency (the "Agency") as its duly appointed conservator (the Agency in such capacity, "Conservator").

Background

- A. Purchaser and Seller have heretofore entered into the Amended and Restated Senior Preferred Stock Purchase Agreement dated as of September 26, 2008 (the "Amended and Restated Agreement").
- B. In the Amended and Restated Agreement, Purchaser committed itself to provide to Seller, on the terms and conditions provided in the Amended and Restated Agreement, immediately available funds in an amount as determined from time to time as provided in the Amended and Restated Agreement, but in no event in an aggregate amount exceeding \$100,000,000,000.
- C. Purchaser and Seller now desire to enter into an amendment to the Amended and Restated Agreement for the purpose of increasing to \$200,000,000,000 the maximum aggregate amount permitted to be provided to Seller under the Amended and Restated Agreement, and for the purpose of amending the terms of the Amended and Restated Agreement in certain other respects.
- D. Purchaser and Seller are each authorized to enter into this Amendment to the Amended and Restated Agreement ("this Amendment") increasing to \$200,000,000,000 the maximum aggregate amount permitted to be provided to Seller under the Amended and Restated Agreement, and amending the terms of the Amended and Restated Agreement in certain other respects.

THEREFORE, for and in consideration of the mutual agreements herein contained and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Purchaser and Seller agree as follows:

Terms and Conditions

1. Definitions.

Capitalized terms used and not defined in this Amendment shall have the respective meanings given such terms in the Amended and Restated Agreement.

2. <u>Amendment to Section 1 (Relating to Definition of New Defined Term "Executive Officer").</u>

Section 1 of the Amended and Restated Agreement is hereby amended to insert the following new defined term and corresponding definition after the definition of the term "Exchange Act":

"Executive Officer" has the meaning given to such term in Exchange Act Rule 3b-7, as in effect on the date hereof.

3. Amendment to Section 1 (Relating to Definition of "Indebtedness").

The definition of "Indebtedness" in Section 1 of the Amended and Restated Agreement is hereby amended to read as follows:

"Indebtedness" of any Person means, for purposes of Section 5.5 only, without duplication, (a) all obligations of such Person for money borrowed by such Person, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (d) all obligations of such Person issued or assumed as the deferred purchase price of property or services, other than trade accounts payable, (e) all Capital Lease Obligations of such Person, (f) obligations, whether contingent or liquidated, in respect of letters of credit (including standby and commercial), bankers' and similar instruments, and (g) any obligation of such Person, contingent or otherwise, guaranteeing or having the economic effect of guaranteeing and Indebtedness of the types set forth in clauses (a) through (f) payable by another Person other than Mortgage Guarantee Obligations (and, for the avoidance of doubt, without giving effect to any change that may be made hereafter in respect of Statement of Financial Accounting Standards No. 140 or any similar accounting standard).

4. Amendment to Section 1 (Relating to Definition of "Maximum Amount").

The definition of "Maximum Amount" in Section 1 of the Amended and Restated Agreement is hereby amended to read as follows:

"Maximum Amount" means, as of any date of determination, \$200,000,000,000 (two hundred billion dollars), less the aggregate amount of funding under the Commitment prior to such date.

5. Amendment to Section 2.1 (Relating to the Commitment).

Section 2.1 of the Amended and Restated Agreement is hereby amended to read as follows:

2.1 Commitment. Purchaser hereby commits to provide to Seller, on the terms and conditions set forth herein, immediately available funds in an amount up to but not in excess of the Available Amount, as determined from time to time (the "Commitment"); provided, that in no event shall the aggregate amount funded under the Commitment exceed \$200,000,000,000 (two hundred billion

dollars). The liquidation preference of Senior Preferred Stock shall increase in connection with draws on the Commitment, as set forth in Section 3.3 below.

6. Amendment to Section 2.5 (Relating to Termination of Purchaser's Obligations).

Section 2.5 of the Amended and Restated Agreement is hereby amended to read as follows:

2.5 Termination of Purchaser's Obligations. Subject to earlier termination pursuant to Section 6.7, all of Purchaser's obligations under and in respect of the Commitment shall terminate upon the earliest of: (a) if the Liquidation End Date shall have occurred, (i) the payment in full of Purchaser's obligations with respect to any valid request for funds pursuant to Section 2.4 or (ii) if there is no Deficiency Amount on the Liquidation End Date or if no such request pursuant to Section 2.4 has been made, the close of business on the 15th Business Day following the determination of the Deficiency Amount, if any, as of the Liquidation End Date; (b) the payment in full of, defeasance of or other reasonable provision for all liabilities of Seller, whether or not contingent, including payment of any amounts that may become payable on, or expiry of or other provision for, all Mortgage Guarantee Obligations and provision for unmatured debts; and (c) the funding by Purchaser under the Commitment of an aggregate of \$200,000,000,000 (two hundred billion dollars). For avoidance of doubt, the Commitment shall not be terminable by Purchaser solely by reason of (i) the conservatorship, receivership or other insolvency proceeding of Seller or (ii) the Seller's financial condition or any adverse change in Seller's financial condition.

7. Amendment to Section 5.5 (Relating to Indebtedness).

Section 5.5 of the Amended and Restated Agreement is hereby amended to read as follows:

5.5. Indebtedness. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, incur, assume or otherwise become liable for (a) any Indebtedness if, after giving effect to the incurrence thereof, the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis would exceed (i) through and including December 30, 2010, 120.0% of the amount of Mortgage Assets Seller is permitted by Section 5.7 to own on December 31, 2009; and (ii) beginning on December 31, 2010, and through and including December 30, 2011, and each year thereafter, 120.0% of the amount of Mortgage Assets Seller is permitted by Section 5.7 to own on December 31 of the immediately preceding calendar year, or (b) any Indebtedness if such Indebtedness is subordinated by its terms to any other Indebtedness of Seller or the applicable subsidiary. For purposes of this covenant the acquisition of a subsidiary with Indebtedness will be deemed to be the incurrence of such Indebtedness at the time of such acquisition.

8. Amendment to Section 5.7 (Relating to Owned Mortgage Assets).

Section 5.7 of the Amended and Restated Agreement is hereby amended to read as follows:

5.7. Mortgage Assets. Seller shall not own, as of any applicable date, Mortgage Assets in excess of (i) on December 31, 2009, \$900 billion, or (ii) on December 31 of each year thereafter, 90.0% of the aggregate amount of Mortgage Assets of Seller as of December 31 of the immediately preceding calendar year; provided, that in no event shall Seller be required under this Section 5.7 to own less than \$250 billion in Mortgage Assets.

9. Amendment to Section 5.10 (Relating to Executive Compensation).

Section 5.10 of the Amended and Restated Agreement is hereby amended to read as follows:

5.10. Executive Compensation. Seller shall not, without the consent of the Director, in consultation with the Secretary of the Treasury, enter into any new compensation arrangements with, or increase amounts or benefits payable under existing compensation arrangements of, any Named Executive Officer or other Executive Officer of Seller.

10. Amended and Restated Agreement to Continue, as Amended.

Except as expressly modified by this Amendment, the Amended and Restated Agreement shall continue in full force and effect.

11. Effective Date.

This Amendment shall not become effective until it has been executed by both of Purchaser and Seller. When this Amendment has been so executed, it shall become effective as of the date first above written.

FEDERAL NATIONAL MORTGAGE ASSOCIATION, by

Federal Housing Finance Agency, its Conservator

James B. Lockhart III

Director

UNITED STATES DEPARTMENT OF THE TREASURY

Timoth F. Geithner

Secretary of the Treasury

EXHIBIT H

AMENDMENT TO AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT

AMENDMENT dated as of May 6, 2009, to the AMENDED AND RESTATED SENIOR PREFERRED STOCK PURCHASE AGREEMENT dated as of September 26, 2008, between the UNITED STATES DEPARTMENT OF THE TREASURY ("Purchaser"), and FEDERAL HOME LOAN MORTGAGE CORPORATION ("Seller"), acting through the Federal Housing Finance Agency (the "Agency") as its duly appointed conservator (the Agency in such capacity, "Conservator").

Background

- A. Purchaser and Seller have heretofore entered into the Amended and Restated Senior Preferred Stock Purchase Agreement dated as of September 26, 2008 (the "Amended and Restated Agreement").
- B. In the Amended and Restated Agreement, Purchaser committed itself to provide to Seller, on the terms and conditions provided in the Amended and Restated Agreement, immediately available funds in an amount as determined from time to time as provided in the Amended and Restated Agreement, but in no event in an aggregate amount exceeding \$100,000,000,000.
- C. Purchaser and Seller now desire to enter into an amendment to the Amended and Restated Agreement for the purpose of increasing to \$200,000,000,000 the maximum aggregate amount permitted to be provided to Seller under the Amended and Restated Agreement, and for the purpose of amending the terms of the Amended and Restated Agreement in certain other respects.
- D. Purchaser and Seller are each authorized to enter into this Amendment to the Amended and Restated Agreement ("this Amendment") increasing to \$200,000,000,000 the maximum aggregate amount permitted to be provided to Seller under the Amended and Restated Agreement, and amending the terms of the Amended and Restated Agreement in certain other respects.

THEREFORE, for and in consideration of the mutual agreements herein contained and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Purchaser and Seller agree as follows:

Terms and Conditions

1. Definitions.

Capitalized terms used and not defined in this Amendment shall have the respective meanings given such terms in the Amended and Restated Agreement.

2. Amendment to Section 1 (Relating to Definition of New Defined Term "Executive Officer").

Section 1 of the Amended and Restated Agreement is hereby amended to insert the following new defined term and corresponding definition after the definition of the term "Exchange Act":

"Executive Officer" has the meaning given to such term in Exchange Act Rule 3b-7, as in effect on the date hereof.

3. Amendment to Section 1 (Relating to Definition of "Indebtedness").

The definition of "Indebtedness" in Section 1 of the Amended and Restated Agreement is hereby amended to read as follows:

"Indebtedness" of any Person means, for purposes of Section 5.5 only, without duplication, (a) all obligations of such Person for money borrowed by such Person, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (d) all obligations of such Person issued or assumed as the deferred purchase price of property or services, other than trade accounts payable, (e) all Capital Lease Obligations of such Person, (f) obligations, whether contingent or liquidated, in respect of letters of credit (including standby and commercial), bankers' and similar instruments, and (g) any obligation of such Person, contingent or otherwise, guaranteeing or having the economic effect of guaranteeing and Indebtedness of the types set forth in clauses (a) through (f) payable by another Person other than Mortgage Guarantee Obligations (and, for the avoidance of doubt, without giving effect to any change that may be made hereafter in respect of Statement of Financial Accounting Standards No. 140 or any similar accounting standard).

4. Amendment to Section 1 (Relating to Definition of "Maximum Amount").

The definition of "Maximum Amount" in Section 1 of the Amended and Restated Agreement is hereby amended to read as follows:

"Maximum Amount" means, as of any date of determination, \$200,000,000,000 (two hundred billion dollars), less the aggregate amount of funding under the Commitment prior to such date.

Amendment to Section 2.1 (Relating to the Commitment).

Section 2.1 of the Amended and Restated Agreement is hereby amended to read as follows:

2.1 Commitment. Purchaser hereby commits to provide to Seller, on the terms and conditions set forth herein, immediately available funds in an amount up to but not in excess of the Available Amount, as determined from time to time (the "Commitment"); provided, that in no event shall the aggregate amount funded under the Commitment exceed \$200,000,000,000 (two hundred billion

dollars). The liquidation preference of Senior Preferred Stock shall increase in connection with draws on the Commitment, as set forth in Section 3.3 below.

6. Amendment to Section 2.5 (Relating to Termination of Purchaser's Obligations).

Section 2.5 of the Amended and Restated Agreement is hereby amended to read as follows:

2.5 Termination of Purchaser's Obligations. Subject to earlier termination pursuant to Section 6.7, all of Purchaser's obligations under and in respect of the Commitment shall terminate upon the earliest of: (a) if the Liquidation End Date shall have occurred, (i) the payment in full of Purchaser's obligations with respect to any valid request for funds pursuant to Section 2.4 or (ii) if there is no Deficiency Amount on the Liquidation End Date or if no such request pursuant to Section 2.4 has been made, the close of business on the 15th Business Day following the determination of the Deficiency Amount, if any, as of the Liquidation End Date; (b) the payment in full of, defeasance of or other reasonable provision for all liabilities of Seller, whether or not contingent, including payment of any amounts that may become payable on, or expiry of or other provision for, all Mortgage Guarantee Obligations and provision for unmatured debts; and (c) the funding by Purchaser under the Commitment of an aggregate of \$200,000,000,000 (two hundred billion dollars). For avoidance of doubt, the Commitment shall not be terminable by Purchaser solely by reason of (i) the conservatorship, receivership or other insolvency proceeding of Seller or (ii) the Seller's financial condition or any adverse change in Seller's financial condition.

Amendment to Section 5.5 (Relating to Indebtedness).

Section 5.5 of the Amended and Restated Agreement is hereby amended to read as follows:

5.5. Indebtedness. Seller shall not, and shall not permit any of its subsidiaries to, in each case without the prior written consent of Purchaser, incur, assume or otherwise become liable for (a) any Indebtedness if, after giving effect to the incurrence thereof, the aggregate Indebtedness of Seller and its subsidiaries on a consolidated basis would exceed (i) through and including December 30, 2010, 120.0% of the amount of Mortgage Assets Seller is permitted by Section 5.7 to own on December 31, 2009; and (ii) beginning on December 31, 2010, and through and including December 30, 2011, and each year thereafter, 120.0% of the amount of Mortgage Assets Seller is permitted by Section 5.7 to own on December 31 of the immediately preceding calendar year, or (b) any Indebtedness if such Indebtedness is subordinated by its terms to any other Indebtedness of Seller or the applicable subsidiary. For purposes of this covenant the acquisition of a subsidiary with Indebtedness will be deemed to be the incurrence of such Indebtedness at the time of such acquisition.

8. Amendment to Section 5.7 (Relating to Owned Mortgage Assets).

Section 5.7 of the Amended and Restated Agreement is hereby amended to read as follows:

5.7. Mortgage Assets. Seller shall not own, as of any applicable date, Mortgage Assets in excess of (i) on December 31, 2009, \$900 billion, or (ii) on December 31 of each year thereafter, 90.0% of the aggregate amount of Mortgage Assets of Seller as of December 31 of the immediately preceding calendar year; provided, that in no event shall Seller be required under this Section 5.7 to own less than \$250 billion in Mortgage Assets.

9. Amendment to Section 5.10 (Relating to Executive Compensation).

Section 5.10 of the Amended and Restated Agreement is hereby amended to read as follows:

5.10. Executive Compensation. Seller shall not, without the consent of the Director, in consultation with the Secretary of the Treasury, enter into any new compensation arrangements with, or increase amounts or benefits payable under existing compensation arrangements of, any Named Executive Officer or other Executive Officer of Seller.

10. Amended and Restated Agreement to Continue, as Amended.

Except as expressly modified by this Amendment, the Amended and Restated Agreement shall continue in full force and effect.

11. Effective Date.

This Amendment shall not become effective until it has been executed by both of Purchaser and Seller. When this Amendment has been so executed, it shall become effective as of the date first above written.

FEDERAL HOME LOAN MORTGAGE CORPORATION, by

Federal Housing Finance Agency, its Conservator

ames B. Lockhart II

UNITED STATES DEPARTMENT OF THE TREASURY

Timothy F. Geithner Secretary of the Treasury